THE DBL-1/TGF-β SIGNALING PATHWAY REGULATES AN ARRAY OF BEHAVIORAL, MOLECULAR, AND PHYSIOLOGICAL DEFENSES IN CAENORHABDITIS ELEGANS

A DISSERTATION

SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIRMENTS FOR THE DEGREE OF DOCTOR OF MOLECULAR BIOLOGY IN THE GRADUATE SCHOOL OF THE TEXAS WOMAN'S UNIVERSITY

DEPARTMENT OF BIOLOGY

COLLEGE OF ARTS AND SCIENCES

ΒY

BHOOMI MADHU, M.S.

DENTON, TEXAS

AUGUST 2021

Copyright © 2021 by Bhoomi Madhu

DEDICATION

To my parents, grandparents, uncle and aunt, Het, Anuj, Smit, Manan, Chintan, and my friends.

ACKNOWLEDGEMENTS

I thank my mentor, Dr. Tina L. Gumienny, for her committed mentoring, constant support and guidance, introducing me to C. elegans, and wholeheartedly supporting me in my journey of crafting a project that integrated my research interests. Dr. G, thank you for training me not only with the bench work but also training me to be a strong, adept, and successful professional. I thank my committee members, Dr. Christopher Brower, Dr. Heather Conrad-Webb, Dr. Laura Hanson, and Dr. K. Shane Broughton for their valuable insights, guidance, and support. I am grateful to the entire Gumienny lab for their advice, brainstorming discussions, and support. I thank Dr. Lionel Faure for being an awesome teacher to me and training me on the lipid work. I thank Farhan Lakdawala for being constantly supportive and collaborating on multiple research projects. I thank our collaborator, Dr. Douglas Jasmer, and his lab members for working with me on studying the epicuticle composition of the parasitic nematode, Ascaris suum. I am very grateful to Mr. James Lundgren, Dr. Paul Yeatts, Dr. Wanyi Wang, and TWU's Center for Research Design and Analysis for assistance with statistical analyses. I thank Dr. Amy Jo Hammett for providing some bacterial strains for this work. Some *C. elegans* strains were obtained from the Caenorhabditis Genetics Center (CGC) which is funded by NIH Office of Research Infrastructure Programs (P40 OD010440). Some figures were created with BioRender.

iii

I am very grateful to Dr. Juliet Spencer for being a wonderful leader and an advocate for Biology graduate students. I thank the Biology department staff and faculty for helping me throughout my journey at TWU. I thank the funding agencies that supported me and my work at TWU. This work was supported by NIH R01GM097591, USDA AFRI, TWU Research Enhancement program, and TWU internal funding to Dr. Gumienny, TWU Experiential Learning Scholar Award, and TWU Student Small grant. I express my gratitude for the TWU Scholarships that were awarded to me. I deeply thank my family, Farhan, Yasar, Geetika, Sanika, Gaurav, Nirali, Akshaya, and Sukhbir for being my strongest support system all throughout my graduate studies.

ABSTRACT

BHOOMI MADHU

THE DBL-1/TGF-β SIGNALING PATHWAY REGULATES AN ARRAY OF BEHAVIORAL, MOLECULAR, AND PHYSIOLOGICAL DEFENSES IN *CAENORHABDITIS ELEGANS*

AUGUST 2021

Organisms possess mechanisms to protect themselves from environmental threats. Animals have innate immune responses that include behavioral, molecular, and physiological components. A conserved cell-cell signaling pathway, transforming growth factor-beta (TGF- β) signaling pathway, is a major regulator of innate immune responses in animals. However, the requirement for this pathway in generating specific, robust responses to different bacterial challenges has not previously been well characterized. We used the roundworm *Caenorhabditis elegans*, which has conserved innate immune responses and TGF- β signaling pathways, to address how organisms use TGF- β signaling to tailor immune responses to different environmental threats. We tested the requirement for DBL-1/TGF- β signaling in a diverse array of immune responses to a selected panel of Gram-negative and Gram-positive bacteria. We showed that robust, protective, and specific responses to different bacteria require functional DBL-1 signaling. Animals lacking DBL-1 were more susceptible to all test bacteria. We discovered that canonical DBL-1 signaling is required to

suppress avoidance to Gram-negative bacteria, but non-canonical DBL-1 signaling represses the avoidance to Gram-positive bacteria. Furthermore, this work identified a novel role for SMA-4/co-Smad that is independent of the DBL-1 signaling pathway. This indicates that the DBL-1/TGF- β signaling pathway plays an important role in tailoring the animals' innate immune responses to the bacterial threat. Additionally, to investigate the role of saposin-like antimicrobial proteins in response to different environmental threats, we characterized the regulation of *spp-9* in response to the bacterial panel. We identified that *spp-9* is affected by both bacterial exposure and by starvation. We reported that the pathogen-specific regulation of *spp-9* expression was dependent on not only DBL-1/TGF- β signaling but also other innate immune signaling pathways including insulin-like and p38/MAP kinase. Lastly, to determine the role of DBL-1 signaling in regulating the epicuticle, a secreted lipid-rich barrier layer, we developed a method to isolate epicuticular lipids and characterized their composition. We showed that DBL-1 signaling regulates the composition and levels of both epicuticle and internal lipids. DBL-1 signaling also regulates expression of lipid metabolism genes. Collectively, these findings demonstrate bacteria-specific host immune responses regulated by the DBL-1/TGF-B signaling pathway.

vi

TABLE OF CONTENTS

P	age
DEDICATION	ii
ACKNOWLEDGEMENTS	iii
ABSTRACT	v
LIST OF TABLES	х
LIST OF FIGURES	xi
LIST OF ABBREVIATIONS.	. xiv
Chapter	
Chapter	
	1
IL THE DBI -1/TGE-B SIGNALING PATHWAY REGULATES PATHOGE	-N-
SPECIFIC INNATE IMMUNE RESPONSES IN C. FLEGANS	12
Abstract	12
Introduction	11
Matariala And Mathada	. 14
Materials And Meintenenee	. 10
Strains And Maintenance	. 10
C. elegans Strains	.16
Bacterial Strains	. 16
Lifespan Assay	. 17
Pharyngeal Pumping Rate	. 17
Intestinal Barrier Function Assay	. 17
Microbial Avoidance Assay	. 18
RNA Isolation	. 19
Differential Expression Analysis by RNA Sequencing	. 19
cDNA Synthesis and qRT-PCR	. 19
Imaging	20
Statistical Analyses	21
Results	22
Loss Of DBL-1 Reduces Lifespan Of Animals Fed On Specific	
Bacteria	22
Loss Of DBL-1 and Exposure to Specific Bacteria Reduce	
Feeding	26
Intestinal Integrity of Animals Is Not Altered by Loss of <i>dbl-1</i> or	
Exposure to Specific Bacteria	28
DBI -1 Signaling Is Required to Suppress Avoidance Against Gr.	20 2m-
Negative Bacteria	20
SMA-4 Acts Independently of Other DRI -1 Core Signaling	. 23
Components to Suppress Avoidance to Gram Desitive Pasteria	30
omponents to Suppress Avoluance to Gram-Fusilive Dacteria.	. 52
Sina-4 Expression is specifically induced in Response to Gram-	<u>ог</u>
Positive bacteria	35

DBL-1 Signaling Is Activated in Response to Gram-Negative Bacteria and Is Repressed in Response to Gram-Positive Bacteri	ia
	38
DBL-1 Mediates Both Common and Specific Gene Expression	00
Responses to Gram-Negative and -Positive Bacteria	40
DBI -1 Differentially Regulates Expression of Innate Immunity	10
Genes Specific to the Gram Nature of Bacteria	ΔΔ
Discussion	49
<u>Acknowledgements</u>	54
Author Contributions	5/
Corresponding Author	5/
Competing Interests	54
References	55
Supplementary Material	65
Supplementary material	00
III. CAENODHARDITIS ELEGANS BROOD SIZE AND EGG-LAVING	
	71
EFIDERIVIDIS ARE INDEPENDENT OF DBE-T SIGNALING	11
Description	71
Methods	74
Reagents	7/
References	75
	10
IV. CAENORHABDITIS ELEGANS SAPOSIN-LIKE SPP-9 IS INVOLVED IN	
SPECIFIC INNATE IMMUNE RESPONSES	77
	••
Abstract	.78
Introduction	.79
Materials and Methods	. 81
Strains and Maintenance	. 81
Dauer Assav	. 82
	. 82
RNA Isolation and gRT-PCR	.83
Data and Reagent Availability	.84
Results	.84
spp-9 Localizes in the Intestine and Head Neurons at All	0.
Developmental Stages	84
spp-9 Expression Is Increased in Starved Animals	86
spp-9 Expression Is Altered in Response to Select Gram-Positive	ڊ د
and Gram-Negative Pathogens	87
dhl-1 Regulates Endogenous spn-9 Expression Levels	90
spp-9 Expression Depends on DRI -1 and Other Innate Immune	00
Response Signaling Pathways, Depending on Pathogen	91
i tespence eignamig i aantajo, Boponang on i aanogon minimi	. .

spp-9 Reporter Activity Is Regulated by Innate Immunity/Stress Signaling Pathways94Discussion98Acknowledgements101Conflict of Interest101References102Supplementary Information111
V. DBL-1 REGULATES THE COMPOSITION OF THE PROTECTIVE SURFACE BARRIER OF <i>C. ELEGANS</i>
Introduction
Quantitative Reverse Transcription PCR
Loss of DBL-1 Does Not Affect Total Lipid Varieties but Reduces Unsaturated Fatty Acid Quantity
Discussion
REFERENCES

LIST OF TABLES

Table	Page
1.1.	DBL-1-associated phenotypes8
2.S1.	List of strains
2.S2.	List of primers for qRT-PCR65
2.S3.	Survival assay summary
2.S4.	RAD-SMAD reporter activity in response to Gram-positive bacteria 67
5.1.	Primer sequences for real-time PCR 122
5.2.	DBL-1-regulated lipid metabolism and putative transport genes identified
	using RNA-seq133

LIST OF FIGURES

Figure Page
1.1. Anatomy of <i>C. elegans</i> and different modes of infection in <i>C. elegans</i> 3
1.2. Structure of nematode cuticle4
1.3. DBL-1 signaling pathway7
1.4. Graphical representation of specific aims10
2.1. Loss of DBL-1 decreases lifespan of animals exposed to Gram-negative
and Gram-positive bacteria26
2.2. Loss of DBL-1 and exposure to specific bacteria results in decreased
pharyngeal pumping28
2.3. Avoidance to Gram-negative bacteria increases upon loss of DBL-1
signaling32
2.4. Avoidance to Gram-negative bacteria increases upon loss of canonical
DBL-1 signaling
2.5. Smad transcription factors gene expression is altered by specific bacteria
2.6. DBL-1 signaling is activated upon exposure to Gram-negative bacteria but
is repressed in response to Gram-positive bacteria40
2.7. Expression of innate immunity genes is differentially regulated by DBL-1
signaling in different bacterial environments43

2.8. Innate immune reporter activity is regulated by exposure to specific bacteria
and by DBL-1 signaling49
2.S1. Loss of DBL-1 does not affect intestinal integrity upon exposure to specific
bacteria69
2.S2. DBL-1 regulates differential gene expression in response to Gram-
negative and Gram-positive bacteria70
3.1. Effects of S. marcescens and S. epidermidis on egg laying and brood size
in wild-type and <i>dbl-1(nk3)</i> populations72
4.1. The spp-9 reporter is expressed throughout larval and adult stages in
intestine and head neurons87
4.2. Starvation increases <i>spp-9</i> reporter activity
4.3. spp-9 reporter activity is altered upon exposure to specific Gram-positive
and Gram-negative bacteria91
4.4. <i>spp-9</i> reporter activity is altered in response to specific pathogens by DBL-1
and other innate immunity signaling pathways95
4.5. spp-9 reporter activity is regulated by innate immunity signaling pathways
4.S1. Expression of the <i>spp-9</i> reporter increases over time
4.S2. <i>dbl-1</i> regulates endogenous <i>spp-9</i> expression levels
5.1. Loss of DBL-1 results in reduction of unsaturated fatty acid levels but does
not alter total lipid composition125

5.2.	2. Identification of surface-enriched and internal lipids of wild-type and <i>dbl-1(-)</i>		
	animals130		
5.3.	DBL-1 positively regulates expression of some lipid metabolism genes 134		
5.4.	Model of DBL-1 pathway-mediated regulation of both internal and		
	epicuticular lipids137		
6.1.	Model figure indicating involvement of DBL-1 in regulating pathogen-		
	specific response		

LIST OF ABBREVIATIONS

ANOVA	Analysis of variance	
B. megaterium	Bacillus megaterium	
C. elegans	Caenorhabditis elegans	
DNA	Deoxyribonucleic acid	
E. cloacae	Enterobacter cloacae	
E. coli	Escherichia coli	
E. faecalis	Enterococcus faecalis	
FA	Fatty acids	
FAMEs	Fatty acid methyl esters	
FUdR	5-Fluorodeoxyuridine	
GC	Gas chromatography	
GC-MS	Gas chromatography-mass spectrometry	
GFP	Green fluorescent protein	
K. oxytoca	Klebsiella oxytoca	
LPC	Lyso-phosphatidylcholine	
МАРК	Mitogen-activated protein kinase	
MAP2K	Mitogen-activated protein kinase kinase	
МАРЗК	Mitogen-activated protein kinase kinase kinase	
M. luteus	Micrococcus luteus	
mRNA	Messenger ribonucleic acid	

PA	Phosphatidic acid
PC	Phosphatidylcholine
PCR	Polymerase chain reaction
PE	Phosphatidylethanolamine
PI	Phosphatidylinositol
P. aeruginosa	Pseudomonas aeruginosa
qRT-PCR	Quantitative real-time polymerase chain reaction
RFP	Red fluorescent protein
RNA	Ribonucleic acid
RNA-seq	RNA sequencing
S. enterica	Salmonella enterica
S. epidermidis	Staphylococcus epidermidis
S. marcescens	Serratia marcescens
TEM	Transmission electron microscopy
TGF-β	Transforming growth factor beta
TLC	Thin layer chromatography

CHAPTER I

INTRODUCTION

All living organisms possess mechanisms to protect themselves from environmental threats. Notably, animals employ coordinated immune responses against pathogenic bacteria. In humans, immune responses are mounted through the innate immune system and the adaptive immune system¹. Innate immunity, which all animals have, is the first line of defense used by organisms to fight pathogens. Innate immunity also plays an important role in the regulation of adaptive immunity². Some organisms use adaptive immunity to fight pathogens if the innate immune responses are not sufficient³. Adaptive immunity is long lasting and involves antigen-specific responses^{4,5}. Innate immune responses are dynamically regulated. Living organisms utilize multiple cell signaling pathways to coordinate innate immune responses against pathogen exposure^{1,4}. How are these innate immune responses specific upon exposure to different pathogenic bacteria? What are the underlying mechanisms of the dynamic interplay between host signaling and different pathogen exposure? These are some questions that need to be addressed to reduce the fundamental knowledge gap in understanding the mechanisms of appropriate coordination of protective host responses against specific bacterial challenges. Dissecting the network of specific protective mechanisms against different bacterial exposures will provide a deeper understanding of how hosts tailor protective responses. These findings

can be used to identify specific mechanisms that can be targeted to regulate host defense responses.

Studying such robust interactions if innate immune responses in vivo in higher organisms is challenging because of confounding issues of adaptive immunity and ethical and practical constraints⁶. To identify dynamic innate immune responses in vivo, we use the model organism Caenorhabditis elegans. *C. elegans* is a free-living roundworm and is an established system to dissect evolutionarily conserved innate immunity mechanisms⁷. Roundworms are a diverse group of organisms (comprising both free-living and parasitic members) that have adapted to various habitats ranging from terrestrial to marine environments^{8,9}. In these environments, roundworms encounter bacteria, which can be beneficial or pathogenic to the roundworms¹⁰. Roundworms lack adaptive immunity and use their innate immune system of defenses to respond to pathogenic bacteria. To effectively fight infection, it is important that organisms appropriately coordinate and regulate innate immune responses¹. How animals coordinate innate immune responses against the various challenges in their natural environment—to protect roundworms or other animals, including humans—is a fundamental knowledge gap in the field.

C. elegans naturally thrive in a soil environment and are in constant association with a diverse range of microbes that are food and possible threat to these roundworms^{11,12}. Bacteria can infect *C. elegans* though multiple routes including intestine, cuticle, and tail (see Fig. 1.1)^{13,14}. As such, *C. elegans* have

developed powerful protective mechanisms for their defense. The first line of defense is behavioral: avoidance behavior and reduced food (bacteria) intake^{15–20}. *C. elegans* can sense and avoid pathogenic bacteria for protection against harmful environments^{16,21,22}.



The second defense mechanism is through physical means^{12,23,24}. One way is through using the pharyngeal grinder and another is by means of cuticle.

Bacteria that *C. elegans* feed on are destroyed mechanically by the pharyngeal grinder. The pharyngeal grinder keeps live bacteria from passing to the gut and colonizing the intestinal lumen. Intact bacteria that escape the grinder colonize (infect) the nematode intestine⁷. It is reported that several pathogenic bacteria including *Serratia marcescens*, *Pseudomonas aeruginosa*, and *Enterococcus faecalis* colonize nematode intestine and reduce lifespan^{25–27}. The cuticle (exoskeleton) is a physical barrier that protects the nematodes from their external environment (see Fig. 1.2). The cuticle is composed of multiple layers, including protein-rich inner layers, the lipid-rich epicuticle layer, and the outermost glycoprotein-rich surface coat (or glycocalyx). The protein composition of the cuticle is well characterized; however, its epicuticular lipid composition has not yet been dissected²⁸.



The third mechanism is molecular: induction of antimicrobial genes upon infection. Several pathogens are reported to infect *C. elegans*, including bacterial

strains *P. aeruginosa*, *Salmonella enterica*, *S. marcescens*, *Bacillus megaterium*, *E. faecalis*, *Mycobacterium fortuitum*, *Staphylococcus aureus*, and *Staphylococcus epidermidis*, and the fungus *Drechmeria coniospora*^{29–34}. Upon infection by these pathogens, Toll-like receptor, MAPK signaling, insulin-like signaling, and TGF- β signaling pathways are induced^{29–36}. These pathways regulate an overlapping set of target genes, indicating cross-regulation of the pathways when *C. elegans* are exposed to infection. As in other animals, including humans, genes targeted by these signaling pathways encode antimicrobial proteins. However, it is unclear how the host coordinates appropriate defenses against a variety of Gram-negative and Gram-positive bacteria. While many studies have elucidated robust roles of MAPK and insulinlike signaling pathways in defense responses, the role of TGF- β signaling in regulating defense responses tailored to the pathogenic challenge is not well characterized^{37–40}.

The mammalian TGF- β signaling pathway functions and pathway components are highly conserved in *C. elegans*. DBL-1 is one of the five TGF- β superfamily ligands in *C. elegans*. DBL-1 is secreted from neurons and binds to its receptors SMA-6 and DAF-4, which are present on the cell membrane of the hypodermis, pharynx, and intestine. Upon ligand binding, the SMA-6/DAF-4 receptor complex is activated by auto-phosphorylation. The phosphorylated receptor complex in turn activates downstream transcription factor SMA-3/Smad by phosphorylation, which then binds other Smad transcription factors SMA-2

and SMA-4. This activated transcription factor complex translocates to the nucleus to regulate transcription of target genes (see Fig. 1.3)^{41–43}. DBL-1 is known to play a role in innate immunity by regulating transcription of innate immunity genes and is also known to be involved in protecting animals from Gram-negative *S. marcescens* Db11³⁸. Additionally, the DBL-1 signaling pathway is also involved in regulating body size, male tail development, and reproductive aging (see Table 1.1)^{41,44–46}. The Gumienny lab reported another protective role of DBL-1/TGF- β signaling, regulating epicuticle lipid levels, in a dose-dependent manner⁴⁷. Previous reports indicate that overexpression of *dbl-1* enriches expression of many immune response genes, including lectins, saposin-like, and lysozymes^{32,48–51}. While the role of DBL-1 in defending nematodes from a few Gram-negative bacteria has been reported, its role in protection against Gram-positive bacterial infection is not well elucidated^{25,38,52}.



Phenotypes	Wild-type (WT) <i>dbl-1</i>	Loss of <i>dbl-1</i>
body size	WT	small
control of innate immune response genes	yes	altered
male tail morphology	WT	fused rays crumpled spicules
epicuticle lipid levels	WT	decreased
permeability to drugs	WT	increased
lifespan	WT	decreased
reproductive aging	WT	reduced and extended

Table 1.1. DBL-1-associated phenotypes observed in *C. elegans*.

We hypothesize that *C. elegans* use the DBL-1/TGF- β signaling pathway to coordinate specific innate immune responses against a variety of Gramnegative and Gram-positive bacteria. These test bacteria were selected because they can cause opportunistic infections in humans. We tested the hypothesis by addressing if *C. elegans* use the DBL-1/TGF-β signaling pathway to coordinate specific innate immune behavioral responses (Aim 1), molecular responses (Aim 2), and physiological responses (Aim 3) against a panel of Gram-negative and Gram-positive bacteria that cause opportunistic infections in humans (see Fig. 1.4). Aim 1 is summarized in Chapters II and III, Aim 2 is summarized in Chapters II and IV, and Aim 3 is summarized in Chapter V. Our major findings summarized in Chapter II indicate that the DBL-1/TGF-β signaling pathway plays an important role in influencing the animals' perception of pathogenicity and in turn appropriately balances innate immune responses. Animals lacking DBL-1 signaling are more susceptible to the bacteria irrespective of the Gram nature. We also identified a correlation between reduced feeding (a protective response) and animal survival in response to select bacteria. However, animals with reduced canonical DBL-1 signaling displayed a strong avoidance response selectively to the Gram-negative bacteria. Additionally, we identified signature molecular responses regulated by DBL-1 in response to different Gram-negative and Gram-positive bacteria. Finally, we discovered that differential regulation of

DBL-1 signaling is part of the host's molecular response to Gram-negative and Gram-positive bacteria.



Chapter III summarizes the findings on the role of DBL-1 signaling in coordinating changes in egg-laying, which is a known protective response to pathogenic bacteria. We identified that loss of DBL-1 resulted in reduced brood size upon exposure to Gram-negative *S. marcescens* and Gram-positive *S. epidermidis*. We found that the brood size alterations were independently regulated by loss of DBL-1 signaling and exposure to the pathogenic bacteria.

Chapter IV summarizes the findings on bacteria-specific regulation of an antimicrobial gene, *spp-9*. We identified that expression of this antimicrobial gene is not only altered by selective pathogens but also by starvation. We reported that the pathogen-specific regulation of *spp-9* expression is dependent on DBL- $1/TGF-\beta$ signaling. Additionally, we identified that other innate immunity signaling pathways, including insulin-like and p38/MAP kinase signaling, also regulate *spp-9* expression.

Chapter V provides a foundational understanding of the lipid composition of the *C. elegans* epicuticle and highlights the role of DBL-1/TGF- β signaling in regulating the epicuticle lipid composition. Additionally, we report that the DBL-1 signaling regulates the composition of internal lipids. We identified fatty acid metabolism genes regulated by DBL-1 signaling, and propose that changes in the expression of these genes results in the DBL-1-dependent lipid composition alterations.

Finally, Chapter VI provides a comprehensive model to summarize the role of DBL-1/TGF- β signaling in tailoring protective immune responses to a variety of Gram-negative and Gram-positive bacteria. This chapter culminates in questions that can be addressed using this work as a foundation to study differential host pathogen immune responses.

CHAPTER II

THE DBL-1/TGF-β SIGNALING PATHWAY REGULATES PATHOGEN-

SPECIFIC INNATE IMMUNE RESPONSES IN C. ELEGANS

This chapter contains a preprint: Madhu, B. and Gumienny, T.L. (2021) The DBL-1/TGF-β signaling pathway regulates pathogen-specific innate immune responses in *C. elegans*. bioRxiv 2021.03.30.437693; doi: https://doi.org/10.1101/2021.03.30.437693

Bhoomi Madhu and Tina L. Gumienny*

Department of Biology, Texas Woman's University, Denton, TX 76204-5799

*Corresponding author

Running title: pathogen-specific innate immune responses mediated by DBL-1 Key words: *dbl-1*, innate immunity, *C. elegans*, pathogen-specificity Correspondence: Tina L. Gumienny

> PO Box 425799 Denton, TX 76204-5799 Phone: 940-898-2295 Email: tgumienny@twu.edu

Abstract

Innate immunity in animals is orchestrated by multiple cell signaling pathways, including the TGF- β superfamily pathway. While the role of TGF- β signaling in innate immunity has been clearly identified, the requirement for this pathway in generating specific, robust responses to different bacterial challenges has not been characterized. Here, we address the role of DBL-1/TGF- β in regulating signature host defense responses to a wide range of bacteria in C. *elegans*. This work reveals a role of DBL-1/TGF-β in animal survival, organismal behaviors, and molecular responses in different environments. Additionally, we identify a novel role for SMA-4/Smad that suggests both DBL-1/TGF-βdependent and -independent functions in host avoidance responses. RNA-seq analyses and immunity reporter studies indicate DBL-1/TGF- β differentially regulates target gene expression upon exposure to different bacteria. Furthermore, the DBL-1/TGF- β pathway is itself differentially affected by the bacteria exposure. Collectively, these findings demonstrate bacteria-specific host immune responses regulated by the DBL-1/TGF- β signaling pathway.

Introduction

Living organisms recognize and respond to potential environmental insults by coordinating protective defenses¹. Invertebrates and vertebrates both employ conserved innate immune response as immediate front-line protection from challenges including pathogenic bacteria^{2–5}. These responses are tailored to the bacterial challenge. However, how these responses are specified and what the responses are to different pathogens remains a challenge⁶.

The roundworm *Caenorhabditis elegans* is an established model system to study regulation of immune responses *in vivo*⁷. *C. elegans* naturally thrives in a soil environment where it feeds on bacteria and is in constant association with a diverse range of microbes that are both food and threat^{3,7}. A limited number of pathogens are known to infect *C. elegans*, including mycobacterial, Gramnegative, and Gram-positive bacterial species, and fungi^{8–15}. *C. elegans* has an innate immune system that confers protection through behavioral, physical (exoskeleton), and molecular mechanisms⁷.

Infection in *C. elegans* induces molecular immune defenses coordinated by conserved Toll-like receptors, MAPK (mitogen-activated protein kinase) signaling, insulin-like signaling, and DBL-1/TGF- β (transforming growth factor β) signaling^{8–13,16,17}. These pathways regulate an overlapping set of target defense genes, indicative of coordinated crosstalk between these signaling pathways. While the roles of Toll, MAPK, and insulin-like signaling pathways in immune responses to a wide variety of bacteria are well characterized in *C. elegans* and

other organisms, a role for DBL-1/TGF-β signaling in eliciting robust targeted immune responses to different bacterial challenges has been identified but is not well defined^{16,18–22}. Previous reports indicate that overexpression of *dbl-1* enriches expression of many immune response genes including lectins, saposinlike proteins, and lysozymes^{11,23–25}. While the role of DBL-1 in defending nematodes from a few Gram-negative bacteria is reported, its possible role in protection against Gram-positive bacterial infection has not been well characterized^{20,26,27}.

In this work, we examined the role of DBL-1/TGF-β signaling in regulating an array of microbe-specific immune responses. Using behavioral and molecular approaches, we identified DBL-1-dependent and -independent immune responses that are tailored to the specific bacterial exposure. We also identified a non-canonical role for the DBL-1 pathway transcription regulator SMA-4 in an avoidance response to specific bacteria. Additionally, we show that DBL-1 signaling is induced in response to Gram-negative bacteria but is repressed in response to Gram-positive bacteria. We propose that animals lacking DBL-1 signaling respond with heightened avoidance behaviors to selected bacterial environments because they perceive the environment as more hostile. Collectively, our findings highlight a central role for DBL-1 in regulating a suite of bacteria-specific host defenses and also demonstrate bacteria-responsive regulation of DBL-1 signaling.

Materials and Methods

Strains and Maintenance

C. elegans strains

All *C. elegans* strains were maintained on EZ media plates at 20°C²⁸. *C. elegans* strains were maintained without contamination and starvation for at least five generations before every experiment. Supplementary Table 1 includes the list of all strains used in this study. These strains were generated by standard genetic crosses and confirmed by small body size phenotype and presence of fluorescence.

Bacterial strains

The bacterial strains used in this study include *Bacillus megaterium* (Carolina Biological Supply Company), *Escherichia coli* (OP50), *Enterobacter cloacae* (49141TM), *Enterococcus faecalis* (51299TM), *Klebsiella oxytoca* (49131TM), *Serratia marcescens* (Carolina Biological Supply Company), and *Staphylococcus epidermidis* (49134TM). *E. faecalis* in brain heart infusion media and all other bacteria in tryptic soy broth were grown for nine hours at 37°C as previously described²⁸. Bacterial cells were pelleted at 5000 rpm for 15 minutes and concentrated twenty-fold. EZ media plates were freshly seeded with concentrated bacteria in full lawns. The plates were incubated at 37°C overnight before using for experiments.

Lifespan Assay

Lifespan assay was performed as previously described^{29–31}. Concentrated bacterial cultures were spread on 6 cm diameter EZ media plates (full lawn plates) containing 50 µg/ml 5-fluorodeoxyuridine (FUdR) to cover the surface of the plates entirely. Wild-type and *dbl-1(-)* animals (n = at least 30) were fed on control and test bacteria on full lawn plates at the L4 stage in quadruplicate. The plates were scored for live and dead nematodes every 24 hours until all animals were dead. Animals were scored as dead if they did not respond to gentle touch with a sterilized platinum wire and were removed from the plate. At least three independent trials were performed. Worms that died by desiccating on the walls of the plates were censored from the analysis.

Pharyngeal Pumping Rate

Wild-type and *dbl-1(-)* L4 animals (n = 12) were fed on control and test bacteria on full lawn plates. The number of contractions of the pharyngeal bulb was counted for 20 seconds to calculate the pharyngeal pumping rate of animals. Two counts were made and averaged for each animal. Three independent trials were performed in triplicate³².

Intestinal Barrier Function Assay

The intestinal barrier function assay was performed as previously described³³. Wild-type and *dbl-1(-)* L4 animals were fed on control and test bacteria on full lawn plates. The assay was performed when about 50% of the population with the lowest mean lifespan remained alive. At least 15 animals

were sampled at the specified times to examine intestinal tissue integrity. The intestinal barrier integrity was assessed using a blue dye, eriogluacine disodium salt (5% wt/v), as an indicator of tissue integrity as the animals age. Leaking of this blue dye outside the intestinal lumen indicates reduced intestinal integrity. The animals were washed with S buffer and were incubated in erioglaucine disodium salt solution in a 1:1 ratio for 3 hours. The animals were then washed thrice with S buffer and were mounted on 2% agarose pads on glass slides. 10 µM levamisole was added to paralyze the animals. The animals were imaged on a Nikon DS-Ri2 camera mounted to a Nikon SMZ18 dissecting microscope. The leakiness of the intestine was assessed and scored as '1' for no leakage/no Smurf, '2' for mild leakage/mild Smurf, and '3' for severe leakage/severe Smurf phenotypes. The experiment was performed in at least three independent trials for each experimental condition.

Microbial Avoidance Assay

Microbial avoidance assays were performed as previously described³⁴. 20 μ l of the concentrated bacterial cultures were spotted on 6 cm diameter EZ media plates and incubated at 37°C overnight. Wild-type, *dbl-1(-)*, *sma-2(-)*, *sma-3(-)*, and *sma-4(-)* L4 hermaphrodites were placed on control and test bacterial lawn (n = 30 per condition/trial, performed in triplicate). The plates were scored for number of worms occupying the lawn at the indicated time points. Three independent trials were performed. The avoidance ratio (A) was calculated using the formula:

A = number of animals off the lawn/ total number of animals Avoidance of populations was scored as mild if less than 40% of the population was not on the food source (A < 0.4), as moderate avoidance if 0.4 < A < 0.6, and as strong avoidance if A > 0.6.

RNA Isolation

Animals were synchronized as embryos by bleaching mixed-stage populations³⁵. Total RNA was extracted from animals at 48 hours after the L4 stage. Total RNA was extracted by the freeze cracking method as previously described³⁶.

Differential Expression Analysis by RNA Sequencing

RNA from wild-type and *dbl-1(-)* adult populations fed on control and test bacteria was extracted in three independent trials. Sequencing libraries from the extracted RNA were generated using the NEBNext[®] RNA Library Prep Kit for Illumina[®] (NEB, USA) following manufacturer's recommendations. 1 µg RNA of each sample was used as input material for the RNA sample preparations. Novogene performed RNA sequencing of samples. Differential expression analysis of wild-type compared to *dbl-1(-)* populations grown on different bacteria was performed using the DESeq R package (1.18.0)³⁷. Genes with an adjusted *p*-value < 0.05 found by DESeq were assigned as differentially expressed.

cDNA Synthesis and qRT-PCR

After RNA isolation, cDNA was synthesized and quantitative real-time PCR was performed as previously described³⁸. 2 µg of total RNA isolated was

primed with oligo(dT) and reverse transcribed to yield cDNA using the SuperScript III reverse transcriptase kit as per manufacturer's protocol (Invitrogen). Real-time PCR was performed on a QuantStudio3 system (Applied Biosystems by Thermo Fisher Scientific) using the PowerUP SYBR Green master mix (Applied Biosystems) according to manufacturer's instructions. Three independent biological trials were performed. Each biological trial was performed in three technical replicates for each condition. Primer sequences are available in Supplementary Table 2. QuantStudio Design and Analysis Software v1.5.1 was used to calculate raw Ct values (Applied Biosystems by Thermo Fisher Scientific). The Ct values for the target genes were normalized to the housekeeping gene *act-1* (actin). Fold change in gene expression between experimental sample and the control was determined by using the formula: $2^{(c \Delta\Delta C_{t})}$.

Imaging

RAD-SMAD reporter strains were placed on full lawns of the control or test bacteria at the L4 stage. L2 progeny were mounted on 2% agarose pads and anesthetized by using 1 mM levamisole and fluorescence was captured by a Zeiss LSM 900 confocal microscope using a 40X oil objective. At least 15 animals with at least five hypodermal nuclei per worm in the focal plane were imaged per condition, giving a moderate effect size as determined by power analysis. The experiment was performed in three independent trials. The microscope conditions were optimized with respect to the control and test

conditions and kept consistent within each trial. Mean fluorescence intensities were measured as previously described using the Zeiss ZEN lite software³⁹.

The innate immune reporter strains were transferred to full lawns of the control and test bacteria at the L4 stage and were imaged after 48 hours of exposure. Fluorescence of the reporter strains was captured by a Nikon DS-Ri2 camera mounted on a Nikon SMZ18 dissecting microscope. Animals were mounted on 2% agarose pads and anesthetized with 1 mM levamisole. At least 15 animals were imaged per condition as determined by power analysis with a moderate effect size. The microscope conditions were optimized with respect to the control and test conditions and kept consistent within each trial. However, imaging exposure times were different between some trials to prevent saturation of signal in experimental conditions. Three independent trials were performed. Mean fluorescence intensities were measured using the Nikon NIS Elements AR v5.02 software.

Statistical Analyses

Lifespans of *C. elegans* populations were calculated by the Kaplan-Meier method and statistical analysis was performed using the two-tailed log-rank test. The average pharyngeal pumping rates were compared by the two-tailed unpaired *t*-test. The intestinal barrier function phenotypes were statistically analyzed by the Chi-square test. The avoidance ratio was compared by repeated measures ANOVA using Tukey's post-hoc test. qRT-PCR values and mean fluorescence intensities were evaluated using the two-tailed unpaired *t*-test. RNA
sequencing analysis was performed using the DESeq R package (1.18.0). The resulting *p*-values were adjusted using the Benjamini and Hochberg's approach for controlling the false discovery rate.

Results

Loss of DBL-1 Reduces Lifespan of Animals Fed on Specific Bacteria

To study the requirement for DBL-1 in specific responses to pathogens at behavioral, molecular, and physiological levels, we first established a panel of bacteria for innate immune studies in *C. elegans* that would facilitate genetic and molecular studies over time: previous studies have been limited in range of challenge and used a pathogen that killed animals in hours or a few days. The control bacteria chosen was Gram-negative *E. coli* OP50, a commonly used strain for laboratory culture of *C. elegans*. The panel of test bacteria comprises Gram-negative and -positive bacteria that are opportunistic pathogens in humans and are found in the natural habitat of *C. elegans*⁴⁰. We selected three Gram-negative test strains, *S. marcescens, E. cloacae*, and *K. oxytoca*, and three Gram-positive test strains, *B. megaterium* and *E. faecalis*, and *S. epidermidis*.

We first asked if our panel of opportunistic bacteria affect lifespan of *C. elegans*. Wild-type animals on Gram-negative *S. marcescens* have lifespans comparable to *E. coli*-fed animals. Interestingly, we noted an extended lifespan of wild-type animals on the other two Gram-negative strains and all three Grampositive strains (see Fig. 2.1, Supplementary Table 3).

Loss of DBL-1 has previously been shown to reduce lifespan of animals exposed to fungus D. coniospora, Gram-negative strains E. coli and S. marcescens (Db11), and Gram-positive *E. faecalis*^{13,20,41}. To determine if DBL-1 is required in maintaining lifespan of animals subjected to our bacterial panel, we compared lifespans of dbl-1(-) animals exposed to the test or control bacteria to the wild type. In our conditions, loss of DBL-1 does not alter lifespan of animals fed on *E. coli* (see Fig. 2.1a). However, loss of *dbl-1* results in a significantly shortened lifespan on *S. marcescens*, consistent with a previous report that used the more virulent *S. marcescens* strain Db11 (see Fig. 2.1d, Supplementary Table 3)²⁰. dbl-1 mutant animals did not have the lifespan extension seen in wildtype populations on *E. cloacae* or *K. oxytoca*: lifespans of *dbl-1* mutant populations were the same on these two Gram-negative bacterial strains as on *E. coli* (see Fig. 2.1b, c, Supplementary Table 3). The lifespan of *dbl-1(-)* animals was extended upon exposure to Gram-positive B. megaterium and E. faecalis compared to the *E. coli*-fed population's lifespan, but was not as extended as the wild-type lifespan (see Fig. 2.1e, f, Supplementary Table 3). Lastly, *dbl-1* mutant animals displayed a significantly decreased lifespan on S. epidermidis (see Fig. 2.1g, Supplementary Table 3).

In conclusion, we have identified a panel of human opportunistic pathogens that can be used to interrogate genetic contributions to innate immune defenses. DBL-1 is required for normal lifespan responses regardless of bacterial Gram nature (see Fig. 2.1b–g). These results provide evidence that DBL-1

signaling normally confers protection against these bacteria and may play a role in the lifespan extension observed on most bacteria in the panel.





Figure 2.1. Loss of DBL-1 decreases lifespan of animals exposed to Gramnegative and Gram-positive bacteria. Wild-type and *dbl-1(-)* animals were scored for survival over time from the L4 stage (t = 0 hours) on the following bacteria: a) *E. coli* OP50 (control), b) *E. cloacae*, c) *K. oxytoca*, d) *S. marcescens*, e) *B. megaterium*, f) *E. faecalis*, and g) *S. epidermidis*. Survival fraction was calculated by the Kaplan-Meier method. *p*-values were calculated using log-rank test and *p* < 0.01 compared to wild-type animals exposed to the same bacteria was considered significant. One representative trial of at least three is presented.

Loss of DBL-1 and Exposure to Specific Bacteria Reduce Feeding

One reason for the increased lifespan of *C. elegans* on select bacteria could be that the animals experience dietary restriction because they reduce bacterial consumption^{42,43}. To determine if animals reduce feeding on bacteria that increase lifespan, we measured and compared pharyngeal pumping rates of wild-type animals on control and test bacteria. Wild-type animals fed on *E. cloacae* exhibited a small but significant decrease in pharyngeal pumping. Pharyngeal pumping was not significantly reduced after exposure to *K. oxytoca*, which is not consistent with the mild lifespan extension. Wild-type animals fed on *S. marcescens* have the same pumping rate as on the control bacteria (see Fig. 2.2a). For wild-type animals fed on the three Gram-positive bacteria, though, the pumping rate was dramatically decreased, consistent with the lifespan extension these strains conferred to wild-type *C. elegans* (see Fig. 2.2b).

To determine whether DBL-1 affects feeding response on our panel of test bacteria, we compared the pharyngeal pumping rate of wild-type and *dbl-1(-)* animals on these bacteria. Loss of DBL-1 does not alter the pharyngeal pumping rate of animals fed on the control bacteria (see Fig. 2.2a, b). *dbl-1(-)* animals fed

on *E. cloacae* do not alter the pumping rate compared to either wild-type animals on E. cloacae or dbl-1 animals on E. coli (control). Animals lacking DBL-1 show a mild but significant decrease when they are fed on K. oxytoca and S. marcescens (see Fig. 2.2a). The feeding rate of *dbl-1(-)* animals is further reduced from the wild-type rate on *E. faecalis* and *S. epidermidis* (see Fig. 2.2b). There is no reproducibly significant decrease in the pharyngeal pumping rate of *dbl-1(-)* animals fed on *B. megaterium* in comparison to that of the wild type (see Fig. 2.2b). These results collectively indicate that while the feeding reduction caused by exposure to these Gram-positive bacteria is independent of DBL-1, a stronger pharyngeal pumping depression in *dbl-1(-)* populations occurs in response to some bacteria (both Gram-negative and -positive bacteria), providing support to the idea that loss of DBL-1 sensitizes animals to certain pathogenic stressors. These findings suggest that even though the organismal responses (lifespan) requiring DBL-1 are similar, the underlying causes might be different, differences in animal feeding being one.



Figure 2.2. Loss of DBL-1 and exposure to specific bacteria results in decreased pharyngeal pumping. Wild-type and *dbl-1(-)* animals were exposed to the following bacteria: a, b) *E. coli* OP50 (control); a) *E. cloacae, K. oxytoca, S. marcescens*; b) *B. megaterium, E. faecalis,* or *S. epidermidis* at the L4 stage. After 48 hours of exposure, the number of pharyngeal pumps were counted twice per 20 seconds. The pharyngeal pumps were averaged for each animal. One representative trial of at least three is presented. Error bars represent standard deviation. n = 12 per condition. * p < 0.01, ns not significant, compared to wild-type animals exposed to the same bacteria, and # p < 0.01, respective genotype exposed to test bacteria in comparison to control bacteria by two-way ANOVA followed by unpaired *t*-test.

Intestinal Integrity of Animals Is Not Altered By Loss of *dbl-1* or Exposure

to Specific Bacteria

The integrity of intestine can also be disrupted by exposure to pathogenic

bacteria, but progressive loss of intestinal integrity is also a feature of aging^{33,44}.

To further examine the role of DBL-1 in the organismal responses to this panel of

bacteria, we investigated the integrity of the intestinal barrier of animals fed on

different bacteria. Using a cell-impermeable blue dye, we compared the intestinal

barrier function of wild-type and *dbl-1(-)* animals exposed to control and test

bacteria when 50% of the population on the test bacteria was still alive.

Consistent with previous reports, wild-type animals on *E. coli* exhibited an agedependent reduction of intestinal integrity⁴⁴. Animals lacking DBL-1 and grown on *E. coli* also displayed a significant decline in intestinal integrity similar to the wild type (see Supplementary Fig. 1a). Exposure of wild-type or *dbl-1* mutant animals to any of the Gram-negative and -positive bacteria in the panel did not result in further, reproducible decreases of intestinal integrity compared to *E. coli* (see Supplementary Fig. 1). Therefore, *dbl-1* is not required for intestinal integrity, nor its age-related decline. Furthermore, this result suggests the lifespan changes observed in *dbl-1* mutant populations are not caused by loss of intestinal integrity.

DBL-1 Signaling Is Required to Suppress Avoidance Against Gram-Negative Bacteria

C. elegans can sense and avoid pathogenic bacteria for protection against harmful environments ^{45–47}. To test how our panel of selected Gram-negative and -positive bacteria evokes an avoidance response in *C. elegans*, we performed the avoidance assay. We measured the avoidance response of wild-type animals fed on the test bacteria over the first two days of adulthood and compared it with the avoidance response on the control bacteria. Wild-type animals do not avoid *E. coli* (see Fig. 2.3a). We found that wild-type animals mildly avoid *S. marcescens*, but do not avoid *E. cloacae* and *K. oxytoca* (see Fig. 2.3b–d). Animals did not avoid the Gram-positive bacteria *E. faecalis* and *S. epidermidis*,

but had a moderate to strong response to *B. megaterium* (see Fig. 2.3e–g). This indicates that *S. marcescens*, though it did not extend lifespan, and *B. megaterium* are mildly pathogenic to animals.

To determine if DBL-1 regulates this avoidance phenotype, we tested avoidance behavior of *dbl-1(-)* exposed to the test bacteria and compared it with *dbl-1(-)* animals fed on the control bacteria. Loss of *dbl-1* usually results in mild but significantly higher avoidance to the control bacteria than the wild type (see Fig. 2.3a), in support of previous findings⁴⁸. *dbl-1(-)* animals displayed a striking, strong avoidance response to all three tested Gram-negative bacteria and a mild response to *S. epidermidis* (see Fig. 2.3b–d, g). Interestingly, *dbl-1(-)* animals exhibited a similar avoidance behavior in response to *B. megaterium* and *E. faecalis* exposure (see Fig. 2.3e, f). This indicates that upon loss of DBL-1, animals perceive *E. cloacae*, *K. oxytoca*, *S. marcescens*, and *S. epidermidis* as more pathogenic. Avoidance responses to *B. megaterium* and *E. faecalis* is independent of DBL-1 levels. These results indicate that DBL-1 normally suppresses avoidance and loss of DBL-1 results in robust avoidance responses that depend on the type of bacterial exposure.



Figure 2.3. Avoidance to Gram-negative bacteria increases upon loss of DBL-1 signaling. Wild-type and *dbl-1(-)* animals at the L4 stage (t = 0 hours) were exposed to the following bacteria: a) *E. coli* OP50 (control), b) *E. cloacae*, c) *K.*

oxytoca, d) S. marcescens, e) B. megaterium, f) E. faecalis, or g) S. epidermidis. Avoidance of animals to the bacteria was monitored over time. The avoidance ratio was calculated and compared between wild-type and *dbl-1(-)* animals. One representative trial of three is presented. Error bars represent standard deviation. n = 30 per condition per trial. * p < 0.05, and ns not significant, compared to wildtype animals exposed to the same bacteria by repeated measures ANOVA using Tukey's post-hoc test.

SMA-4 Acts Independently of Other DBL-1 Core Signaling Components to

Suppress Avoidance to Gram-Positive Bacteria

Because we observed strong bacteria-specific avoidance responses that were DBL-1-dependent, we next asked if the canonical DBL-1 signaling pathway is required to attenuate this response. Canonical signaling occurs by DBL-1 ligand binding to receptors SMA-6 and DAF-4, which activate downstream transcription factors SMA-2, SMA-3, and SMA-4⁴⁹. A non-canonical DBL-1 pathway, which does not signal through SMA-2 and SMA-4, is required for C. elegans to respond to the fungus D. coniospora¹³. We measured the avoidance response of sma-2(-), sma-3(-), and sma-4(-) animals fed on the test Gramnegative or -positive bacteria and compared them with the avoidance response on the control bacteria. On the control bacteria, loss of sma-3 did not result in significantly increased avoidance. However, populations lacking sma-2 avoided the control bacteria mildly to moderately, and populations lacking sma-4 showed a moderate to strong avoidance response (see Fig. 2.4a). sma-2(-), sma-3(-), and sma-4(-) populations displayed a strong, reproducible avoidance response to all Gram-negative bacterial strains tested that was comparable to the dbl-1(-) response (see Fig. 2.4b–d). In comparison, the response of Smad mutant

populations to the panel of Gram-positive bacteria was notably different. Similar to loss of *dbl-1*, loss of *sma-2* or *sma-3* did not increase avoidance responses to the three Gram-positive bacterial strains. However, loss of *sma-4* resulted in moderate to strong avoidance to all three Gram-positive strains (see Fig. 2.4e–g). Our findings indicate that while DBL-1 and SMA-3 do not play a major role in responding to *E. coli*, SMA-2 and SMA-4 are required to suppress avoidance to this standard lab food. Furthermore, our results support that canonical DBL-1 signaling plays a major role in suppressing animal avoidance responses to Gram-negative bacteria, but is not required for responding to Gram-positive bacteria. Interestingly, because of the strong effect loss of SMA-4 has on avoidance behavior, SMA-4 may act with another factor independent of the DBL-1 signaling pathway that heavily influences avoidance responses to Gram-positive bacteria.



Figure 2.4. Avoidance to Gram-negative bacteria increases upon loss of canonical DBL-1 signaling. Wild-type, *sma-2(-)*, *sma-3(-)*, and *sma-4(-)* animals

at the L4 stage (t = 0 hours) were exposed to the following bacteria: a) *E. coli* OP50 (control), b) *E. cloacae*, c) *K. oxytoca*, d) *S. marcescens*, e) *B. megaterium*, f) *E. faecalis*, or g) *S. epidermidis*. Avoidance of animals to the bacteria was monitored over time. The avoidance ratio was calculated and compared between wild-type and Smad mutant animals. Avoidance ratio = number of animals off the bacterial lawn/ total number of animals. One representative trial of three is presented. Error bars represent standard deviation. n = 30 per condition per trial. * *p* < 0.05, ns not significant, compared to wild-type animals exposed to the same bacteria by repeated measures ANOVA using Tukey's post-hoc test.

sma-4 Expression is Specifically Induced in Response to Gram-Positive

Bacteria

We next asked if the different avoidance responses to Gram-negative and -positive bacteria are associated with altered gene expression of the DBL-1 Smads. We tested gene expression levels of *sma-2*, *sma-3*, and *sma-4* in wild-type and *dbl-1* mutant backgrounds in response to our panel of bacteria. With the exception of *S. epidermidis* exposure, the relative levels of *sma-2* mRNA were consistently decreased in the *dbl-1* mutant background, but the test bacterial strains had no effect on *sma-2* expression (see Fig. 2.5a–d). The relative levels of *sma-3* were not reproducibly different between *dbl-1* and wild-type backgrounds, and the overall expression of *sma-3* in animals exposed to the test bacterial strains was not altered. *K. oxytoca* and *E. faecalis* conditions did result in significantly increased *sma-3* levels that were not observed in the *dbl-1* background (see Fig. 2.5e–h). On control and Gram-negative bacteria, *sma-4* expression was similar in the *dbl-1* background as in the wild type. In addition, *sma-4* expression was not changed by exposure to the test Gram-negative

bacteria (see Fig. 2.5i, k). However, *sma-4* was significantly induced in response to all three Gram-positive bacteria in both wild-type and *dbl-1* mutant backgrounds, albeit less in the *dbl-1(-)* populations (see Fig. 2.5j, l). Together, these results suggest that the DBL-1 Smads are differently regulated at the level of gene expression by molecular pathways that are responsive to specific bacterial challenges. *sma-2*, but not *sma-3*, requires DBL-1 for full expression regardless of bacterial food source. Neither *sma-2* nor *sma-3* expression is largely affected by test bacteria. In contrast, *sma-4* expression is responsive to Gram-positive bacteria but not Gram-negative bacteria in the panel, possibly by both DBL-1 and DBL-1-independent mechanisms.



37

■ WT ■ dbl-1(-)

E. coli OP50 Z B. megaterium ⊠ E. faecalis ⊠ S. epidermidis

њ

Figure 2.5. Smad transcription factors gene expression is altered by specific bacteria. Wild-type and *dbl-1(-)* animals at the L4 stage were exposed to *E. coli* OP50 (control), *E. cloacae*, *K. oxytoca*, *S. marcescens*, *B. megaterium*, *E. faecalis*, or *S. epidermidis* for 48 hours. mRNA expression levels of a–d) *sma-2*, e–h) *sma-3*, and i–l) *sma-4* were quantitated by real-time PCR. Experiments were performed in triplicate and in three independent trials. One representative trial of three is presented. Error bars represent standard deviation. * p < 0.05 in a, b, e, f, i, j, mRNA expression level in *dbl-1(-)* population compared to wild-type population exposed to the same bacteria by unpaired *t*-test. * p < 0.05 in c, d, g, h, k, l, mRNA expression level in respective genotype exposed to test bacteria compared to control bacteria by unpaired *t*-test.

DBL-1 Signaling Is Activated in Response to Gram-Negative Bacteria and Is

Repressed in Response to Gram-Positive Bacteria

Because we observed changes in expression levels of the Smads, we asked if DBL-1 signaling activity is altered in response to the bacteria panel. To address this question, we challenged wild-type or *dbl-1(-)* animals expressing an integrated fluorescent DBL-1 pathway reporter to these bacteria and analyzing reporter fluorescence in L2 hypodermal nuclei. The expression of this reporter is robust in hypodermal nuclei at the L2 stage, and changes in DBL-1 affect hypodermal expression of RAD-SMAD³⁹. This reporter (called RAD-SMAD) consists of the GFP gene under the control of multiple copies of a Smad binding element sequence⁵⁰. In *dbl-1(-)* animals on *E. coli*, RAD-SMAD fluorescence was not detectable or very faint. In response to Gram-negative bacteria, reporter fluorescence in the wild-type background was significantly (two- to four-fold) increased compared to control bacterial conditions (see Fig. 2.6a–d, h). Reporter intensity in *dbl-1(-)* animals on all Gram-negative bacteria remained either

undetectable or very faint. In stark contrast, RAD-SMAD hypodermal fluorescence in wild-type animals was lost upon exposure to Gram-positive S. epidermidis (94%, n = 66 had no detectable expression) and *B. megaterium* (100%, n = 42). In about half of the wild-type population fed *E. faecalis*, hypodermal fluorescence was not detected (46%, n = 53), but fluorescence levels were wild type in those animals expressing RAD-SMAD in the hypoderm (see Fig. 2.6e–g, Supplementary Table 4). For animals lacking *dbl*-1 fed on any of the Gram-positive bacterial conditions, hypodermal RAD-SMAD fluorescence was undetectable or very faint. Exposure of *dbl-1(-)* animals to S. epidermidis and *B. megaterium* resulted in a stronger repression of RAD-SMAD activity than exposure to *E. faecalis*, similar to the wild-type responses to these Gram-positive bacteria (see Supplementary Table 4). In general, these results indicate that animals induce a DBL-1 signaling response to Gram-negative bacteria but repress DBL-1 signaling in response to Gram-positive bacteria. Furthermore, DBL-1 signaling levels appear to be modulated depending on the specific bacterial challenge that animals encounter.



Figure 2.6. DBL-1 signaling is activated upon exposure to Gram-negative bacteria but is repressed in response to Gram-positive bacteria. L4 animals expressing the RAD-SMAD reporter in a wild-type background were exposed to a, h) *E. coli* OP50 (control), b, h) *E. cloacae*, c, h) *K. oxytoca*, d, h) *S. marcescens*, e) *B. megaterium*, f) *E. faecalis*, or g) *S. epidermidis* and L2-stage progeny were imaged. Mean RAD-SMAD fluorescence intensity of five hypodermal nuclei per animal was quantitated and compared. Experiments were performed in three independent trials. One representative trial is presented. Error bars represent 95% confidence intervals. n = 15 per condition in each trial. * p < 0.05, mean fluorescence intensity in wild-type background on test bacteria compared to control bacteria by unpaired *t*-test. Scale bar, 10 µm.

DBL-1 Mediates Both Common and Specific Gene Expression Responses

to Gram-Negative and -Positive Bacteria

We then asked if this differential modulation of DBL-1 signaling activity

translated to bacterial-specific downstream transcriptional responses. To identify

the role of DBL-1 in differentially regulating transcription of downstream genes,

we performed RNA sequencing using wild-type and dbl-1(-) animals exposed to

the control or test bacteria (Gram-negative S. marcescens or Gram-positive E.

faecalis). The animals were synchronized as L4s and fed on the control or test

bacteria for 48 hours before analysis. In animals lacking DBL-1 and fed control bacteria, 83 genes were down-regulated and 49 genes were upregulated compared to the wild type (p < 0.01, see Supplementary Fig. 2a). Some genes and gene classes previously reported to be regulated by DBL-1 at different developmental stages were also regulated by DBL-1 in two-day adults^{20,24,25,51}. In dbl-1(-) animals fed on S. marcescens, 102 genes were down-regulated and 117 genes were upregulated (see Supplementary Fig. 2b). In *dbl-1(-)* animals fed on *E. faecalis*, 63 genes were down-regulated and 64 genes were upregulated compared to the wild type (see Supplementary Fig. 2c). The lower number of highly regulated genes between wild-type and *dbl-1(-)* animals fed on *E. faecalis* is consistent with the reduced DBL-1 reporter fluorescence—and therefore DBL-1 pathway signaling—in wild-type animals on *E. faecalis* (see Fig. 2.6, Supplementary Fig. 2). Notably, some highly regulated genes were common in response to both pathogenic bacterial strains, but some genes that were differentially regulated by DBL-1 were unique in response to either S. marcescens or *E. faecalis* exposure. Using WormCat, gene enrichment analysis of genes regulated by DBL-1 in response to S. marcescens or E. faecalis exposure revealed differential regulation of sets of genes involved in pathogen response, stress response, lipid metabolism, and transmembrane transport (see Supplementary Data 1)⁵².

We focused on the DBL-1-regulated genes that have known or putative roles in innate immunity. These genes were induced in wild-type animals upon

exposure to *S. marcescens* or *E. faecalis* and this induction was lost in *dbl-1(-)*. We also found some genes to be upregulated upon loss of *dbl-1* in response to *S. marcescens* or *E. faecalis*. Some gene classes are highly regulated in response to these bacteria, but the specific genes within these families differed, including lysozyme, aspartyl protease, saposin-like, and C-type lectin genes (see Fig. 2.7). These results suggest that DBL-1 is involved in regulating (positively and negatively) transcription of some innate immunity genes specific to the bacterial exposure and some genes that are commonly regulated upon exposure to different bacteria. This supports a role for DBL-1 signaling in differentially regulating host responses to bacteria.



Figure 2.7. Expression of innate immunity genes is differentially regulated by DBL-1 signaling in different bacterial environments. L4 wild-type and *dbl-1(-)* animals were exposed for two days to *E. coli* OP50 (control), *S. marcescens*, or *E. faecalis*. RNA-seq analysis reveal significant changes (adjusted *p*-value < 0.01) in gene expression of animals lacking DBL-1. Heatmaps show differential innate immunity gene expression in animals lacking DBL-1 exposed to a) *E. coli* OP50, b) *S. marcescens*, and c) *E. faecalis* in comparison to wild-type animals exposed to the same bacteria. Average log FPKM values from three independent trials are represented.

DBL-1 Differentially Regulates Expression of Innate Immunity Genes Specific to the Gram Nature of Bacteria

From the RNA-sequencing results, we identified a panel of DBL-1 responsive innate immune genes that displayed differential responses to pathogen exposure. To determine if the expression of candidate target immunity genes is regulated by DBL-1 signaling in response to a wider variety of bacterial exposures, we used reporters of select immunity-related genes. Based on the RNA sequencing results, we selected *dod-22*, F55G11.7, *irg-4*, and *dod-24*. We also selected *ilys-3*, a known Gram-positive-responsive gene⁹. We tested expression of transcriptional reporters of these genes upon exposure to different Gram-negative and -positive bacteria. We measured and compared expression of these reporter genes in wild-type and *dbl-1(-)* backgrounds exposed to control or test bacteria. Animals were synchronized as L4s and fed on control, Gram-negative, or Gram-positive bacteria for 48 hours. Basal expression of these selected genes was measured in animals fed on the control *E. coli*.

dod-22 is a gene that is known to be regulated by the insulin-like signaling pathway transcription factor, DAF-16²¹. It is known to be involved in defense response to Gram-negative bacteria¹⁶. The *dod-22* reporter is induced in the wild-type background in the presence of all Gram-negative test bacteria compared to the control, but is not induced in response to the panel of Gram-positive bacteria. This induction of the *dod-22* reporter on Gram-negative bacteria is partly lost in the *dbl-1(-)* background, though loss of *dbl-1* does not affect expression levels on the

control *E. coli. B. megaterium* does not lead to a reproducible reduction of *dod-22* reporter fluorescence in the wild-type background, but a further reduction of fluorescence is observed in the *dbl-1(-)* background (see Fig. 2.8a, b). These results indicate that DBL-1 is not required for the basal expression of *dod-22*, but is required for *dod-22* induction on Gram-negative bacteria.

F55G11.7 is involved in innate immune responses to both Gram-negative and -positive bacteria in *C. elegans* and has been shown to be regulated by DAF-16/insulin, MAPK, and DBL-1 signaling pathways¹⁶. Expression of the F55G11.7 reporter did not change upon exposure to all test bacteria in the wildtype background except *B. megaterium*, where reporter expression was reduced (in two of three trials). We observed a significant reduction of F55G11.7 reporter activity in animals lacking DBL-1 except in response to *E. faecalis*. Additionally, a further reduction of F55G11.7 reporter fluorescence in *dbl-1(-)* animals was observed in response to Gram-positive *B. megaterium* and *S. epidermidis* compared to the response on control bacteria (see Fig. 2.8c, d). These findings indicate that expression of F55G11.7 is not altered upon exposure to most of the test bacteria but is generally down-regulated upon loss of DBL-1, in contrast with previous findings¹⁶.

irg-4 is known to be involved in defense response to Gram-negative bacteria and has been shown to be regulated by DAF-16/insulin, MAPK, and DBL-1 signaling pathways^{53–56}. In the wild-type background, we observed a visible, reproducible induction of *irg-4* reporter activity in response to Gram-

negative *S. marcescens*, but not to *K. oxytoca* or *E. cloacae*. Expression of the *irg-4* reporter in wild-type animals mildly increased in response to *S. epidermidis*, but not the other Gram-positive bacteria reproducibly. Loss of DBL-1 did not reproducibly alter *irg-4* reporter activity in control conditions. However, *irg-4* reporter induction in response to Gram-negative *S. marcescens* and Gram-positive *S. epidermidis* was lost in the *dbl-1(-)* background (see Fig. 2.8e, f). Indeed, *irg-4* reporter expression was reduced compared to the wild-type background on all bacteria except *E. cloacae* and *E. faecalis*. These results indicate that *irg-4* is responsive to a broad range of bacteria and is regulated in part by DBL-1 signaling.

dod-24, which is regulated by the insulin-like signaling transcription factor DAF-16, is involved in defense response to Gram-negative bacteria⁵³. We observed robust expression of *dod-24* reporter activity in all tested Gram-negative bacteria, including the control, *E. cloacae, K. oxytoca*, and *S. marcescens*. Additionally, we observed further induction in response to *E. cloacae* and *S. marcescens*. We observed a striking decrease of *dod-24* reporter activity in wild-type animals exposed to all tested Gram-positive bacteria including *B. megaterium, E. faecalis*, and *S. epidermidis*. Loss of DBL-1 resulted in a significant reduction of *dod-24* reporter activity in control conditions. *dod-24* reporter activity was also drastically reduced in all Gram-negative bacterial conditions to levels significantly lower than the wild type (see Fig. 2.8g). In the three Gram-positive conditions, loss of DBL-1 resulted in a further decrease of

dod-24 reporter fluorescence relative to the wild type (see Fig. 2.8h). These results confirm that *dod-24* is differentially expressed in response to an array of Gram-negative and -positive bacteria and indicate that DBL-1 signaling plays a major role in regulating *dod-24* expression.

ilys-3 exhibits lysozyme activity and is involved in responding to Grampositive bacteria⁹. *ilys-3* reporter activity in wild-type animals was unchanged or reduced in response to the tested Gram-negative bacteria (see Fig. 2.8i). We observed induction of *ilys-3* reporter activity upon exposure to all tested Grampositive bacteria including *B. megaterium*, *E. faecalis*, and *S. epidermidis* (see Fig. 2.8j). Loss of DBL-1 did not alter *ilys-3* reporter activity in control conditions. The *ilys-3* reporter activity remained at relatively low levels in animals lacking DBL-1 exposed to Gram-negative bacteria (see Fig. 2.8i). However, *ilys-3* reporter activity also remained at relatively low levels upon loss of DBL-1 in response to Gram-positive bacteria *B. megaterium* and *S. epidermidis*, but was wild type in response to *E. faecalis* (see Fig. 2.8j). These results suggest that while DBL-1 is not required for basal levels of *ilys-3* expression, it is required for the induction of *ilys-3* expression in response to some Gram-positive bacteria.





Figure 2.8. Innate immune reporter activity is regulated by exposure to specific bacteria and by DBL-1 signaling. Comparison of a, b) *dod-22::*GFP, c, d) F55G11.7::GFP, e, f) *irg-4::*GFP, g, h) *dod-24p::*GFP, and i, j) *ilys-3p::*GFP intensities in adult wild-type and *dbl-1(-)* animals after a two-day exposure to the following bacteria; control *E. coli* OP50, *E. cloacae*, *K.* oxytoca, *S. marcescens*, *B. megaterium*, *E. faecalis*, or *S. epidermidis*. Imaging conditions including exposure times were consistent with respective control. Three independent trials were performed. One representative trial is shown. Error bars represent standard deviation. *n* = at least 15 per condition in each trial. * *p* < 0.05 compared to wild-type animals exposed to the same bacteria and # *p* < 0.05 respective genotype exposed to test bacteria in comparison to control bacteria, by two-way ANOVA followed by unpaired *t*-test.

These results collectively indicate the specific role of DBL-1 signaling in antimicrobial gene expression that helps tailor defense responses to specific bacterial challenges.

Discussion

Animals are subjected to a range of bacterial challenges, and how they respond is critical to the animals' health. Understanding how hosts respond to different pathogens is important for developing therapeutic strategies to help fight infections and prevent diseases. Dissecting the cell signaling pathways involved in host responses and their roles is critical. Our work expands the current understanding of how an organism integrates an arsenal of responses—from the molecular to the organismal—to different bacteria, and identifies role of the DBL- $1/TGF-\beta$ pathway in robust host-specific responses to different types of bacteria. For this work, we established a panel of human opportunistic pathogens, including three Gram-negative and three Gram-positive strains, for the study of long-term innate immune responses in the roundworm *C. elegans*. The bacteria we selected for the panel elicit unique host response patterns that allowed us to

interrogate the role of the DBL-1 pathway in responding to different bacterial exposures. While DBL-1 has a known role in transcriptionally regulating innate immune gene expression, we show that the specific responses mediated by DBL-1 are not only molecular, but are also behavioral.

Our results support a model that the DBL-1 signaling pathway influences the organisms' perception of pathogenicity and helps keep the innate immune responses in check. Animals with reduced DBL-1 signaling—whether by downregulating signaling or by mutation—perceive the environment as more threatening and respond accordingly. DBL-1 pathway mutants display an outsized avoidance response to the Gram-negative bacteria. These animals also reduce intake of select Gram-negative bacteria as yet another way to reduce animals' interaction with the threat. Animals lacking DBL-1 also display a reduced lifespan in response to the Gram-negative bacteria that are sensed to be more pathogenic. There is also a correlation between the reduced feeding and extension of lifespan observed in animals exposed to select bacteria. Because exposure to the Gram-positive bacteria reduce the DBL-1 signaling activity, the avoidance response did not alter dramatically upon loss of DBL-1. The intake of the Gram-positive bacteria was reduced in both wild-type and *dbl-1(-)* populations, which correlated with extended lifespan observed for both populations on the Gram-positive bacteria. Damage to the intestine was not the underlying cause for the DBL-1-mediated lifespan alterations in response to the

tested bacteria. Determining how DBL-1 is involved in such behavioral modifications in response to different bacteria warrants future investigation.

DBL-1 signaling is also important for molecular responses to both Gram negative and Gram-positive bacteria. Regulation of DBL-1 signaling is part of the host's molecular response: the Gram-negative bacteria of our panel induced DBL-1 signaling while the Gram-positive bacteria repressed DBL-1 signaling. However, within these two bacterial groups, the host responses were tailored to the specific bacterial challenge. Our results show that DBL-1 signaling is also an important part of this molecular antimicrobial "fingerprint"¹⁶. Our RNA-sequencing analyses indicate both common and unique transcriptome-wide alterations mediated by DBL-1 after two days of exposure to Gram-negative and Grampositive bacteria. We find that DBL-1 signaling is involved in activating as well as repressing innate immunity genes to maintain a balance of host immune responses (to possibly avoid overactivation of host immunity). In response to our panel of Gram-negative bacteria, DBL-1 signaling activity was induced and it further regulated expression of unique downstream innate immunity genes. In contrast, even though the DBL-1 signaling activity was repressed in response to the tested Gram-positive bacteria, DBL-1 was required to regulate expression of target immunity genes. While many gene classes differentially regulated by specific pathogens have been previously identified as important innate immune response genes, our work highlights the role that DBL-1 plays in tailoring the

molecular responses *C. elegans* engages against a range of pathogens^{9,11,16,53,54,57}.

Another major finding of this work is the differential requirement of the SMAD machinery to mediate avoidance responses to our panel of bacteria. Olofsson previously showed that loss of DBL-1, SMA-2, or SMA-4 increases avoidance of *E. coli*⁴⁸. While our results with sma-2 and sma-4 are similar to theirs, we observe only mild avoidance of *E. coli* upon loss of DBL-1⁴⁸. However, our work demonstrates that canonical DBL-1 signaling strongly suppresses avoidance to Gram-negative bacteria, but not to Gram-positive bacteria (see Fig. 2.3). *sma-4* mutants generally displayed stronger avoidance responses to both control and test bacteria suggesting a DBL-1-independent role for SMA-4 in suppressing avoidance responses. SMA-4 appears to play a double role in innate immune responses. Its starring role in immunity is with the DBL-1 pathway. DBL-1-independent induction of *sma-4* in response to Gram-positive bacterial conditions was also observed. This specificity of sma-4 induction by the Grampositive bacteria indicates specificity of innate immune responses. These results indicate that SMA-4 is not only recruited for defenses by something other than DBL-1, but also acts independently of the core DBL-1 pathway. PMK-1/MAPK signaling may be involved in regulating it as SMA-4 is predicted to genetically interact with PMK-1⁵⁸. Interestingly, ATF-7, a transcription factor activated by PMK-1, is required for downregulation of *sma-4*—but not other DBL-1 pathway component genes—in wild-type animals exposed to Gram-negative P.

aeruginosa PA14⁵⁹. It will be of interest to discover if SMA-4 plays a broader role in the innate immune response than acting in the DBL-1 pathway.

Overall, we propose that loss of DBL-1 signaling changes the animal's perception of the environment as more hostile, and this results in more robust protective responses that depend on the specific bacterial challenge. This may help explain the neuronal source of DBL-1 secretion that then targets hypoderm, intestine, and pharynx. DBL-1 secreted from the AVA interneurons activates DBL-1 signaling in the hypodermis to regulate aversive learning upon exposure to Gram-negative *P. aeruginosa*, but the neuronal circuit(s) used by DBL-1 to direct aversive behaviors remains to be identified²⁷. SMA-4 plays a double role in innate immune responses, acting as part of the core DBL-1 signaling pathway but also acting in another additive way, suggesting crosstalk with other signaling pathways. Several studies report roles of signaling pathways including MAPK and insulin-like signaling pathways in mounting immune responses to many Gram-positive bacteria. Future work may identify the possible crosstalk mechanisms with the DBL-1 pathway in regulating organismal defense responses. In summary, these findings support a central role for DBL-1/TGF- β signaling not only in crafting tailored responses to unique bacterial challenges, from transcription of specific innate immunity genes to behavioral responses but also being modulated in response to bacteria.

Acknowledgements

We thank M. Farhan Lakdawala and Rosylin Roy for technical assistance. We thank James Lundgren, Paul Yeatts, and TWU's Center for Research Design and Analysis for assistance with statistical analyses. Some bacterial strains were provided by Amy Jo Hammett. Some strains were obtained from the *Caenorhabditis* Genetics Center (CGC), which is funded by NIH Office of Research Infrastructure Programs (P40 OD010440). We thank WormBase. We thank Laura Hanson and Gumienny lab members for constructive feedback. Novogene provided Supplementary Figure 2. This work was supported by NIH R01GM097591 to TLG, TWU Research Enhancement Program funding to T.L.G., internal funding by Texas Woman's University to T.L.G., and TWU Experiential Learning Scholar Awards to B.M.

Author Contributions

B.M. and T.L.G. conceptualized and designed experiments, interpreted data, and contributed to writing; B.M. performed experiments.

Corresponding Author

Correspondence to tgumienny@twu.edu

Competing Interests

The authors declare no competing interests.

References

- Medzhitov, R. & Janeway, C. A., Jr. Innate immune recognition and control of adaptive immune responses. *Semin Immunol* 10, 351–353, https://doi.org/10.1006/smim.1998.0136 (1998).
- 2 MacGillivray, D. M. & Kollmann, T. R. The role of environmental factors in modulating immune responses in early life. *Front Immunol* 5, 434, https://doi.org/10.3389/fimmu.2014.00434 (2014).
- Cheesman, H. K. et al. Aberrant activation of p38 MAP kinase-dependent innate immune responses is toxic to *Caenorhabditis elegans*. *G3* (*Bethesda*) 6, 541–549, https://doi.org/10.1534/g3.115.025650 (2016).
- Pukkila-Worley, R. Surveillance immunity: An emerging paradigm of innate defense activation in *Caenorhabditis elegans*. *PLoS Pathog* 12, e1005795, https://doi.org/10.1371/journal.ppat.1005795 (2016).
- Sellegounder, D., Yuan, C. H., Wibisono, P., Liu, Y. & Sun, J.
 Octopaminergic signaling mediates neural regulation of innate immunity in *Caenorhabditis elegans. mBio* 9, https://doi.org/10.1128/mBio.01645-18 (2018).
- Akira, S., Uematsu, S. & Takeuchi, O. Pathogen recognition and innate immunity. *Cell* 124, 783–801, https://doi.org/10.1016/j.cell.2006.02.015 (2006).
- 7 Engelmann, I. & Pujol, N. Innate immunity in *C. elegans. Adv Exp Med Biol* 708, 105–121, https://doi.org/10.1007/978-1-4419-8059-5_6 (2010).

- Couillault, C. & Ewbank, J. J. Diverse bacteria are pathogens of *Caenorhabditis elegans*. *Infect Immun* 70, 4705–4707, https://doi.org/10.1128/iai.70.8.4705-4707.2002 (2002).
- Gravato-Nobre, M. J. et al. Multiple genes affect sensitivity of
 Caenorhabditis elegans to the bacterial pathogen *Microbacterium nematophilum. Genetics* 171, 1033–1045,
 https://doi.org/10.1534/genetics.105.045716 (2005).
- Begun, J. et al. Staphylococcal biofilm exopolysaccharide protects against
 Caenorhabditis elegans immune defenses. *PLoS Pathog* 3, e57,
 https://doi.org/10.1371/journal.ppat.0030057 (2007).
- 11 Wong, D., Bazopoulou, D., Pujol, N., Tavernarakis, N. & Ewbank, J. J. Genome-wide investigation reveals pathogen-specific and shared signatures in the response of *Caenorhabditis elegans* to infection. *Genome Biol* 8, R194, https://doi.org/10.1186/gb-2007-8-9-r194 (2007).
- Singh, V. & Aballay, A. Regulation of DAF-16-mediated Innate Immunity in Caenorhabditis elegans. J Biol Chem 284, 35580–35587, https://doi.org/10.1074/jbc.M109.060905 (2009).
- 13 Zugasti, O. & Ewbank, J. J. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. *Nat Immunol* **10**, 249–256, https://doi.org/10.1038/ni.1700 (2009).

- 14 Pukkila-Worley, R. et al. Stimulation of host immune defenses by a small molecule protects *C. elegans* from bacterial infection. *PLoS Genet* 8, e1002733, https://doi.org/10.1371/journal.pgen.1002733 (2012).
- Ahamefule, C. S. et al. *Caenorhabditis elegans*-based *Aspergillus fumigatus* infection model for evaluating pathogenicity and drug efficacy.
 Front Cell Infect Microbiol **10**, 320, https://doi.org/10.3389/fcimb.2020.00320 (2020).
- 16 Alper, S., McBride, S. J., Lackford, B., Freedman, J. H. & Schwartz, D. A. Specificity and complexity of the *Caenorhabditis elegans* innate immune response. *Mol Cell Biol* 27, 5544–5553, https://doi.org/10.1128/MCB.02070-06 (2007).
- Berg, M. et al. TGFβ/BMP immune signaling affects abundance and function of *C. elegans* gut commensals. *Nat Commun* 10, 604, https://doi.org/10.1038/s41467-019-08379-8 (2019).
- Pujol, N. et al. A reverse genetic analysis of components of the Toll signaling pathway in *Caenorhabditis elegans*. *Curr Biol* 11, 809–821, https://doi.org/10.1016/s0960-9822(01)00241-x (2001).
- Kim, D. H. et al. A conserved p38 MAP kinase pathway in *Caenorhabditis* elegans innate immunity. *Science* 297, 623–626, https://doi.org/10.1126/science.1073759 (2002).
- Mallo, G. V. et al. Inducible antibacterial defense system in *C. elegans*.
 Curr Biol **12**, 1209–1214, https://doi.org/10.1016/s0960-9822(02)00928-4
 (2002).
- Murphy, C. T. et al. Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans*. *Nature* 424, 277–283, https://doi.org/10.1038/nature01789 (2003).
- Garsin, D. A. et al. Long-lived *C. elegans daf-2* mutants are resistant to bacterial pathogens. *Science* 300, 1921, https://doi.org/10.1126/science.1080147 (2003).
- Mochii, M., Yoshida, S., Morita, K., Kohara, Y. & Ueno, N. Identification of transforming growth factor-β regulated genes in *Caenorhabditis elegans* by differential hybridization of arrayed cDNAs. *Proc Natl Acad Sci U S A* 96, 15020–15025, https://doi.org/10.1073/pnas.96.26.15020 (1999).
- Liang, J., Yu, L., Yin, J. & Savage-Dunn, C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related signaling outputs. *Dev Biol* 305, 714–725, https://doi.org/10.1016/j.ydbio.2007.02.038 (2007).
- Roberts, A. F., Gumienny, T. L., Gleason, R. J., Wang, H. & Padgett, R.
 W. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. *BMC Dev Biol* 10, 61, https://doi.org/10.1186/1471-213X-10-61 (2010).

- Tan, M. W., Mahajan-Miklos, S. & Ausubel, F. M. Killing of *Caenorhabditis* elegans by *Pseudomonas aeruginosa* used to model mammalian bacterial pathogenesis. *Proc Natl Acad Sci U S A* 96, 715–720, https://doi.org/10.1073/pnas.96.2.715 (1999).
- Zhang, X. & Zhang, Y. DBL-1, a TGF-β, is essential for *Caenorhabditis elegans* aversive olfactory learning. *Proc Natl Acad Sci U S A* 109, 17081–17086, https://doi.org/10.1073/pnas.1205982109 (2012).
- 28 Madhu, B., Salazar, A. E. & Gumienny, T. L. *Caenorhabditis elegans* egglaying and brood-size changes upon exposure to *Serratia marcescens* and *Staphylococcus epidermidis* are independent of DBL-1 signaling. *microPublication Biology* 2019, 10.17912/2r51-b476 (2019).
- 29 Sifri, C. D., Begun, J., Ausubel, F. M. & Calderwood, S. B. *Caenorhabditis elegans* as a model host for *Staphylococcus aureus* pathogenesis. *Infect Immun* **71**, 2208–2217, https://doi.org/10.1128/iai.71.4.2208-2217.2003 (2003).
- 30 Reddy, K. C., Andersen, E. C., Kruglyak, L. & Kim, D. H. A polymorphism in *npr-1* is a behavioral determinant of pathogen susceptibility in *C. elegans. Science* **323**, 382–384, https://doi.org/10.1126/science.1166527 (2009).
- Amrit, F. R., Ratnappan, R., Keith, S. A. & Ghazi, A. The *C. elegans*lifespan assay toolkit. *Methods* 68, 465–475,
 https://doi.org/10.1016/j.ymeth.2014.04.002 (2014).

- 32 Clark, J. F., Meade, M., Ranepura, G., Hall, D. H. & Savage-Dunn, C. *Caenorhabditis elegans* DBL-1/BMP regulates lipid accumulation via interaction with insulin signaling. *G3 (Bethesda)* 8, 343–351, https://doi.org/10.1534/g3.117.300416 (2018).
- Kissoyan, K. A. B. et al. Natural *C. elegans* microbiota protects against infection via production of a cyclic lipopeptide of the viscosin group. *Curr Biol* 29, 1030–1037 e1035, https://doi.org/10.1016/j.cub.2019.01.050 (2019).
- Chang, H. C., Paek, J. & Kim, D. H. Natural polymorphisms in *C. elegans*HECW-1 E3 ligase affect pathogen avoidance behaviour. *Nature* 480,
 525–529, https://doi.org/10.1038/nature10643 (2011).
- 35 Stiernagle, T. Maintenance of *C. elegans. WormBook, The C. elegans Research Community* (2006).
- 36 Portman, D. S. Profiling *C. elegans* gene expression with DNA microarrays. *WormBook, The C. elegans Research Community* (2006).
- Love, M. I., Huber, W. & Anders, S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. *Genome Biol* 15, 550, https://doi.org/10.1186/s13059-014-0550-8 (2014).
- Madhu, B., Lakdawala, M. F., Issac, N. G. & Gumienny, T. L.
 Caenorhabditis elegans saposin-like *spp-9* is involved in specific innate immune responses. *Genes Immun* 21, 301–310, https://doi.org/10.1038/s41435-020-0108-6 (2020).

- 39 Savage-Dunn, C., Gleason, R. J., Liu, J. & Padgett, R. W. Mutagenesis and imaging studies of BMP signaling mechanisms in *C. elegans*. *Methods Mol Biol* **1891**, 51–73, https://doi.org/10.1007/978-1-4939-8904-1_6 (2019).
- Samuel, B. S., Rowedder, H., Braendle, C., Felix, M. A. & Ruvkun, G.
 Caenorhabditis elegans responses to bacteria from its natural habitats.
 Proc Natl Acad Sci U S A 113, E3941–3949,
 https://doi.org/10.1073/pnas.1607183113 (2016).
- Tenor, J. L. & Aballay, A. A conserved Toll-like receptor is required for *Caenorhabditis elegans* innate immunity. *EMBO Rep* 9, 103–109, https://doi.org/10.1038/sj.embor.7401104 (2008).
- Lakowski, B. & Hekimi, S. The genetics of caloric restriction in
 Caenorhabditis elegans. Proc Natl Acad Sci U S A 95, 13091–13096, https://doi.org/10.1073/pnas.95.22.13091 (1998).
- Greer, E. L. & Brunet, A. Different dietary restriction regimens extend lifespan by both independent and overlapping genetic pathways in *C. elegans. Aging Cell* 8, 113–127, https://doi.org/10.1111/j.1474-9726.2009.00459.x (2009).
- Gelino, S. et al. Intestinal autophagy improves healthspan and longevity in
 C. elegans during dietary restriction. *PLoS Genet* 12, e1006135,
 https://doi.org/10.1371/journal.pgen.1006135 (2016).

- Pradel, E. et al. Detection and avoidance of a natural product from the pathogenic bacterium *Serratia marcescens* by *Caenorhabditis elegans*.
 Proc Natl Acad Sci U S A **104**, 2295–2300, https://doi.org/10.1073/pnas.0610281104 (2007).
- Beale, E., Li, G., Tan, M. W. & Rumbaugh, K. P. *Caenorhabditis elegans* senses bacterial autoinducers. *Appl Environ Microbiol* 72, 5135–5137, https://doi.org/10.1128/AEM.00611-06 (2006).
- 47 Anderson, A. & McMullan, R. Neuronal and non-neuronal signals regulate *Caernorhabditis elegans* avoidance of contaminated food. *Philos Trans R Soc Lond B Biol Sci* **373**, https://doi.org/10.1098/rstb.2017.0255 (2018).
- Olofsson, B. The olfactory neuron AWC promotes avoidance of normally palatable food following chronic dietary restriction. *J Exp Biol* 217, 1790–1798, https://doi.org/10.1242/jeb.099929 (2014).
- 49 Savage, C. et al. *Caenorhabditis elegans* genes *sma-2*, *sma-3*, and *sma-4* define a conserved family of transforming growth factor beta pathway components. *Proc Natl Acad Sci U S A* **93**, 790–794, https://doi.org/10.1073/pnas.93.2.790 (1996).
- Tian, C. et al. The RGM protein DRAG-1 positively regulates a BMP-like signaling pathway in *Caenorhabditis elegans*. *Development* 137, 2375–2384, https://doi.org/10.1242/dev.051615 (2010).
- 51 Lakdawala, M. L. et al. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body size-associated genes in *Caenorhabditis elegans*.

Molecular Biology of the Cell **30**(26), 3151–3160, https://doi.org/10.1091/mbc.E19-09-0500 (2019).

- Holdorf, A. D. et al. WormCat: An online tool for annotation and visualization of *Caenorhabditis elegans* genome-scale data. *Genetics* 214, 279–294, https://doi.org/10.1534/genetics.119.302919 (2020).
- Shapira, M. et al. A conserved role for a GATA transcription factor in regulating epithelial innate immune responses. *Proc Natl Acad Sci U S A*103, 14086–14091, https://doi.org/10.1073/pnas.0603424103 (2006).
- 54 Troemel, E. R. et al. p38 MAPK regulates expression of immune response genes and contributes to longevity in *C. elegans. PLoS Genet* 2, e183, https://doi.org/10.1371/journal.pgen.0020183 (2006).
- 55 Nandakumar, M. & Tan, M. W. Gamma-linolenic and stearidonic acids are required for basal immunity in *Caenorhabditis elegans* through their effects on p38 MAP kinase activity. *PLoS Genet* **4**, e1000273, https://doi.org/10.1371/journal.pgen.1000273 (2008).
- 56 Peterson, N. D. et al. The nuclear hormone receptor NHR-86 controls antipathogen responses in *C. elegans. PLoS Genet* **15**, e1007935, https://doi.org/10.1371/journal.pgen.1007935 (2019).
- 57 Engelmann, I. et al. A comprehensive analysis of gene expression changes provoked by bacterial and fungal infection in *C. elegans. PLoS One* **6**, e19055, https://doi.org/10.1371/journal.pone.0019055 (2011).

- Zhong, W. & Sternberg, P. W. Genome-wide prediction of *C. elegans* genetic interactions. *Science* 311, 1481–1484, https://doi.org/10.1126/science.1123287 (2006).
- 59 Fletcher, M., Tillman, E. J., Butty, V. L., Levine, S. S. & Kim, D. H. Global transcriptional regulation of innate immunity by ATF-7 in *C. elegans. PLoS Genet* 15, e1007830, https://doi.org/10.1371/journal.pgen.1007830 (2019).

Supplementary Material

Supplementary Table 1. List of strains

Strains used include:

Strain	Genotype					
N2	Wild type					
NU3	<i>dbl-1(nk3)</i> V (referred to as <i>dbl-1(-)</i> in this work)					
CB502	sma-2(e502) III (referred to as sma-2(-) in this work)					
CB491	sma-3(e502) III (referred to as sma-3(-) in this work)					
DR1369	sma-4(e729) III (referred to as sma-4(-) in this work)					
LW2436	jjls2277[pCXT51(5*RLR::pes-10p(deleted)::GFP) +					
	LiuFD61(mec-7p::RFP)] I or IV (RAD-SMAD)					
CB6710	eEx650[ilys-3p::GFP + unc-119(+)]					
CF3556	agls6[dod-24p::GFP]					
SAL139	denEx17[dod-22::GFP + unc-119(+)]					
SAL143	denEx21[F55G11.7::GFP + unc-119(+)]					
SAL148	denEx26[irg-4::GFP + unc-119(+)]					

Strains created for this work include:

Strain	Genotype
TLG803	dbl-1(nk3) V; agls6[dod-24p::GFP]
TLG804	dbl-1(nk3) V; eEx650[ilys-3p::GFP + unc-119(+)]
TLG805	dbl-1(nk3) V;
TLG806	dbl-1(nk3) V; denEx21[F55G11.7::GFP + unc-119(+)]
TLG807	dbl-1(nk3) V; denEx17[dod-22::GFP + unc-119(+)]
TLG810	jjls2277[pCXT51(5*RLR::pes-10p(deleted)::GFP) +
	<i>LiuFD61(mec-7p::RFP)]</i> I or IV; <i>dbl-1(nk3)</i> V

Supplementary Table 2. List of primers for qRT-PCR

Target		
gene	Forward primer (5'->3')	Reverse primer (5'->3')
sma-2	TCCACCAGGAGTTCCAACAT	ACCTGTTCTCCGACTCTTGT
sma-3	GAGAACACACGGATGCATATTGG	ACTGTGCGGTGGTATTCGG
sma-4	GATGCTCCGACGTTCTCGAT	CGCATCCTGTCAACTCCACT
act-1	GCCGGAATCCACGAGACTTC	TCTGGTGGGGCGATGATCTT

Supplementary Table 3. Survival assay summary													
Bacteria	Nematode strain	Trial 1			Trial 2			Trial 3					
		test mean lifespan ± SE	n	control mean lifespan ± SE	n	test mean lifespan ± SE	n	control mean lifespan ± SE	n	test mean lifespan ± SE	n	control mean lifespan ± SE	n
	WT	16.22±0.39	104	12.3±0.37	110	15.58±0.30	105	14.98±0.35	102	16.68±0.52	71	13.84±0.40	75
E. cloacae	dbl-1(-)	11.64±0.33 *	105	11.87±0.38	111	13.67±0.41 ns	112	14.36±0.36	104	11.66±0.47 *	58	14.17±0.40	58
K. oxytoca	WT	14.34±0.31	120	12.3±0.37	110	16.16±0.27	110	14.98±0.35	102	16.41±0.30	75	13.84±0.40	75
	dbl-1(-)	12.55±0.32 *	108	11.87±0.38	111	14.87±0.31 ns	100	14.36±0.36	104	9.31±0.55 *	62	14.17±0.40	58
S. maraaaaaaa	WT	15.09±0.28	107	15.19±0.23	111	12.40±0.38	115	12.3±0.37	110	13.43±0.39	113	14.98±0.35	102
S. Marcescens	dbl-1(-)	8.56±0.25 *	111	14.58±0.22	97	6.58±0.29 *	108	11.87±0.38	111	6.21±0.30 *	105	14.36±0.36	104
P. mogotorium	WT	16.60±0.21	101	13.44±0.21	115	16.54±0.21	84	14.06±0.24	95	12.86±0.46	63	9.07±0.55	71
B. megaterium	dbl-1(-)	13.82±0.19 *	119	14.2±0.18	116	15.25±0.22 *	84	13.99±0.22	84	13.47±0.27 ns	73	8.61±0.37	77
E. faecalis	WT	17.77±0.28	121	13.82±0.25	121	16.54±0.29	119	13.20±0.22	123	17.46±0.32	129	13.32±0.24	121
	dbl-1(-)	15.83±0.34 *	75	13.20±0.29	99	13.25±0.39 *	142	13.72±0.28	129	14.97±0.33 *	126	13.58±0.23	122
S opidormidis	WT	17.92±0.21	122	13.82±0.25	121	15.97±0.34	104	13.20±0.22	123	16.60±0.20	115	13.32±0.24	121
S. epidermidis	dbl-1(-)	8.84±0.42 *	138	13.20±0.29	99	11.88±0.35 *	138	13.72±0.28	129	13.32±0.39 *	143	13.58±0.23	122
* log rank test p<0.0001, ns log rank test p>0.01, comparing mean dbl-1(-) lifespan with wild-type lifespan on each bacteria													

Supplementary Table 3. Survival assay summary

Supplementary Table 4. RAD-SMAD reporter activity in response to Gram-

positive bacteria

Bacteria	Nematode strain	% animals with no detectable fluorescence				
		Trial 1	Trial 2	Trial 3		
B. megaterium	WT	100%	100%	100%		
	dbl-1(-)	100%	100%	100%		
E. faecalis	WT	95%	22%	21%		
	dbl-1(-)	62%	40%	65%		
S. epidermidis	WT	95%	95%	91%		
	dbl-1(-)	100%	95%	78%		

Supplementary Data 1. Gene enrichment analysis by WormCat of differentially expressed genes between wild-type and *dbl-1(-)* populations exposed to a) *S. marcescens* and b) *E. faecalis*

a) <u>http://www.wormcat.com/static/dynamic/RGS_Mar-25-2021-05_56_38/sunburst.html</u>

b) <u>http://www.wormcat.com/static/dynamic/RGS_Mar-25-2021-05_58_42/sunburst.html</u>



W UDI

W abbin



W dol'



W BDI'

W UDI'







W dol-

W abbin

Severe smurf

W dol-1

Day 11

W dol-1





а

Supplementary Figure 1. Loss of DBL-1 does not affect intestinal integrity upon exposure to specific bacteria. Wild-type and *dbl-1(-)* animals at the L4 stage were exposed to the following bacteria a) *E. coli* OP50 (control), b) *E. cloacae*, c) *K. oxytoca*, d) *S. marcescens*, e) *B. megaterium*, f) *E. faecalis*, or g) *S. epidermidis*. Intestinal barrier function was assessed using erioglaucine disodium salt a) over time or b–g) when *dbl-1(-)* populations neared their half lifespan. The leakiness of the intestine was assessed and scored as '1' for no leakage/no Smurf, '2' for mild leakage/mild Smurf, and '3' for severe leakage/severe Smurf phenotypes. The fraction of animals indicating these phenotypes was calculated. One representative trial of at least three is presented. n = at least 10 per condition.



Supplementary Figure 2. DBL-1 regulates differential gene expression in response to Gram-negative and Gram-positive bacteria. Wild-type and *dbl-1(-)* animals were exposed to *E. coli* OP50 (control), *S. marcescens*, or *E. faecalis* at the L4 stage for two days. RNA-seq analysis volcano plots show differential gene expression in animals lacking DBL-1 exposed to a) *E. coli* OP50, b) *S. marcescens*, and c) *E. faecalis* in comparison to wild-type animals exposed to the same bacteria (adjusted *p*-value < 0.01). Genes down-regulated in *dbl-1(-)* animals are represented in green, genes up-regulated in *dbl-1(-)* animals are represented in red, and genes with no change in expression are represented in blue.

CHAPTER III

CAENORHABDITIS ELEGANS BROOD SIZE AND EGG-LAYING RESPONSES TO SERRATIA MARCESCENS AND STAPHYLOCOCCUS EPIDERMIDIS ARE INDEPENDENT OF DBL-1 SIGNALING

This chapter contains a publication: Madhu, BJ; Salazar, AE; Gumienny, TL (2019). *Caenorhabditis elegans* egg-laying and brood-size changes upon exposure to *Serratia marcescens* and *Staphylococcus epidermidis* are independent of DBL-1 signaling. microPublication Biology. https://doi.org/10.17912/2r51-b476.

Description

Caenorhabditis elegans naturally thrives in a soil environment where they feed on bacteria and are in constant association with a diverse range of microbes (Barker et al. 1994). *C. elegans* egg laying is delayed or reduced when animals are infected with *Burkholderia pseudomallei*, *Burkholderia thailandensis*, *Staphylococcus aureus*, and *Serratia marcescens* (Irazoqui et al. 2010; Mallo et al. 2002; O'Quinn et al. 2001). These changes in egg laying may be a protective response to pathogenic bacteria. Mutants of TGF-β-like DBL-1 signaling pathway also display reduced brood size (Luo et al. 2009; Roberts et al. 2010). While the peak of egg-laying activity seen in normal animals between days 2 and 4 is depressed in *dbl-1* pathway mutants, the reproductive span of these *dbl-1* pathway mutants is increased to up to 13 days (Luo et al. 2009). To determine if the egg-laying observed during infection is DBL-1 pathway-dependent, we tested the effect of the DBL-1 signaling pathway on egg

laying when *C. elegans* were fed on representative Gram-negative (*S. marcescens*) and Gram-positive bacteria (*Staphylococcus epidermidis*).

Similar to previously published reports, we found that loss of DBL-1 pathway signaling decreases brood size and increases reproductive span in normal laboratory conditions (*E. coli* strain OP50 and 20°C incubation; see Figure 3.1A and B; Luo et al. 2009; Roberts et al. 2010).



Figure 3.1. Effects of *S. marcescens* and *S. epidermidis* on egg laying and brood size in wild-type and *dbl-1(nk3)* populations. (A) Comparison of mean eggs laid between wild-type (WT) and *dbl-1(nk3)* populations fed on *E. coli* OP50, *S. marcescens*, or *S. epidermidis*. The counts of number of eggs laid were reported only for plates with live animals. Error bars represent SEM. n = 7–10 surviving animals each. (B) Comparison of total brood size between wild-type and *dbl-1(nk3)* populations fed on *E. coli* OP50, *S. marcescens*, or *S. epidermidis*. Total brood sizes of animals that desiccated were censored. Error bars represent SEM. n = 7–10 surviving animals each. Within each genetic background, significant differences between the *E. coli* OP50-fed control and the pathogen-fed populations are marked with asterisks (*, p < 0.05, **, p < 0.001).

Here, we report three new results. First, brood size reductions caused by infection and by loss of DBL-1 signaling are independent (see Figure 3.1A). Wild-type and dbl-1(nk3) animals both significantly decrease their brood size when

grown on *S. marcescens* (p = 0.005 and p < 0.001, respectively). *dbl-1* mutant animals laid even fewer eggs than the wild-type animals on *S. marcescens*, suggesting that the reduced brood size phenotype is independently affected by both *S. marcescens* exposure and by loss of DBL-1 (p < 0.001). While the decrease in brood size of wild-type animals on *S. epidermidis* was not significant (p = 0.115), the decreased brood size of *dbl-1(nk3)* animals was significant on this pathogenic bacterial strain (p = 0.045). Indeed, the decreases in brood size upon infection with either *S. marcescens* or *S. epidermidis* in both wild-type and *dbl-1(nk3)* populations are similar (p = 0.57), suggesting the pathogenic bacteria affect brood size independent of DBL-1. Because *dbl-1(nk3)* populations display a further reduced brood size upon infection by either pathogen compared to the wild type, the negative effects of pathogen exposure and loss of DBL-1 signaling on brood size appear to be additive (p < 0.05).

Second, while the wild-type population on *S. marcescens* survived until all animals ceased laying eggs, all *dbl-1(nk3)* animals died on *S. marcescens* by Day 5. These results explain why the extended reproductive span normally seen in *dbl-1(nk3)* populations was not observed on *S. marcescens*. These results also support previous reports of decreased viability of *dbl-1* mutant animals on another variety of *S. marcescens*, Db11 (Mallo et al. 2002).

Third, *S. epidermidis* affects egg-laying patterns similar to loss of *dbl-1* function. Initially, both wild-type and *dbl-1(nk3)* strains on *S. epidermidis* have reduced eggs laid in the first four days compared to strains grown on the *E.*

coli control. Loss of DBL-1 further reduced the number of eggs laid during each of these days, suggesting that this phenotype is independently affected by both *S. epidermidis* exposure and by loss of DBL-1 (p = 0.004). Then, both wild-type and *dbl-1(nk3)* strains on *S. epidermidis* have similar extended reproductive spans, extending to at least Day 13 (one tenacious wild-type hermaphrodite laid embryos until Day 15). This *S. epidermidis*-induced reproductive span extension appears to be independent of DBL-1 signaling, because the numbers of eggs laid by both wild-type and *dbl-1(nk3)* populations between Days 5 and 16 were similar at these time points (p = 0.509).

Methods

Animals were age-synchronized by hypochlorite treatment (Stiernagle, 2006) and grown on plates seeded with *Escherichia coli* OP50. Ten L4 animals were manually transferred to individual plates seeded with *S. marcescens* or *S. epidermidis*. Plates were completely covered by bacteria to prevent animals from avoiding the bacteria. Adults were daily transferred to new plates and the number of eggs laid on each plate was counted every 24 hours until no more eggs were laid. Statistical analyses were performed using repeated measures ANOVA and Tukey's post-hoc test.

Reagents

Strains were maintained on EZ media plates at 20°C (0.55 g Tris-Cl, 0.24 g Tris base, 3.1 g BD BactoTM Peptone, 8 mg cholesterol, 2 g sodium chloride, 20 g agar, in water to 1 L (E. Lambie, personal communication). The *C.*

elegans strains used were N2 and NU3 *dbl-1(nk3)*. The Gram-negative bacterial strains used were *Escherichia coli* OP50 (CGC) and *Serratia marcescens* (Carolina Biological Supply Company). The Gram-positive bacterial strain used was *Staphylococcus epidermidis* (ATCC 49134). *S. marcescens* and *S. epidermidis* were provided by A. J. Hammett, TWU. All bacterial strains were grown for 9 hours in tryptic soy broth at 37°C before plating on EZ media plates.

References

Barker KR, Hussey RS, Krusberg LR, Bird GW, Dunn RA, Ferris H, et al., MacGuidwin AE. 1994. Plant and soil nematodes: Societal impact and focus for the future. J Nematol. 26(2):127. <u>PubMed</u>

Irazoqui JE, Troemel ER, Feinbaum RL, Luhachack LG, Cezairliyan BO, Ausubel

FM. 2010. Distinct pathogenesis and host responses during infection of C.

elegans by P. aeruginosa and S. aureus. PLoS Pathog. 6(7):e1000982. PubMed

Luo S, Shaw WM, Ashraf J, Murphy CT. 2009. TGF-β Sma/Mab signaling

mutations uncouple reproductive aging from somatic aging. PLoS Genet.

5(12):e1000789. PubMed

Mallo GV, Kurz CL, Couillault C, Pujol N, Granjeaud S, Kohara Y, Ewbank JJ.

2002. Inducible antibacterial defense system in *C. elegans*. Curr Biol.

12(14):1209–1214. PubMed

O'Quinn AL, Wiegand EM, Jeddeloh JA. 2001. *Burkholderia pseudomallei* kills the nematode *Caenorhabditis elegans* using an endotoxin-mediated paralysis. Cellular Microbiol. 3(6):381–393. <u>PubMed</u>

Roberts AF, Gumienny TL, Gleason RJ, Wang H, Padgett RW. 2014. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. BMC Dev Biol. 10(1):61. <u>PubMed</u> Stiernagle T. 2006. Maintenance of *C. elegans*. Pasadena (CA): WormBook; [accessed March 28, 2019]. http://doi.org/10.1895/wormbook.1.101.1.

CHAPTER IV

CAENORHABDITIS ELEGANS SAPOSIN-LIKE SPP-9 IS INVOLVED IN

SPECIFIC INNATE IMMUNE RESPONSES

This chapter contains a publication: Madhu, B., Lakdawala, M.F., Issac, N. G., Gumienny, T.L. (2020) *Caenorhabditis elegans* saposin-like *spp-9* is involved in specific innate immune responses. Genes & Immunity. https://doi.org/10.1038/s41435-020-0108-6

Bhoomi Madhu, Mohammed Farhan Lakdawala, Neethu Issac, and Tina L.

Gumienny*

Department of Biology, Texas Woman's University, Denton, TX 76204-5799

*Corresponding author

Running title: specificity of *spp-9* expression

Key words: saposin-like proteins, *spp-9*, *dbl-1*, innate immune responses, TGF-β

superfamily signaling

Correspondence: Tina L. Gumienny

PO Box 425799

Denton, TX 76204-5799

Phone: 940-898-2295

Email: tgumienny@twu.edu

Abstract

Animals counter specific environmental challenges with a combination of broad and tailored host responses. One protein family enlisted in the innate immune response includes the saposin-like anti-microbial proteins. We investigated the expression of a *Caenorhabditis elegans* saposin-like gene, *spp-9*, in response to different stresses. *spp-9* expression was detected in the intestine and six amphid neurons, including AWB and AWC. *spp-9* expression is increased in response to starvation stress. In addition, we discovered pathogen-specific regulation of *spp-9* that was not clearly demarcated by Gram nature of the bacterial challenge. Multiple molecular innate immune response pathways, including DBL-1/TGF-β-like, insulin-like, and p38/MAPK, regulate expression of *spp-9*. Our results suggest *spp-9* is involved in targeted responses to a variety of abiotic and bacterial challenges that are coordinated by multiple signaling pathways.

Introduction

All living organisms possess mechanisms to protect themselves from potentially harmful environments (1, 2). Roundworms are a diverse group of organisms (comprising both free-living and parasitic members) that have adapted to various habitats ranging from terrestrial to marine environments (3). In these environments, roundworms encounter a variety of challenges, including low food availability and pathogenic bacteria (4). *Caenorhabditis elegans*, a free-living roundworm, is an established model organism to understand defense mechanisms nematodes use, including innate immune responses to bacteria (5). It is important that these responses are well coordinated and appropriately regulated by the host (1, 2). Aberrant activation of host immune responses could be a potential cellular stress for the host, making tight regulation of host immune responses important for host health in low-stress environments, as well (6, 7). How animals respond to the various challenges in their natural environment by mounting specific defenses is not well understood.

Several pathogens infect *C. elegans*, including bacterial strains *Pseudomonas aeruginosa*, *Staphylococcus aureus*, *Salmonella enterica* serovar Typhimurium, *Serratia marcescens*, *Enterococcus faecalis*, *Mycobacterium fortuitum*, *Bacillus megaterium*, *Microbacterium nematophilum*, and *Staphylococcus epidermidis*. *C. elegans* is also infected by fungi, including *Candida albicans* and *Drechmeria coniospora* (8–15). MAPK signaling, insulinlike signaling, and DBL-1/TGF-β-like signaling pathways are induced upon

infection by these pathogens (8–13, 16–20). These pathways regulate an overlapping set of antimicrobial genes, which suggests crosstalk between the pathways upon infection (18, 21, 22). These antimicrobial genes encode lipases, lysozymes, defensin-like proteins, and saposin-like proteins (SAPLIPs or caenopores).

C. elegans caenopores have structural similarities with the protozoan amoebapores and the mammalian peptides NK-lysin and granulysin (23, 24). 28 genes are predicted to encode 33 SAPLIP-domain containing proteins, but antimicrobial activity of only a few saposins (SPPs) has been characterized. Functional analyses of SPP-3, SPP-5, and SPP-12 show that they display poreforming activity, permeabilize the cytoplasmic membrane of bacteria, and kill bacteria (24–26). SPP-1 is required for protection of *C. elegans* against *S. enterica* serovar Typhimurium and *P. aeruginosa* through the DAF-2/DAF-16 signaling pathway (18, 27). Some *spp* genes are highly up- or down-regulated in response to bacterial challenges (18, 22, 24–27). In addition, one caenopore gene, *spp-3*, has been shown to be induced in starved animals (24, 25). Although some caenopores are strongly expressed in the intestine, some are also expressed in specific nerves (18, 24–28). One caenopore, *spp-12*, is expressed only in NSML/R and I6 pharyngeal neurons (26).

To expand our understanding of the function of caenopores within the context of innate immunity and stress, we chose *spp*-9, one member of the caenopore family of antimicrobial proteins. Phylogenetic analyses show that *spp*-

9 is closely related to *spp-3*, whose role as an antimicrobial protein has been functionally characterized (24, 25). *spp-9* is expressed in intestine, an organ on the front line of innate immunity (21). *spp-9* expression is upregulated in response to loss of DKF-2/protein kinase D (29). *spp-9* is also negatively regulated by the DBL-1 signaling pathway and is used as DBL-1 pathway reporter (21, 30). While *spp-9* has been shown to be a DBL-1-responsive gene, here we characterized the role of *spp-9* in the larger context of starvation stress and innate immunity. We also determined additional molecular innate immune/stress pathways that regulate *spp-9* expression in *C. elegans*.

Materials and Methods

Strains and Maintenance

All *C. elegans* strains were maintained on EZ media plates at 20°C except *daf-2* strains, which were maintained at 17°C, unless specified otherwise (31). Strains used were: wild-type N2, NU3 *dbl-1(nk3)* V, TLG182 *texls100[dbl-1::dbl-1:gfp; ttx-3p::rfp]* IV, NL2099 *rrf-3(pk1426)* II, TLG697 *texls127[spp-9p::gfp]* X, LT998 *wkls40[spp-9p::gfp]*, TLG707 *dbl-1(nk3)* V; *texls127* X, TLG755 *oyls44* V; *texls127* X, TLG756 *daf-16(mu86)* I ; *texls127* X, TLG757 *pmk-1(km25)* IV; *texls127* X, TLG758 *wkls40*; *mut-2(r459)* I; *mek-1(pk97)* X, TLG759 *daf-2(e1370)* III; *texls127* X, TLG760 *wkls40*; *sek-1(km4)* X, TLG761 *wkls40*; *tir-1(tm3036)* III. These strains were generated by standard genetic crosses and confirmed by PCR.

The bacteria used in this study include *Bacillus megaterium* (Carolina Biological Supply Company), *Escherichia coli* (OP50), *Enterobacter cloacae* (49141TM), *Klebsiella oxytoca* (49131TM), *Serratia marcescens* (Carolina Biological Supply Company), and *Staphylococcus epidermidis* (49134TM). All bacteria were grown in tryptic soy broth for nine hours at 37°C. Bacterial cells were pelleted at 5000 rpm for 15 minutes and concentrated twenty-fold. EZ media plates were freshly seeded with concentrated bacteria in full lawns. The plates were incubated at 37°C overnight before they were used in experiments.

Dauer Assay

daf-2(e1370) III; texls127 X and *texls127 X* populations were synchronized as embryos by bleaching (32). Animals were transferred to 25°C at the L2 stage. Dauered *daf-2(e1370) III; texls127 X* animals and L4 *texls127 X* controls were picked. Two days later, animals were imaged as dauer (*daf-2(e1370) III; texls127* X) or adults (*texls127* X). The experiment was performed in three independent trials.

Imaging

Fluorescence of the *spp-9* reporter strain at different developmental stages was captured by a Nikon A1 confocal system (Nikon Instruments, Melville, NY). Colocalization studies of *oyIs44 V*; *texIs127 X* animals were performed on a Nikon swept-field confocal system (Nikon Instruments, Melville, NY). Hermaphrodites were synchronized as embryos by bleaching and imaged 48 hours after the L4 stage at 20°C, unless otherwise noted (32). *daf-16(mu86) I*;

texls127 X animals and respective control animals (*texls127 X*) were grown at 25°C and imaged 48 hours after the L4 stage. Fluorescence of the *spp-9* reporter strains was captured by a Nikon DS-Ri2 camera mounted on a Nikon SMZ18 dissecting microscope (Nikon Instruments, Melville, NY). Animals were mounted on 2% agarose pads and anesthetized by using 1 mM levamisole. At least 15 animals were imaged per condition as determined by power analysis with a moderate effect size. The microscope conditions were optimized with respect to the control and test conditions and kept consistent within each trial. However, imaging exposure times were different between some trials to prevent saturation of signal in experimental conditions. All imaging experiments were performed in three independent trials. Mean fluorescence intensities were measured using the Nikon NIS Elements AR v5.02 software and were analyzed using the unpaired *t*-test.

RNA Isolation and qRT-PCR

Total RNA was extracted from animals at 48 hours after the L4 stage. Animals were synchronized by bleaching (32). Total RNA was extracted by the freeze cracking method as previously described (33). After RNA isolation, 2 µg of total RNA was primed with oligo(dT) and reverse transcribed to yield cDNA using the SuperScript III reverse transcriptase kit as per manufacturer's protocol (Invitrogen). Real-time PCR was performed on QuantStudio3 (Applied Biosystems by Thermo Fisher Scientific) instrument using the PowerUP SYBR Green master mix (Applied Biosystems) according to manufacturer's instructions.

The experiment was performed in three technical replicates for each condition. Primer sequences for *dbl-1* are forward primer GCCATTCTCCACCTCTTCCT and reverse primer GGAACATCAATGCTCGGACC (34). Primer sequences for *spp-9* are forward primer GTTCTCTTTCTGGTTGCGGT and reverse primer GCTCTACAAACATCTTCTGGTGCA. Primer sequences for *act-1* are forward primer CCATCATGAAGTGCGACATTG and reverse primer CATGGTTGATGGGGCAAGAG (13). QuantStudio Design and Analysis Software v1.5.1 was used to calculate raw Ct values and to normalize the values for *dbl-1* and *spp-9* to the housekeeping actin gene *act-1* (Applied Biosystems by Thermo Fisher Scientific). Fold change in gene expression between experimental sample and the wild-type control was determined by this software using the formula: $2^{(c-\Delta\Delta C_t)}$. Experimental ΔC_t values were compared to wild-type ΔC_t values using the

unpaired *t*-test.

Data and Reagent Availability

Strains are available upon request. Supplementary figures are available in Figshare.

Results

spp-9 Localizes in the Intestine and Head Neurons at All Developmental Stages

To determine where and when *spp-9* is expressed, we used a strain expressing an integrated transgene with the *spp-9* promoter driving expression of green fluorescent protein (*spp-9p::GFP*). Promoter activity of *spp-9* was

observed in the entire intestine of animals, strongly in the anterior and posterior ends (see Figure 4.1A–E). spp-9 transgene expression in the intestine was visible at all developmental stages starting from the first larval stage up to adulthood. We also observed *spp-9* expression in six head neurons (see Figure 4.1F). We confirmed the expression of *spp-9* in the AWB and AWC neurons by showing co-localization with a fluorescent marker, odr-1::RFP (oyls44), that is specifically expressed in the two AWB and two AWC sensory neurons (35, 36). The *spp-9* reporter was also visible in a third amphid pair. VisCello for Visualization of Single Cell Data expression data suggests *spp-9* is expressed in the ASH neurons, which is consistent with the observed position of this third neuron pair (37). Because robust intestinal expression of *spp-9* was observed only after L3, we quantitated spp-9 reporter activity at 24 hours, 36 hours, and 48 hours after staging animals at L4. We observed an increase in reporter activity over this time frame, with the highest level of expression detected in this time frame in two-day old adults (see Figure S1). These results indicate that spp-9 is expressed majorly in the intestinal tissue and also in six head neurons throughout all the developmental stages of animals, with *spp-9* intestinal expression increasing from late larval to adult stages.



Figure 4.1. The *spp-9* reporter is expressed throughout larval and adult stages in intestine and head neurons. Confocal images show expression of *spp-9p::*GFP in A) L1, B) L2, C) L3, D) L4, and E) 1-day adult in the intestine, with strong expression in the anterior and posterior ends of intestine. F) Rotated swept-field confocal projection reveals *spp-9p*::GFP is expressed in six head neurons, colocalizing with ODR-1::RFP in AWB and AWC head neurons of an adult animal (marked by filled and unfilled arrowheads, respectively). Asterisks mark a third neuron that expresses the *spp-9* reporter. Scale bar indicates A, B) 50 µm, C, D, E) 100 µm, and F) 10 µm.

spp-9 Expression is Increased in Starved Animals

Animals can mount responses to a variety of environmental stresses. Loss of some *dpy* genes that encode cuticle collagens induce glycerol, osmotic, and

detoxification responses, but do not change spp-9 reporter expression (30, 38-

41). Starvation is another stress, and *spp-3* expression is induced in starved

animals (24, 25). However, *spp-5* expression, which is constitutively expressed

and remains unchanged in pathogenic conditions, is also high in starved animals

(24). We asked if starvation affects expression of *spp-9*. By 8 hours without food

starting in the L4 stage, C. elegans show dramatic organismal responses,

including a global reduction in protein synthesis and changes in stress response pathway gene expression (42). L4 animals that were starved were imaged 8 hours after starvation and compared to identically staged animals that remained fed during that time. *spp-9* expression was significantly increased when animals were deprived of food (see Figure 4.2). This suggests that expression of *spp-9* is upregulated in starved animals.



Figure 4.2. Starvation increases *spp-9* reporter activity. Fluorescence images show *spp-9p::*GFP intensities of A) fed L4 animals and B) age-matched animals starved for 8 hours post L4 stage. Imaging conditions including exposure times were consistent. Scale bar indicates 100 µm. C) Comparison of *spp-9p::*GFP intensities of fed animals (control) with starved animals indicates that *spp-9* reporter activity was significantly increased in starved animals in comparison to fed animals (p = 0.0003). One representative trial of three is presented. Error bars represent 95% confidence intervals. n = 30 per condition. *, *p* < 0.05 compared to fed control by unpaired *t*-test.

spp-9 Expression Is Altered in Response to Select Gram-Positive and

Gram-Negative Pathogens

Because caenopores are predicted antimicrobial effector genes, a subset

of this 23-member family has been tested for induction upon pathogen challenge.

Expression of *spp-1* is induced upon infection with *S. enterica* serovar

Typhimurium (43). *spp*-3 expression is induced upon confrontation with two Gram-positive bacterial species, B. megaterium and M. luteus, compared to normal lab food E. coli OP50, but was not induced on a different Gram-positive bacterium (Lactobacillus lactis) or any of the four Gram-negative species tested. Expression of *spp-5*, though, was consistent in animals exposed to different Gram-positive and Gram-negative bacteria, including E. coli (24). To determine if expression of *spp-9* is responsive to exposure to pathogenic bacteria, we detected activity of the spp-9p::gfp reporter in response to a panel of Grampositive and Gram-negative bacterial exposures. This panel included two Grampositive (B. megaterium and S. epidermidis) and three Gram-negative strains (E. cloacae, K. oxytoca, and S. marcescens), plus the standard lab food, Gramnegative *E. coli* OP50. Animals expressing *spp-9p::gfp* were synchronized as L4s and fed on Gram-positive or Gram-negative bacteria for 48 hours. Animals were tested at 48 hours because by that time, there was a robust response of the spp-9 reporter on E. coli (see Figure S1). Basal expression was measured for spp-*9p*::GFP fluorescence in animals fed *E. coli* OP50 (see Figure 4.3A, E, H). We observed a significant increase in the spp-9 reporter activity when animals were fed on select Gram-positive (B. megaterium and S. epidermidis), and Gramnegative (K. oxytoca) bacteria (see Figure 4.3B, C, D, G, J). On the other hand, exposure to S. marcescens caused a significant decrease in spp-9 expression in two of the three independent trials (see Figure 4.3I, J). This reduced spp-9 expression is consistent with previous studies showing exposure to S.

marcescens induces the DBL-1 signaling pathway (which negatively regulates *spp-9*) (44). Our results suggest that activity of *spp-9* is differentially regulated in response to specific Gram-positive and Gram-negative bacterial challenges.



Figure 4.3. *spp-9* reporter activity is altered upon exposure to specific Grampositive and Gram-negative bacteria. Fluorescence images show *spp-9p::*GFP intensities in adult animals after a two-day exposure to the following bacteria; A, E, H) control *E. coli* OP50 (n = 17, 20, 20), B) *B. megaterium* (n = 17), C) *S. epidermidis* (n = 20), F) *E. cloacae* (n = 21), G) *K. oxytoca* (n = 21), or I) *S. marcescens* (n = 21). Imaging conditions including exposure times were consistent with respective controls. Scale bar indicates 100 µm. D) Comparison

of *spp-9p::*GFP intensities of wild-type animals fed on *E. coli* OP50 (control) with animals fed on *B. megaterium* or *S. epidermidis. spp-9p::*GFP intensity was significantly increased when animals were exposed to *B. megaterium* and *S. epidermidis* in comparison to the control (p < 0.0001). J) Comparison of *spp-9p::*GFP intensities of wild-type animals fed on *E. coli* OP50 (control) with animals fed on *E. cloacae*, *K. oxytoca*, or *S. marcescens*. No significant change in *spp-9p::*GFP intensity was seen in animals fed on *E. cloacae* (p = 0.1353). Exposure to *K. oxytoca* caused a significant increase in the GFP intensity as compared to the control (p < 0.0001). On the other hand, exposure to *S. marcescens* caused a significant decrease in the GFP intensity in comparison to the control (p = 0.0153). One representative trial of three is presented. Error bars represent 95% confidence intervals. *, p < 0.05 compared to *E. coli* OP50 by unpaired *t*-test.

dbl-1 Regulates Endogenous spp-9 Expression Levels

Studies by our lab and others have shown that the activity of *spp-9* is highly regulated by DBL-1 in an inverse fashion. Animals overexpressing DBL-1 show reduced *spp-9p*::GFP activity and animals lacking DBL-1 display high reporter activity (21, 30). We tested if DBL-1 regulates endogenous levels of *spp-9* expression by quantitative real-time PCR, comparing *spp-9* expression levels from *dbl-1* mutants to the wild-type control. Animals with the *dbl-1(nk3)* null allele, a deletion that deletes about 5 kb of the 3' end of the 7 kb open reading frame, have almost no *dbl-1* mRNA detected by qRT-PCR. Animals overexpressing *dbl-1* have about a 10-fold increase in *dbl-1* mRNA levels (see Figure S2A). We see an increase (about 2.6-fold) in the expression of *spp-9* in animals lacking *dbl-1*, whereas we see a decrease (about 0.5-fold), however not significant, in *spp-9* mRNA levels of animals overexpressing *dbl-1* (see Figure S2B). These qRT-PCR results support previously reported microarray results and the use of *spp-9p*::GFP as a reporter for DBL-1 pathway signaling (21, 30).

spp-9 Expression Depends on DBL-1 and Other Innate Immune Response Signaling Pathways, Depending on Pathogen

DBL-1 signaling pathway regulates expression of many innate immunity genes and is also involved in mounting protective immune responses against pathogens (13, 18, 44–48). We showed that the DBL-1 pathway target gene spp-9 is differentially regulated upon exposure to different types of Gram-positive and Gram-negative bacteria. Therefore, we asked if the differential regulation of the spp-9 reporter activity in response to the panel of Gram-positive and Gramnegative bacteria is DBL-1 mediated. To test this, we exposed *spp-9p::gfp* and *dbl-1(-);* spp-9p::gfp animals to our panel of different bacteria at the L4 stage. These animals were imaged 48 hours after L4 stage and we measured and compared GFP intensities of the two genotypes fed on different bacteria. In three independent trials, the *dbl-1(-);* spp-9p::gfp animals when fed on standard lab food, *E. coli* strain OP50, showed a significantly increased fluorescence, consistent with previous reports (see Figure 4.4A, E; 21, 30). We observed that animals lacking *dbl-1* when fed on *S. epidermidis*, *E. cloacae*, and *S.* marcescens showed a further increase in fluorescence in comparison to the control (dbl-1(-); spp-9p::gfp animals fed on E. coli OP50) (see Figure 4.4C, D, F, H, I). These findings indicate that loss of *dbl-1* increases the *spp-9* reporter activity and exposure to select Gram-positive (S. epidermidis) and Gramnegative (E. cloacae and S. marcescens) bacteria further increases the reporter

activity when *dbl-1* is absent, suggestive of an additive effect on *spp-9* reporter activity, perhaps by other innate immune response pathways.

On the other hand, *dbl-1(-); spp-9p::gfp* animals showed similar mean fluorescence intensities when fed on *B. megaterium* in comparison to *dbl-1(-); spp-9p::gfp* animals exposed to the *E. coli* OP50 control (see Figure 4.4B, D). This response is different from *spp-9* reporter activity in the wild type, which showed significantly increased fluorescence with *B. megaterium* exposure as compared to the control (see Figure 4.3B, D). This lack of further *spp-9* reporter induction in *dbl-1* mutants on *B. megaterium*, like was observed on *E. cloacae*, *S. marcescens*, and *S. epidermidis*, indicates that *spp-9* reporter activity induced on *B. megaterium* is DBL-1 mediated (see Figures 4.3 and 4.4).

Interestingly, while animals lacking *dbl-1* fed on *K. oxytoca* displayed increased fluorescence than *spp-9p::gfp* animals, that increase was significantly less in comparison to the same genotype fed on *E. coli* in two out of three trials (see Figures 4.3 and 4.4, data from both genotypes was collected in each trial). This suggests that some other signaling pathway that negatively regulates *spp-9* activity is induced in *dbl-1(-)* animals upon exposure to *K. oxytoca*, resulting in decrease of *spp-9* activity. This suggests that the reporter activity response to *K. oxytoca* exposure depends on an innate immune response that is at least partially independent of *dbl-1*.



Figure 4.4. spp-9 reporter activity is altered in response to specific pathogens by DBL-1 and other innate immunity signaling pathways. Fluorescence images show spp-9p::GFP intensities in adult dbl-1(-) animals after a two-day exposure to the following bacteria; A, E) control E. coli OP50 (n = 20), B) B. megaterium (n= 18), C) S. epidermidis (n = 26), F) E. cloacae (n = 19), G) K. oxytoca (n = 22), or H) S. marcescens (n = 21). Imaging conditions including exposure times were consistent with respective controls. Scale bar indicates 100 µm. D) Comparison of spp-9p::GFP intensities in dbl-1(-) animals fed on E. coli OP50 (control) with animals fed on *B. megaterium* or *S. epidermidis*. No significant change in spp-*9p*::GFP intensity was seen in animals fed on *B. megaterium* (p = 0.3534). The GFP intensity was significantly increased when animals were exposed to S. epidermidis in comparison to the control (p = 0.0117). I) Comparison of spp-9p::GFP intensities in dbl-1(-) animals fed on E. coli OP50 (control) with animals fed on either E. cloacae, K. oxytoca, or S. marcescens. Exposure to E. cloacae (p = 0.0001) and S. marcescens (p = 0.0385) caused an increase in the GFP intensity as compared to the control. Exposure to K. oxytoca caused a decrease
in the GFP intensity in comparison to the control (p = 0.0001). One representative trial of three is presented. Error bars represent 95% confidence intervals. *, p < 0.05 compared to *E. coli* OP50 by unpaired *t*-test.

spp-9 Reporter Activity Is Regulated by Innate Immunity/Stress Signaling

Pathways

Because we see an additive effect of DBL-1 and infection on the activity of spp-9, and also DBL-1-independent but pathogen-dependent regulation of spp-9, we asked whether *spp-9* is also regulated by other signaling pathways in the context of immunity. Besides the DBL-1 signaling pathway, two major pathways that are required for animals to respond to pathogens are the insulin-like and p38/MAPK signaling pathways (19, 49, 50). The insulin-like pathway is defined by the insulin receptor DAF-2 and the downstream master transcriptional regulator DAF-16. Loss of *daf-2* function not only causes animals to constitutively enter dauer but also increases resistance to pathogens. Loss of daf-16 prevents animals from entering dauer and increases sensitivity to infection (12, 49, 51-54). We measured reporter fluorescence intensities in *daf-2(-)* and *daf-16(-)* backgrounds and compared it to the reporter in the wild-type background. Interestingly, we observed a significant reduction in *daf-2(-)* mutants (see Figure 4.5B, D). As expected, animals lacking *daf-16* showed increased *spp-9* reporter fluorescence (see Figure 4.5C, D). These findings suggest that spp-9 is responsive to changes in insulin-like signaling: DAF-2 promotes, while DAF-16 normally represses, *spp-9* expression.

The p38/MAPK signaling pathway includes TIR-1, TIR (Toll and Interleukin 1 Receptor) domain protein, which activates MAP3K, which activates MAP2K SEK-1, which activates MAPK PMK-1 (55–57). We captured and measured GFP intensities of the reporter in *pmk-1(-)*, *sek-1(-)*, or *tir-1(-)* mutant backgrounds and compared them to the wild-type reporter control. We observed a significant increase in the mean GFP intensities in tir-1(-) and pmk-1(-) mutants (see Figure 4.5G, H, J, K). However, loss of MAP2K gene sek-1(-) had no effect on the fluorescent intensities in comparison to the control (see Figure 4.5F, H). We then tested if MEK-1, a stress-responsive MAP2K that can also activate PMK-1 independent of TIR-1, affected spp-9 reporter activity (56). Loss of mek-1, like loss of sek-1, also had no effect on spp-9 reporter fluorescence (see Figure 4.5M, N). These findings suggest that the MAPK pathway defined by *tir-1* and *pmk-1* normally suppresses *spp-9* activity, but does so using a MAP2K other than SEK-1 or MEK-1. Alternatively, these MAP2Ks act redundantly in the pathway that leads to expression of *spp-9*. Together, these results indicate *spp-9*. is regulated by multiple signaling pathways.



Figure 4.5. *spp-9* reporter activity is regulated by innate immunity signaling pathways. Fluorescence images show *spp-9p::*GFP intensities in two-day adults with A, E, I, L) wild-type (n = 24, 23, 19, 19), B) *daf-2(-)* (n = 24), C) *daf-16(-)* (n = 20), F) *sek-1(-)* (n = 23), G) *tir-1(-)* (n = 27), J) *pmk-1(-)* (n = 20), and M) *mek-1(-)* (n = 22) backgrounds. Imaging conditions including exposure times were

consistent with respective controls. Scale bar indicates 100 µm. D) Comparison of *spp-9p::*GFP intensities in wild-type (control), *daf-2(-)*, and *daf-16(-)* backgrounds. Loss of *daf-2* caused a decrease in the GFP intensity (p < 0.0001), whereas loss of *daf-16* caused an increase in the GFP intensity (p < 0.0001) as compared to the control. H) Comparison of *spp-9p::*GFP intensities in wild-type (control), *sek-1(-)*, and *tir-1(-)* backgrounds. Loss of *tir-1* caused an increase in the reporter activity in comparison to the control (p < 0.0001). K) Comparison of *spp-9p::*GFP intensities in wild-type (control) and *pmk-1(-)* backgrounds. The GFP intensity increased in animals lacking *pmk-1* (p < 0.0001). N) Comparison of *spp-9p::*GFP intensities in wild-type (control) and *mek-1(-)* backgrounds. No significant change in *spp-9p::*GFP intensity was seen in *mek-1(-)* animals (p = 0.5674). One representative trial of three is presented. Error bars represent 95% confidence intervals. *, p < 0.05 compared to *spp-9p::*GFP in wild-type background by unpaired *t*-test.

Discussion

Here, we show that SPP-9 has a unique role within its large family of saposin-like, pore-forming proteins in *C. elegans*, and its expression is modulated to respond to specific pathogens and other stressors.

spp-9 is visibly expressed from the first larval stage, when animals are initially exposed to environmental conditions, including pathogens, and this expression is maintained throughout the life of the animal. This observation is consistent with previous RNAseq and microarray analyses that do not detect spp-9 expression in embryos (WormBase.org). Basal expression of spp-9 is observed in the intestine of animals fed normal lab food, *E. coli*. Other saposin family members play roles in the digestion of the roundworm's bacterial food. SPP-9 may also help digest bacteria and is upregulated in some environments with pathogenic bacteria. However, our work supports a role for *spp-9* in stress response, not just digestion. For one, *spp-9* expression is increased in starvation conditions (see Figure 4.2). *spp-3* and *spp-5* have also been shown to be expressed in starved adults (24). Second, *spp-9* is expressed in the AWB and AWC sensory neuron pairs, which play roles in sensing numerous stimuli, including bacterial food sources (58–61). C. elegans uses AWB to sense and avoid serrawettin W2, a chemical secreted by the pathogenic S. marcescens (62). *spp-9* expression in amphid neurons might contribute to sensing different pathogens or other stresses. Some other *C. elegans* saposins are expressed not only in the intestine, but also or instead in a specific neuron or a few neurons.

spp-3 is expressed in the SDQR interneuron, *spp-12* is expressed in NSM/L and I6, which can sense bacteria, including pathogens, and *spp-7* is expressed in head neurons (25, 26). Neurons in *C. elegans* and other species express saposins and other antimicrobial proteins, possibly to protect the environmentally exposed neurons or surrounding tissues. Alternatively, the expression of different saposins in different environmentally exposed neurons could help specify distinct antimicrobial responses depending on the stimulus.

Third, a strong induction of *spp-9* was observed in animals exposed to some bacterial challenges, but not all, suggesting specificity of recruiting SPP-9 in a response that is not demarcated by Gram nature (see Figures 4.3 and 4.4). Furthermore, we discovered that *spp-9* is specifically upregulated by loss of specific innate immune/stress response pathways: the DBL-1/TGF- β -like pathway, the DAF-2/DAF-16 insulin-like pathway, and the p38/MAPK defined by TIR-1 and PMK-1, but independent of MAPKKs SEK-1 and MEK-1 (see Figures 4.4 and 4.5). This may explain why we see an additive induction of *spp-9* in animals lacking DBL-1 signaling and challenged with *E. cloacae*, *S. marcescens*, or *S. epidermidis*: other innate immune response pathways may be further inducing *spp-9* expression in this context. However, the response to *B*. *megaterium*, in which the *spp-9* induction depends on functional DBL-1 signaling, suggests that the animals use DBL-1 but not other innate immune pathways to induce a *spp-9* response to *B. megaterium*. However, the reduced *spp-9* induction upon exposure to K. oxytoca in dbl-1 mutant animals suggests that

other signaling pathways are dampening this response. These findings also support that DBL-1, insulin-like, and MAPK pathways act independently but converge to regulate *spp-9* expression. These results suggest that *spp-9* contributes to the "antimicrobial fingerprint" proposed by Alper et al. in which distinct molecular responses are generated against specific pathogen exposures (18).

spp-9 was identified as a highly regulated gene by the DBL-1 signaling pathway and a *spp-9* promoter-GFP transgene has been used as a reporter for this pathway (21, 30). The changes observed in the *spp-9* transcriptional reporter are consistent with the changes in endogenous *spp-9* expression levels in different *dbl-1* backgrounds, indicating that the reporter activity is representative of endogenous *spp-9* expression levels. Our findings support that *spp-9* is a valid reporter of DBL-1 signaling, with the caveat that the experimental environment must be closely controlled.

In summary, we discovered that *spp-9* expression is induced by starvation, specific bacteria, and by not only DBL-1, but other innate immune/stress pathways. It will be of interest to determine the molecular role of SPP-9 in replete and stress conditions, and to identify how multiple signaling pathways coordinate *spp-9* expression. This work revealed further insights into the environmental responsiveness of the saposin family and how organisms generate complex, targeted molecular responses to physiological challenges.

Acknowledgements

We thank Paul Yeatts and TWU CRDA for assistance with statistical analyses. The bacterial strains were provided by Amy Jo Hammett. We thank Robert Waterston for ASH neuron information. We thank Mehul Vora and Richard Padgett for *wkls40* and Maxwell Heiman for *oyls44*. Some strains were obtained from the *Caenorhabditis* Genetics Center (CGC), which is funded by NIH Office of Research Infrastructure Programs (P40 OD010440). We thank WormBase. We thank Laura Hanson, Yasar Arfat Kasu, and members of the Gumienny lab for constructive feedback. This work was supported by NIH R01GM097591 to TLG, a TWU Research Enhancement Program grant to TLG, internal funding by Texas Woman's University, a TWU Experiential Learning Scholar Award to BM, and a TWU Teaching and Research Grant for Equipment and Technology. This work is dedicated to Chimanlal Madhu and Darshan Madhu.

Conflict of Interest

The authors declare no conflict of interest.

References

- Medzhitov R, Janeway CA, Jr. An ancient system of host defense. Curr Opin Immunol. 1998;10(1):12–5.
- Cui J, Chen Y, Wang HY, Wang RF. Mechanisms and pathways of innate immune activation and regulation in health and cancer. Hum Vaccin Immunother. 2014;10(11):3270–85.
- McSorley R. Adaptations of nematodes to environmental extremes. Florida Entomologist. 2003;86 (2):138–42.
- Murfin KE, Dillman AR, Foster JM, Bulgheresi S, Slatko BE, Sternberg PW, et al. Nematode-bacterium symbioses - cooperation and conflict revealed in the "omics" age. Biol Bull. 2012;223(1):85–102.
- Engelmann I, Pujol N. Innate immunity in *C. elegans*. Adv Exp Med Biol. 2010;708:105–21.
- Cheesman HK, Feinbaum RL, Thekkiniath J, Dowen RH, Conery AL, Pukkila-Worley R. Aberrant activation of p38 MAP kinase-dependent innate immune responses is toxic to *Caenorhabditis elegans*. G3. 2016;6(3):541–9.
- Pukkila-Worley R. Surveillance immunity: an emerging paradigm of innate defense activation in *Caenorhabditis elegans*. PLoS Pathog. 2016;12(9):e1005795.
- Couillault C, Ewbank JJ. Diverse bacteria are pathogens of *Caenorhabditis* elegans. Infect Immun. 2002;70(8):4705–7.

- Gravato-Nobre MJ, Nicholas HR, Nijland R, O'Rourke D, Whittington DE, Yook KJ, et al. Multiple genes affect sensitivity of *Caenorhabditis elegans* to the bacterial pathogen *Microbacterium nematophilum*. Genetics. 2005;171(3):1033–45.
- Begun J, Gaiani JM, Rohde H, Mack D, Calderwood SB, Ausubel FM, et al. Staphylococcal biofilm exopolysaccharide protects against *Caenorhabditis elegans* immune defenses. PLoS Pathog. 2007;3(4):e57.
- 11. Wong D, Bazopoulou D, Pujol N, Tavernarakis N, Ewbank JJ. Genome-wide investigation reveals pathogen-specific and shared signatures in the response of *Caenorhabditis elegans* to infection. Genome Biol. 2007;8(9):R194.
- 12. Singh V, Aballay A. Regulation of DAF-16-mediated innate immunity in *Caenorhabditis elegans*. J Biol Chem. 2009;284(51):35580–7.
- Zugasti O, Ewbank JJ. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. Nat Immunol. 2009;10(3):249–56.
- Pukkila-Worley R, Ausubel FM, Mylonakis E. *Candida albicans* infection of *Caenorhabditis elegans* induces antifungal immune defenses. PLoS Pathog. 2011;7(6):e1002074.
- 15. Dzakah EE, Waqas A, Wei S, Yu B, Wang X, Fu T, et al. Loss of miR-83 extends lifespan and affects target gene expression in an age-dependent manner in *Caenorhabditis elegans*. J Genet Genomics. 2018;45(12):651–62.

- Aballay A, Yorgey P, Ausubel FM. Salmonella typhimurium proliferates and establishes a persistent infection in the intestine of *Caenorhabditis elegans*. Curr Biol. 2000;10(23):1539–42.
- 17. Garsin DA, Sifri CD, Mylonakis E, Qin X, Singh KV, Murray BE, et al. A simple model host for identifying Gram-positive virulence factors. Proc Natl Acad Sci U S A. 2001;98(19):10892–7.
- Alper S, McBride SJ, Lackford B, Freedman JH, Schwartz DA. Specificity and complexity of the *Caenorhabditis elegans* innate immune response. Mol Cell Biol. 2007;27(15):5544–53.
- Kim DH, Ewbank JJ. Signaling in the innate immune response. WormBook, The *C. elegans* Research Community. 2018;2018:1–35.
- Sun L, Zhi L, Shakoor S, Liao K, Wang D. microRNAs involved in the control of innate immunity in *Candida* infected *Caenorhabditis elegans*. Sci Rep. 2016;6:36036.
- 21. Roberts AF, Gumienny TL, Gleason RJ, Wang H, Padgett RW. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. BMC Dev Biol. 2010;10:61.
- Dierking K, Yang W, Schulenburg H. Antimicrobial effectors in the nematode *Caenorhabditis elegans*: an outgroup to the Arthropoda. Philos Trans R Soc Lond B Biol Sci. 2016;371(1695).
- Banyai L, Patthy L. Amoebapore homologs of *Caenorhabditis elegans*.
 Biochim Biophys Acta. 1998;1429(1):259–64.

- Roeder T, Stanisak M, Gelhaus C, Bruchhaus I, Grotzinger J, Leippe M. Caenopores are antimicrobial peptides in the nematode *Caenorhabditis elegans* instrumental in nutrition and immunity. Dev Comp Immunol. 2010;34(2):203–9.
- 25. Hoeckendorf A, Leippe M. SPP-3, a saposin-like protein of *Caenorhabditis elegans*, displays antimicrobial and pore-forming activity and is located in the intestine and in one head neuron. Dev Comp Immunol. 2012;38(1):181–6.
- 26. Hoeckendorf A, Stanisak M, Leippe M. The saposin-like protein SPP-12 is an antimicrobial polypeptide in the pharyngeal neurons of *Caenorhabditis elegans* and participates in defence against a natural bacterial pathogen. Biochem J. 2012;445(2):205–12.
- 27. Evans EA, Kawli T, Tan MW. *Pseudomonas aeruginosa* suppresses host immunity by activating the DAF-2 insulin-like signaling pathway in *Caenorhabditis elegans*. PLoS Pathog. 2008;4(10):e1000175.
- Murphy CT, McCarroll SA, Bargmann CI, Fraser A, Kamath RS, Ahringer J, et al. Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans*. Nature. 2003;424(6946):277–83.
- 29. Ren M, Feng H, Fu Y, Land M, Rubin CS. Protein kinase D is an essential regulator of *C. elegans* innate immunity. Immunity. 2009;30(4):521–32.
- 30. Lakdawala MF, Madhu B, Faure L, Vora M, Padgett RW, Gumienny TL. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body

size-associated genes in *Caenorhabditis elegans*. Mol Biol Cell. 2019;30(26):3151–60.

- Madhu BJ, Salazar AE, Gumienny TL. *Caenorhabditis elegans* egg-laying and brood-size changes upon exposure to *Serratia marcescens* and *Staphylococcus epidermidis* are independent of DBL-1 signaling. microPublication Biology. 2019, 10.17912/2r51-b476.
- Stiernagle T. Maintenance of *C. elegans*. WormBook, The *C. elegans* Research Community. 2006:1–11.
- Portman DS. Profiling *C. elegans* gene expression with DNA microarrays.
 WormBook. The *C. elegans* Research Community. 2006:1–11.
- 34. Kwon G, Lee J, Lim YH. Dairy *Propionibacterium* extends the mean lifespan of *Caenorhabditis elegans* via activation of the innate immune system. Sci Rep. 2016;6:31713.
- 35. Kim K, Kim R, Sengupta P. The HMX/NKX homeodomain protein MLS-2 specifies the identity of the AWC sensory neuron type via regulation of the *ceh-36* Otx gene in *C. elegans*. Development. 2010;137(6):963–74.
- Low IIC, Williams CR, Chong MK, McLachlan IG, Wierbowski BM, Kolotuev I, et al. Morphogenesis of neurons and glia within an epithelium. Development. 2019;146(4).
- Packer JS, Zhu Q, Huynh C, Sivaramakrishnan P, Preston E, Dueck H, et al. A lineage-resolved molecular atlas of *C. elegans* embryogenesis at single-cell resolution. Science. 2019;365(6459).

- Wheeler JM, Thomas JH. Identification of a novel gene family involved in osmotic stress response in *Caenorhabditis elegans*. Genetics. 2006;174(3):1327–36.
- Lamitina T, Huang CG, Strange K. Genome-wide RNAi screening identifies protein damage as a regulator of osmoprotective gene expression. Proc Natl Acad Sci U S A. 2006;103(32):12173–8.
- 40. Zugasti O, Thakur N, Belougne J, Squiban B, Kurz CL, Soule J, et al. A quantitative genome-wide RNAi screen in *C. elegans* for antifungal innate immunity genes. BMC Biol. 2016;14.
- Dodd W, Tang L, Lone JC, Wimberly K, Wu CW, Consalvo C, et al. A damage sensor associated with the cuticle coordinates three core environmental stress responses in *Caenorhabditis elegans*. Genetics. 2018;208(4):1467–82.
- Larance M, Pourkarimi E, Wang B, Brenes Murillo A, Kent R, Lamond AI, et al. Global proteomics analysis of the response to starvation in *C. elegans*. Mol Cell Proteomics. 2015;14(7):1989–2001.
- Alegado RA, Tan MW. Resistance to antimicrobial peptides contributes to persistence of *Salmonella typhimurium* in the *C. elegans* intestine. Cell Microbiol. 2008;10(6):1259–73.
- 44. Mallo GV, Kurz CL, Couillault C, Pujol N, Granjeaud S, Kohara Y, et al.
 Inducible antibacterial defense system in *C. elegans*. Curr Biol.
 2002;12(14):1209–14.

- 45. Mochii M, Yoshida S, Morita K, Kohara Y, Ueno N. Identification of transforming growth factor-β-regulated genes in *Caenorhabditis elegans* by differential hybridization of arrayed cDNAs. Proc Natl Acad Sci U S A. 1999;96(26):15020–5.
- Liang J, Yu L, Yin J, Savage-Dunn C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related signaling outputs. Dev Biol. 2007;305(2):714–25.
- 47. Zhang X, Zhang Y. DBL-1, a TGF-β, is essential for *Caenorhabditis elegans* aversive olfactory learning. Proc Natl Acad Sci U S A. 2012;109(42):17081–6.
- 48. Berg M, Monnin D, Cho J, Nelson L, Crits-Christoph A, Shapira M. TGFβ/BMP immune signaling affects abundance and function of *C. elegans* gut commensals. Nat Commun. 2019;10(1):604.
- 49. Garsin DA, Villanueva JM, Begun J, Kim DH, Sifri CD, Calderwood SB, et al.
 Long-lived *C. elegans daf-2* mutants are resistant to bacterial pathogens.
 Science. 2003;300(5627):1921.
- 50. Troemel ER, Chu SW, Reinke V, Lee SS, Ausubel FM, Kim DH. p38 MAPK regulates expression of immune response genes and contributes to longevity in *C. elegans*. PLoS Genet. 2006;2(11):e183.
- 51. Kenyon C, Chang J, Gensch E, Rudner A, Tabtiang R. A *C. elegans* mutant that lives twice as long as wild type. Nature. 1993;366(6454):461–4.

- Kimura KD, Tissenbaum HA, Liu Y, Ruvkun G. *daf-2*, an insulin receptor-like gene that regulates longevity and diapause in *Caenorhabditis elegans*.
 Science. 1997;277(5328):942–6.
- 53. Zhao L, Zhao Y, Liu R, Zheng X, Zhang M, Guo H, et al. The transcription factor DAF-16 is essential for increased longevity in *C. elegans* exposed to *Bifidobacterium longum* BB68. Sci Rep. 2017;7(1):7408.
- 54. Zhou M, Liu X, Yu H, Yin X, Nie SP, Xie MY, et al. Cell signaling of *Caenorhabditis elegans* in response to enterotoxigenic *Escherichia coli* Infection and *Lactobacillus zeae* protection. Front Immunol. 2018;9:1745.
- 55. Kim DH, Feinbaum R, Alloing G, Emerson FE, Garsin DA, Inoue H, et al. A conserved p38 MAP kinase pathway in *Caenorhabditis elegans* innate immunity. Science. 2002;297(5581):623–6.
- 56. Kim DH, Liberati NT, Mizuno T, Inoue H, Hisamoto N, Matsumoto K, et al. Integration of *Caenorhabditis elegans* MAPK pathways mediating immunity and stress resistance by MEK-1 MAPK kinase and VHP-1 MAPK phosphatase. Proc Natl Acad Sci U S A. 2004;101(30):10990–4.
- 57. Liberati NT, Fitzgerald KA, Kim DH, Feinbaum R, Golenbock DT, Ausubel FM. Requirement for a conserved Toll/interleukin-1 resistance domain protein in the *Caenorhabditis elegans* immune response. Proc Natl Acad Sci U S A. 2004;101(17):6593–8.
- Bargmann CI, Hartwieg E, Horvitz HR. Odorant-selective genes and neurons mediate olfaction in *C. elegans*. Cell. 1993;74(3):515–27.

- Troemel ER, Kimmel BE, Bargmann CI. Reprogramming chemotaxis responses: sensory neurons define olfactory preferences in *C. elegans*. Cell. 1997;91(2):161–9.
- Sagasti A, Hobert O, Troemel ER, Ruvkun G, Bargmann CI. Alternative olfactory neuron fates are specified by the LIM homeobox gene *lim-4*. Genes Dev. 1999;13(14):1794–806.
- Olofsson B. The olfactory neuron AWC promotes avoidance of normally palatable food following chronic dietary restriction. J Exp Biol. 2014;217(Pt 10):1790–8.
- Pradel E, Zhang Y, Pujol N, Matsuyama T, Bargmann CI, Ewbank JJ.
 Detection and avoidance of a natural product from the pathogenic bacterium Serratia marcescens by Caenorhabditis elegans. Proc Natl Acad Sci U S A.
 2007;104(7):2295–300.

Supplementary Information



Time after L4 stage (hours)

Figure S1. Expression of the *spp-9* reporter increases over time. Comparison of the *spp-9p::*GFP reporter fluorescence over time indicates that *spp-9* expression increased by 48 (n = 13) hours post L4 stage in comparison to 24 (n = 10) and 36 hours (n = 14) post L4 stage (p < 0.0001). No significant difference in *spp-9p::*GFP intensity was observed between 24 hours after L4 and 36 hours after L4 stage (p = 0.297). Imaging conditions including exposure times were consistent with control. Error bars indicate standard deviation. *, p < 0.05 compared to *spp-9p::*GFP reporter fluorescence 24 hours after L4 by one-way ANOVA with Tukey HSD post-hoc test. #, p < 0.05 compared to *spp-9p::*GFP reporter fluorescence 36 hours after L4 by one-way ANOVA with Tukey HSD post-hoc test. #, p < 0.05 compared to *spp-9p::*GFP reporter fluorescence 36 hours after L4 by one-way ANOVA with Tukey HSD post-hoc test.



Figure S2. *dbl-1* regulates endogenous *spp-9* expression levels. qRT-PCR analysis of A) *dbl-1* and B) *spp-9* gene expression in 48-hour post-L4 wild-type (WT, strain N2), *dbl-1* loss-of-function (*dbl-1(-)*, strain NU3), and DBL-1- overexpressing (*dbl-1(++)*, strain TLG182) strains. *dbl-1* expression was absent in *dbl-1(-)* animals (p = 0.0001) and increased 10-fold in *dbl-1(++)* animals (p = 0.0002). In *dbl-1(-)* animals, *spp-9* expression was increased 2.6-fold (p = 0.0287), whereas *spp-9* expression in *dbl-1(++)* animals is not significantly decreased (p = 0.1409). Experiment was performed in triplicate. Error bars represent standard deviation. *, p < 0.05 compared to wild-type by unpaired *t*-test.

CHAPTER V

DBL-1 REGULATES THE COMPOSITION OF THE PROTECTIVE SURFACE BARRIER OF *C. ELEGANS*

Introduction

All living organisms have external structures that protect the organisms against the environment including microbe infections. Roundworms (nematodes) also possess such protective structures, which make up the first line of defense against harmful organisms. The cuticle (exoskeleton) is a physical barrier that protects the nematodes from their external environment. Nematodes use the cuticle not only to protect themselves from the harmful environments but also help parasitic nematodes evade their hosts' responses^{1,2}. The cuticle is primarily made up of proteins including collagens and cuticlins. The outermost layer of the cuticle, the epicuticle, is rich in lipids. This layer is covered by a glycoprotein-rich layer called the surface coat (or glycocalyx). The protein composition of the protective cuticle is well characterized³. While the general lipid content has been profiled and conserved lipid metabolism pathways/enzymes have been identified in this organism, its epicuticular lipid composition has not yet been dissected^{4–6}.

It has been previously reported that the DBL-1/TGF- β signaling pathway regulates the cuticle barrier function. Loss of the ligand DBL-1 increases cuticular permeability to many drugs while overexpression of DBL-1 reduces permeability to drugs. The report also showed that the DBL-1/TGF- β signaling regulates

epicuticular lipid levels and loss of signaling alters cuticular layer organization⁷. Another study showed that altered DBL-1 levels reduce lipid stores and lipid droplet count and regulate lipid metabolism via the insulin-like signaling pathway⁸. Many transcriptional datasets indicate that DBL-1 signaling regulates several genes involved in lipid metabolism such as lipases, hydrolases, and esterases^{9–11}. The DBL-1/TGF- β signaling also is known to regulate expression of many cuticle collagens that comprise the protein-rich layer^{11,12}.

In this study, we characterized the lipid composition of the *C. elegans* epicuticle (surface-enriched) and propose a model in which DBL-1/TGF- β signaling transcriptionally affects epicuticle lipid composition. We identified that the relative levels of specific surface-enriched lipids are reduced by loss of DBL-1. We also observed a difference in the acyl moieties of the internal lipids of *dbl-1* knockout populations compared with wild-type populations. Furthermore, we demonstrate that loss of DBL-1 decreased expression of fatty acid synthase, *fasn-1*, and fatty acid desaturases, *fat-5* and *fat-7*. This study identified the specific lipid composition of the *C. elegans* surface barrier of both wild-type and *dbl-1* mutant animals helping to characterize the underlying molecular mechanisms for synthesis of surface barrier lipids regulated by DBL-1.

Materials and Methods

Strains and Maintenance

All *C. elegans* strains were maintained at 20°C on EZ media plates (0.55 g Tris-Cl, 0.24 g Tris base, 3.1 g BD Bacto[™] Peptone, 8 mg cholesterol, 2 g

sodium chloride, 20 g agar, in water to 1 L)¹³. The *C. elegans* strains were maintained on *E. coli* OP50 without contamination or starvation for at least five generations before every experiment. The *C. elegans* strains used were N2 (wild-type) and NU3 *dbl-1(nk3)* V (referred to as *dbl-1(-)* in this work).

Lipid Extraction

Asynchronized, *E. coli* OP50-fed populations of wild-type and *dbl-1(-)* animals at a concentration of about 250 animals/µl were harvested with ultrapure water. The animals were subjected to low speed centrifugation to form a pellet of approximately 0.5 ml. The pellet was resuspended in 0.5 ml water and 2 ml chloroform/methanol (2:1, v/v). The animals were vortexed for about 6–7 seconds, then centrifuged at low speed for 25 seconds. After centrifugation, the organic phase containing the surface-enriched lipid fraction was collected. The pellet was resuspended in 1 ml of chloroform. The surface-enriched lipid fraction (organic phase) was added to the first fraction. Following the extraction of surface-enriched lipids, 0.5 ml water and 2 ml of chloroform/methanol (2:1, v/v) was added to the tube containing the animal pellet. The pellet was crushed to extract remaining lipids using a Dounce homogenizer. The organic phase containing the internal lipid fraction was collected in a separate tube. To the original tube containing the animal pellet, 1 ml of chloroform was added and the internal lipid fraction was again collected. Heptadecanoic acid (2.5 µg) was added to the extracted lipids as the standard. The extracted lipids were then

used for thin-layer chromatography or were transmethylated for analysis by gas chromatography mass spectrometry (GC-MS).

Fatty Acid Methyl Ester Derivatization

Transmethylation was performed as previously described¹⁴. Briefly, 1 ml of 2.5% sulfuric acid in methanol was added to the extracted lipids and the mix was incubated at 80°C for 1 hour to transmethylate fatty acids. After the incubation, the tubes were cooled on ice and then 1 ml water and 0.5 ml hexane were added. The fatty acid methyl esters (FAMEs) were then extracted in the hexane layer after brief vortexing followed by centrifugation at low speed. The hexane layer was collected in gas chromatography (GC) vials without transferring the aqueous phase.

Gas Chromatography Mass Spectrometry (GC-MS) Injection

FAMEs were analyzed using an Agilent Intuvo 9000 gas chromatographer coupled with a 30m x 0.25mm x 0.25µm Agilent Intuvo GC column with helium as the carrier gas at a rate of 1.2 ml/min. The gas chromatogram was set at in initial temperature of 50°C for 0.3 min, followed by an increase of 250°C/min to 175°C, followed by an increase of 8°C/min to 190°C, with a 0.5 min hold time. This was followed by an increase of 10°C/min to 240°C with a 2 min hold time.

Thin Layer Chromatography (TLC)

Lipids were separated on silica gel plates using 25:25:25:10:9 (v/v) chloroform/methylacetate/1-propanol/methanol/0.25% potassium chloride solvent mix. Lipids were visualized on the silica plates by using 0.001% primuline dye dissolved in acetone. A phospholipid mix containing phosphatidic acid, phosphatidylethanolamine, phosphatidylinositol, phosphatidylcholine, and lysophosphosphatidylcholine was used as the standard (soy phospholipid mixture, Avanti Polar Lipids, Alabama).

RNA Isolation

Animals were synchronized by bleaching¹⁵. Total RNA was extracted from animals at 48 hours after the L4 stage. Total RNA was extracted by the freeze cracking method as previously described¹⁶.

Quantitative Reverse Transcription PCR

After RNA isolation, 2 µg of total RNA was primed with oligo(dT) and reverse transcribed to yield cDNA using the SuperScript III reverse transcriptase kit as per manufacturer's protocol (Invitrogen). Real-time PCR was performed on QuantStudio3 (Applied Biosystems by Thermo Fisher Scientific) instrument using the PowerUP SYBR Green master mix (Applied Biosystems) according to manufacturer's instructions. The experiment was performed in three technical replicates for each condition in at least three independent biological trials. QuantStudio Design and Analysis Software v1.5.1 was used to calculate raw Ct values and to normalize the values for the candidate genes to the housekeeping actin gene *act-1* (Applied Biosystems by Thermo Fisher Scientific). Fold change in gene expression between experimental sample and the wild-type control was determined with this software using the formula: $2^{(-\Delta\Delta C_t)}$. Experimental ΔC_t values

were compared with wild-type ΔC_t values using an unpaired *t*-test. Table 5.1 contains the list of all primer sequences.

Target			
gene	Forward primer	Reverse primer	
fasn-1	GGAAGCTGCCGTTCATTCGA	ATTCCATGCTCCTTGTGCCC	
fat-5	AGAGATCACCGGTGCCATCA	CGACTGGAATGAAGGTGGGC	
fat-7	ATGGGATGGCTTCTTGTGCG	CCAAGTGGCGTGAAGTGTGA	
act-1	GCCGGAATCCACGAGACTTC	TCTGGTGGGGGCGATGATCTT	

 Table 5.1. Primer sequences for real-time PCR.

Differential Expression Analysis by RNA Sequencing

Sequencing libraries were generated from 1 µg of wild-type and *dbl-1(-)* adult populations' RNA using NEBNext® RNA Library Prep Kit for Illumina® (NEB, USA) following manufacturer's recommendations. Novogene performed RNA sequencing of samples. Differential expression analysis of WT compared to *dbl-1(-)* populations grown on *E. coli* OP50 was performed using the DESeq R package (1.18.0). The resulting *p*-values were adjusted using the Benjamini and Hochberg's approach for controlling the false discovery rate¹⁷. Genes with an adjusted *p*-value < 0.05 found by DESeq were assigned as differentially expressed.

Results

Loss of DBL-1 Does Not Affect Total Lipid Varieties but Reduces Unsaturated Fatty Acid Quantity

Loss of DBL-1 reduced lipid levels of the *C. elegans* epicuticle⁷. To determine if loss of DBL-1 alters the global lipid (total lipid) composition (the types of lipids) in *C. elegans*, we compared fatty acid types and quantities of wild-type and *dbl-1(-)* populations. To quantitate the proportions/levels of and compare fatty acids of wild-type and *dbl-1(-)* animals, we extracted, transmethylated, and identified lipids by GC-MS. We observed fatty acids ranging from 12 to 21 carbon atoms in length in both wild-type and *dbl-1(-)* animals (see Fig. 5.1A). Despite the similar composition of fatty acids in both the populations and preparing similar masses of animal sample from both strains, we found that

the amount of total lipids of *dbl-1(-)* animals was reduced by 27% on average in comparison to the total lipids of wild-type animals. We further analyzed the difference between the saturated and unsaturated fatty acid levels of wild-type and *dbl-1(-)* strains. We found that there was a small, non-significant decrease in the level of *dbl-1(-)* saturated fatty acids. However, we observed a significant reduction in the quantity of unsaturated fatty acids in *dbl-1(-)* animals in comparison to wild-type animals (see Fig. 5.1B). These results indicate that loss of DBL-1 causes a reduction in the total amount of lipids, largely by reduction of unsaturated fatty acid species.





Fig. 5.1. Loss of DBL-1 results in reduction of unsaturated fatty acid levels but does not alter total lipid composition. Total lipids of wild-type and *dbl-1(-)* animals were extracted. The extracted lipids were transmethylated to obtain fatty acid methyl esters (FAMEs) and injected on the Agilent Intuvo 9000 GC-MS. A) Wild-type and *dbl-1(-)* populations total lipids are composed of fatty acids ranging from 12 to 21 carbons. The lipid composition of both the populations is similar. B) The fatty acids (FA) analyzed from total lipids of both the populations were sorted by their saturation levels. The amount of saturated fatty acids in wild-type and *dbl-1(-)* animals is similar. In *dbl-1(-)* animals, there is a significant reduction in the levels of unsaturated fatty acids. Comparison of saturated and unsaturated fatty acids between WT and *dbl-1(-)* populations is mean of at least 6 independent trials. Error bars represent standard deviation. *, *p* < 0.05, unpaired *t*-test.

C. elegans Surface Lipids Consist of Saturated and Unsaturated Fatty

Acids

To identify lipids of the C. elegans surface coat, we developed a protocol

to sequentially isolate surface-enriched lipids and internal lipids. The lipids that

were represented in higher amounts in the surface-enriched lipid fraction in

comparison to the internal lipid fraction were identified by GC-MS analysis and categorized as surface-enriched lipids. We found the presence of both saturated and unsaturated fatty acids in the surface coat of wild-type animals. These fatty acids ranged from 12 to 19 carbon atoms in length (see Fig. 5.2B). In conclusion, the protocol was effective at isolating different lipid populations in measurable quantities.

DBL-1 Affects Both Surface-Enriched and Internal Lipid Levels, Including Saturated and Unsaturated Fatty Acids

To determine if DBL-1 affects lipid composition of the surface coat, we performed thin layer chromatography using the surface-enriched and internal lipid fractions of wild-type and *dbl-1(-)* animals (see Fig. 5.2A). We observed similar lipid species in the surface-enriched fraction of wild-type and *dbl-1(-)* animals. Triacylglycerols, phosphatidic acid, and phosphatidylcholine were the major lipid species in the surface-enriched lipids of both wild-type and *dbl-1(-)* animals. These lipid species were also major components of the internal lipid fraction of both the populations. However, phosphatidylethanolamine, phosphatidylinositol, and lyso-phosphosphatidylcholine were also major lipid species in both wild-type and *dbl-1(-)* internal lipids, but not surface-enriched lipids. These results expand on previous reports that the epicuticular lipid is a subset of total lipid distinct from the lipids within the body⁴. Furthermore, these results indicate that DBL-1 does not alter the variety of lipid species of the surface coat or the internal lipids.

We then identified and compared the fatty acid methyl esters (FAMEs) of surface-enriched and internal lipid fractions of the wild-type and *dbl-1(-)* lipids by GC-MS (see Fig. 5.2B). While dodecanoic acid (12:0) levels remained unchanged between wild-type and *dbl-1(-)* surface-enriched lipid fractions, unsaturated fatty acid levels were significantly reduced in *dbl-1(-)* populations compared to the wild type. This indicates that loss of DBL-1 specifically reduces unsaturated fatty acid levels in the surface coat of *C. elegans*.

Additionally, in the internal lipid fraction of animals lacking DBL-1, there was a decrease in some fatty acids, not only monounsaturated and polyunsaturated fatty acids but also saturated fatty acids. These results indicate that loss of DBL-1 results in reduction in the amount of some internal fatty acids of animals (see Fig. 5.2C). These results collectively indicate that while DBL-1 does not alter the overall lipid types in the surface coat and internal lipids of animals, it does affect the proportions of unsaturated fatty acids comprising both the surface coat and the internal lipids of animals and saturated fatty acids of the internal lipids of animals.





Fig. 5.2. Identification of surface-enriched and internal lipids of wild-type and *dbl-1(-)* animals. A) Lipid species were separated by thin layer chromatography (TLC). Surface-enriched lipids and internal lipids were loaded on silica plates. Separation of lipid classes of total lipids by TLC indicates presence of similar lipid species in wild-type and *dbl-1(-)* populations. The phospholipid standard was used as a reference. This standard contains phosphatidic acid (PA), phosphatidylethanolamine (PE), phosphatidylinositol (PI), phosphatidylcholine (PC), and lyso-phosphosphatidylcholine (LPC). B) The composition of surface-enriched fatty acids of the wild-type and *dbl-1(-)* animals. C) The composition of internal fatty acids of the wild-type and *dbl-1(-)* animals.

DBL-1 Regulates Expression of Lipid Metabolism Genes

Because we identified reduction in levels of fatty acids upon loss of DBL-1, we determined if DBL-1 regulates lipid metabolism in adult animals by performing RNA sequencing (RNA-seq) and assessing changes in the transcription of lipid metabolism genes in wild-type and *dbl-1(-)* strains. Wild-type and *dbl-1(-)* populations were synchronized and harvested as two-day adults for RNA sequencing. We found reduction in the expression levels of lipid metabolism genes fasn-1, fat-5, fat-7, and F10D11.6 (see Table 5.2). fasn-1 is a fatty acid synthase gene required for the synthesis of long-chain saturated fatty acids¹⁸. fat-5 and *fat*-7 are Δ^9 fatty acid desaturases, which are required for the conversion of saturated fatty acids to unsaturated fatty acids^{19,20}. F10D11.6 is an ortholog of a human phospholipid transfer protein, but its function is uncharacterized²¹. Additionally, we found an increase in the expression levels of some lipid metabolism genes in animals lacking DBL-1, including oac-5, far-6, and acs-2 (see Table 5.2). acs-2 is fatty acid CoA synthetase, which is involved in the breakdown of fatty acids¹⁸. far-6 is a nematode-specific protein that can bind lipids, but the cellular function of FAR-6 is unknown²². oac-5 has O-acyl transferase activity and is predicted to add acyl groups to lipids (WormBase.org). To validate these findings, we performed real-time PCR to compare expression of select lipid metabolism genes in wild-type and *dbl-1(-)* animals. Consistent with RNA-seq results, expression of fasn-1, fat-5, and fat-7 was significantly decreased in animals lacking DBL-1 (see Fig. 5.3). These results indicate that

DBL-1 regulates expression of lipid metabolism genes, including *fasn-1*, *fat-5*, and *fat-7*.

RNA sequencing was performed with samples extracted from two-day adult populations of wild-type and *dbl-1(-)* in three independent trials. DBL-1 differentially regulates expression of lipid metabolism genes. Adjusted p < 0.05with Benjamini and Hochberg's approach for controlling the false discovery rate.

Gene name	Function	Read count WT	Read count <i>dbl-1(-)</i>	Log fold change <i>dbl-1(-)/</i> WT
fasn-1	Fatty acid synthesis	12104.75	7319.67	-0.73
fat-5	Fatty acid biosynthesis	934.87	360.00	-1.38
fat-7	Fatty acid biosynthesis	610.95	77.78	-2.97
F10D11.6	Ortholog of human phospholipid transfer protein	2126.86	772.81	-1.46
acs-2	β- oxidation	1942.28	3773.98	0.96
far-6	Lipid binding activity	1014.29	2677.87	1.40
oac-5	Acyl transferase activity	1.93	31.33	4.02



Fig. 5.3. DBL-1 positively regulates expression of some lipid metabolism genes. Relative gene expression levels of A) *fat-7*, B) *fat-5*, and C) *fasn-1* in wild-type and *dbl-1(-)* animals were compared. Data represents the average of at least three independent trials. Error bars represent standard deviation. *, p < 0.05 compared to wild-type by unpaired *t*-test.
Discussion

Roundworms protect themselves from their environments, in part, by secreting a lipid-rich epicuticle. In this study, we isolated and characterized the lipids enriched in the *C. elegans* epicuticle. We identified that the epicuticle is composed of triglycerides, phosphatidylethanolamine, phosphatidylcholine, and phosphatidic acid, adding to previous reports⁴. Additionally, we found that the DBL-1/TGF- β signaling pathway regulates levels and composition of lipids, but not the lipid classes, for both secreted (epicuticular lipids) and inside the body (internal lipids), affecting primarily the unsaturated fatty acids. These results provide an unprecedented level of detail to previous findings that demonstrate DBL-1 signaling has important roles in epicuticle barrier integrity and lipid homeostasis^{7,8}.

We also identified a possible mechanism for how DBL-1 signaling affects lipid levels and composition: DBL-1 pathway signaling may regulate expression of lipid metabolism genes in the hypodermis, which makes and secretes epicuticle lipids. Specifically, expression of fatty acid synthase gene *fasn-1* and fatty acid desaturase genes *fat-5* and *fat-7*, at least, was significantly reduced in populations lacking DBL-1. The DBL-1-dependent reduction of the unsaturated fatty acid desaturases and fatty acid synthase in animals lacking DBL-1. These findings reveal molecular mechanisms by which DBL-1 may regulate the protective surface barrier in *C. elegans* (modelled in Fig. 5.4). Furthermore, RNA-

seq analysis revealed that expression of F10D11.6, an ortholog of human phospholipid transfer protein, was reduced in absence of DBL-1. F10D11.6 is predicted to enable lipid binding activity and may play a role in the transport of lipids to the cuticle surface²¹. Determining if F10D11.6 plays a role in regulating *C. elegans* lipid composition could lead to novel insights into how the epicuticle is formed and how DBL-1 signaling protects animals from environmental threats.



This work also suggests a mechanism for the observations that DBL-1 signaling is required for normal barrier function of the cuticle. Animals lacking DBL-1 have increased cuticle permeability to some anesthetics and animals with increased DBL-1 pathway signaling are more resistant to anesthetics, which correlates with transmission electron microscopy analyses that show dose-dependent effects of DBL-1 on surface lipid levels, cuticle structure, and cuticle organization⁷. Our findings indicate that DBL-1 signaling regulates transcription

of lipid metabolism genes including *fasn-1*, *fat-5*, and *fat-7* and we proposed that reduced expression of these target genes in animals lacking DBL-1 contributes to decreased unsaturated fatty acids of the epicuticle, reducing surface lipid level, altering cuticle structure, and increasing cuticle permeability. Future studies may identify other targets of DBL-1 signaling that contribute to surface barrier function and confer protection to animals against different pathogens and environmental challenges.

CHAPTER VI

CONCLUSIONS AND FUTURE DIRECTIONS

Collectively, the work presented in this dissertation supports a model in which the DBL-1/TGF- β signaling pathway plays an important role in protective host responses against a variety of bacteria (see Fig. 6.1). We propose that DBL-1/TGF- β signaling is involved in sensing pathogens and tailoring immune responses at behavioral, molecular, and physiological levels.





The findings of this work indicate that DBL-1/TGF- β signaling coordinates

differential host responses at a behavioral level (Aim 1). Work reported in

Chapters II and III expands the current understanding of how C. elegans

coordinate organismal defense responses upon sensing harmful environments.

Our findings indicate that DBL-1 is a major player in orchestrating these robust,

protective, and specific responses to different types of bacteria. We identified that

DBL-1 signaling is required to suppress avoidance to the tested Gram-negative,

but not Gram-positive bacteria. This finding broadens our understanding of the known role of the DBL-1 pathway signaling in suppressing avoidance of *E. coli*¹. Animals lacking DBL-1 were susceptible to both Gram-negative and Gram-positive bacteria. We reported that damage to the intestine was not the underlying cause of this susceptibility. Reduced bacterial intake was dependent on both loss of DBL-1 and exposure to select bacteria. Reduced brood size was also dependent on both loss of DBL-1 is involved in such differential behavioral responses. Previous studies have reported that DBL-1 secretion from the AVA interneurons activates DBL-1 signaling in the hypodermal tissue to mediate aversive learning in response to pathogenic *P. aeruginosa*². The neuronal source of DBL-1 might explain its role in sensing the pathogenic environments. However, the downstream physiological targets of this circuit that result in the different DBL-1-mediated responses remain to be identified.

A major finding of this work is the differential requirement of the SMAD machinery to coordinate the avoidance response to test bacteria. Previous studies report the requirement of SMA-3 but not SMA-2 and SMA-4 to regulate transcription of antimicrobial genes against a fungal pathogen, *D. coniospora*³. We identified that canonical DBL-1 signaling suppresses avoidance to Gramnegative bacteria but not to Gram-positive bacteria. We identified a novel role of SMA-4 in suppressing avoidance independent of DBL-1. This is indicative of SMA-4 partnering with a different transcription factor, hinting to a potential

crosstalk with other immune signaling pathways (see Fig. 6.1B). PMK-1/MAPK signaling may be involved in regulating it as SMA-4 is predicted to genetically interact with PMK-1⁴. Interestingly, ATF-7, a transcription factor activated by PMK-1, is required for downregulation of *sma-4*—but not other DBL-1 pathway component genes—in wild-type animals exposed to Gram-negative *P. aeruginosa* PA14⁵. It will be of interest to discover if SMA-4 plays a broader role in the innate immune response than acting in the DBL-1 pathway. In the future, studies focusing on the interplay between multiple signaling pathways to regulate specific immune responses will be important to understand the dynamic immune responses of organisms.

Furthermore, this work reports that DBL-1 differential tailors host responses at a molecular level (Aim 2). Work reported in Chapters II and IV expands the current understanding of how *C. elegans* coordinate molecular defenses in response to environmental challenges^{6,7}. This work reveals signature molecular responses regulated by DBL-1/TGF- β in response to a variety of bacterial pathogens. Our findings support and add to the "antimicrobial fingerprint" proposed by Alper⁸. The results indicate that DBL-1 tailors antimicrobial gene expression depending on the type of bacterial exposure. These findings strengthen our model that canonical DBL-1/TGF- β signaling is activated in response to Gram-negative bacteria, thereby coordinating specific organismal behaviors and expression of innate immunity genes. On the other hand, canonical DBL-1/TGF- β signaling is repressed upon exposure to Gram-

positive bacteria. Interestingly, our results indicate that SMA-4 potentially crosstalks with other immune signaling pathways. This suggests that exposure to Gram-positive bacteria results in DBL-1-independent or SMA-4-dependent organismal behaviors and regulation of innate immunity gene expression. Future work will identify if ATF-7 or other transcription factors partner with SMA-4 to regulate these immune responses. It will be of interest to determine the bacterial components resulting in these differential host responses.

Our findings summarized in Chapter IV provide a snapshot of the dynamic crosstalk between multiple immune signaling pathways converging on the genetic regulation of a caenopore, *spp-9*⁹. This work revealed further insights into the environmental responsiveness of the caenopore family and how organisms generate dynamic, complex, targeted molecular responses to physiological challenges. We identified not only that spp-9 is regulated by DBL-1/TGF- β depending on the environment, but also that insulin-like and p38/MAP kinase signaling pathways regulate *spp-9* expression. The DBL-1-independent regulation of *spp-9* in response to select test bacteria may require SMA-4, as suggested by our finding reported in Chapter II. Our results support a role for *spp-9* as part of the distinct "antimicrobial fingerprint" response upon exposure to different pathogens and suggest it also plays a role in stress response. Expression of *spp-9* in some head sensory neurons suggests a role in sensing different bacterial stimuli that may be independent of DBL-1 because DBL-1 receptors are not expressed on these cells. Identifying the role of spp-9 in this

neuro-immune circuit will provide novel insights into communication along the gut-brain axis within the context of immunity.

Finally, this work reports the role of DBL-1 in regulating host defenses at a physiological level (Aim 3). The findings of Chapter V reveal the lipid composition of the *C. elegans* protective epicuticle. We identified that DBL-1/TGF- β signaling regulates epicuticular lipid composition, specifically the unsaturated fatty acids. Additionally, the total lipid composition—but not the general lipid classes—was regulated by DBL-1/TGF- β signaling. These findings also reveal a mechanism by which DBL-1 regulates the protective surface barrier in *C. elegans*. The DBL-1dependent reduction in the unsaturated fatty acids of the epicuticle can be partly attributed to the decrease in expression of fatty acid synthase (fasn-1) and fatty acid desaturases (fat-5 and fat-7) in animals lacking DBL-1. These results build the foundation to understanding the biochemistry of the protective epicuticle of the free-living nematode *C. elegans* and how DBL-1/TGF-β signaling confer protection to animals from different pathogenic bacteria by means of surface barrier (epicuticle). Overall, these findings indicate an important role of DBL-1 in regulating lipid metabolism in congruence with previous work¹⁰. Previous reports demonstrate that alterations in lipid metabolism modifies innate immune responses in *C. elegans*^{11–13}. These studies and our collective findings hint to a link between lipid metabolism and innate immune regulation controlled by the DBL-1 signaling pathway. Future work will determine if DBL-1-mediated tailored

host defenses—behavioral, molecular, and physiological—require DBL-1dependent lipid metabolism changes.

In conclusion, this work provides novel insights into how DBL-1/TGF- β signaling is altered in response to different pathogens and in turn appropriately coordinates expression of antimicrobial genes to fight pathogens and keep host immune responses in check. We propose a model in which animals use DBL-1/TGF- β signaling to tailor specific organismal responses upon sensing different pathogenic environments (see Fig. 6.1). DBL-1/TGF- β signaling is altered in response to different pathogens and in turn appropriately coordinates expression of antimicrobial genes to fight pathogens and keep host immune responses in check. Our work establishes a strong system in which to characterize dynamic and differential host immune responses *in vivo*. These findings lay the foundation to study the role of TGF- β signaling coordinated with other immune signaling pathways in orchestrating differential host responses against pathogenic bacteria.

REFERENCES

Chapter I

- Medzhitov, R. & Janeway, C. A., Jr. Innate immune recognition and control of adaptive immune responses. *Semin Immunol* **10**, 351–353, https://doi.org/10.1006/smim.1998.0136 (1998).
- Riera Romo, M., Perez-Martinez, D. & Castillo Ferrer, C. Innate immunity in vertebrates: an overview. *Immunology* 148, 125–139, https://doi.org/10.1111/imm.12597 (2016).
- 3 Smith, N. C., Rise, M. L. & Christian, S. L. A comparison of the innate and adaptive immune systems in cartilaginous fish, ray-finned fish, and lobe-finned fish. *Front Immunol* **10**, 2292,

https://doi.org/10.3389/fimmu.2019.02292 (2019).

- Bonilla, F. A. & Oettgen, H. C. Adaptive immunity. *J Allergy Clin Immunol* **125**, S33–40, https://doi.org/10.1016/j.jaci.2009.09.017 (2010).
- Marshall, J. S., Warrington, R., Watson, W. & Kim, H. L. An introduction to immunology and immunopathology. *Allergy Asthma Clin Immunol* 14, 49, https://doi.org/10.1186/s13223-018-0278-1 (2018).
- Kumar, A. et al. *Caenorhabditis elegans*: a model to understand hostmicrobe interactions. *Cell Mol Life Sci* 77, 1229–1249, https://doi.org/10.1007/s00018-019-03319-7 (2020).

- 7 Engelmann, I. & Pujol, N. Innate immunity in *C. elegans. Adv Exp Med Biol* 708, 105–121, https://doi.org/10.1007/978-1-4419-8059-5_6 (2010).
- 8 Blaxter, M. & Denver, D. R. The worm in the world and the world in the worm. *BMC Biol* **10**, 57, https://doi.org/10.1186/1741-7007-10-57 (2012).
- Blaxter, M. & Koutsovoulos, G. The evolution of parasitism in Nematoda.
 Parasitology 142, S26–39, https://doi.org/10.1017/S0031182014000791
 (2015).
- Murfin, K. E., Chaston, J. & Goodrich-Blair, H. Visualizing bacteria in nematodes using fluorescent microscopy. *J Vis Exp*, https://doi.org/10.3791/4298 (2012).
- Samuel, B. S., Rowedder, H., Braendle, C., Felix, M. A. & Ruvkun, G.
 Caenorhabditis elegans responses to bacteria from its natural habitats.
 Proc Natl Acad Sci U S A **113**, E3941–3949,
 https://doi.org/10.1073/pnas.1607183113 (2016).
- 12 Barker, K. R. et al. Plant and soil nematodes: Societal impact and focus for the future. *J Nematol* **26**, 127–137 (1994).
- Kim, D. Studying host-pathogen interactions and innate immunity in *Caenorhabditis elegans. Dis Model Mech* 1, 205–208, https://doi.org/10.1242/dmm.000265 (2008).
- 14 Stutz, K., Kaech, A., Aebi, M., Kunzler, M. & Hengartner, M. O. Disruption of the *C. elegans* intestinal brush border by the fungal lectin CCL2

phenocopies dietary lectin toxicity in mammals. *PLoS One* **10**, e0129381, https://doi.org/10.1371/journal.pone.0129381 (2015).

- Schulenburg, H., Kurz, C. L. & Ewbank, J. J. Evolution of the innate immune system: The worm perspective. *Immunol Rev* 198, 36–58, https://doi.org/10.1111/j.0105-2896.2004.0125.x (2004).
- 16 Pradel, E. et al. Detection and avoidance of a natural product from the pathogenic bacterium Serratia marcescens by Caenorhabditis elegans. Proc Natl Acad Sci U S A 104, 2295–2300, https://doi.org/10.1073/pnas.0610281104 (2007).
- 17 Meisel, J. D., Panda, O., Mahanti, P., Schroeder, F. C. & Kim, D. H. Chemosensation of bacterial secondary metabolites modulates neuroendocrine signaling and behavior of *C. elegans. Cell* **159**, 267–280, https://doi.org/10.1016/j.cell.2014.09.011 (2014).
- Meisel, J. D. & Kim, D. H. Behavioral avoidance of pathogenic bacteria by *Caenorhabditis elegans. Trends Immunol* 35, 465–470, https://doi.org/10.1016/j.it.2014.08.008 (2014).
- Li, Q. & Liberles, S. D. Aversion and attraction through olfaction. *Curr Biol*25, R120–R129, https://doi.org/10.1016/j.cub.2014.11.044 (2015).
- Battisti, J. M. et al. Analysis of the *Caenorhabditis elegans* innate immune response to *Coxiella burnetii*. *Innate Immun* 23, 111–127, https://doi.org/10.1177/1753425916679255 (2017).

- 21 Beale, E., Li, G., Tan, M. W. & Rumbaugh, K. P. *Caenorhabditis elegans* senses bacterial autoinducers. *Appl Environ Microbiol* **72**, 5135–5137, https://doi.org/10.1128/AEM.00611-06 (2006).
- 22 Anderson, A. & McMullan, R. Neuronal and non-neuronal signals regulate Caenorhabditis elegans avoidance of contaminated food. Philos Trans R Soc Lond B Biol Sci 373, https://doi.org/10.1098/rstb.2017.0255 (2018).
- 23 Hodgkin, J., Kuwabara, P. E. & Corneliussen, B. A novel bacterial pathogen, *Microbacterium nematophilum*, induces morphological change in the nematode *C. elegans. Curr Biol* **10**, 1615–1618, https://doi.org/10.1016/s0960-9822(00)00867-8 (2000).
- Tan, M. W. Identification of host and pathogen factors involved in virulence using *Caenorhabditis elegans*. *Methods Enzymol* 358, 13–28, https://doi.org/10.1016/s0076-6879(02)58078-2 (2002).
- Tan, M. W., Mahajan-Miklos, S. & Ausubel, F. M. Killing of *Caenorhabditis* elegans by *Pseudomonas aeruginosa* used to model mammalian bacterial pathogenesis. *Proc Natl Acad Sci U S A* 96, 715–720, https://doi.org/10.1073/pnas.96.2.715 (1999).
- Garsin, D. A. et al. A simple model host for identifying Gram-positive virulence factors. *Proc Natl Acad Sci U S A* 98, 10892–10897, https://doi.org/10.1073/pnas.191378698 (2001).

- Kurz, C. L. et al. Virulence factors of the human opportunistic pathogen
 Serratia marcescens identified by in vivo screening. EMBO J 22, 1451–
 1460, https://doi.org/10.1093/emboj/cdg159 (2003).
- 28 Page, A. P. & Johnstone, I. L. The cuticle. *WormBook, The C. elegans Research Community,* https://doi.org/10.1895/ wormbook.1.138.1 (2007).
- Couillault, C. & Ewbank, J. J. Diverse bacteria are pathogens of Caenorhabditis elegans. Infect Immun 70, 4705–4707, https://doi.org/10.1128/iai.70.8.4705-4707.2002 (2002).
- Gravato-Nobre, M. J. et al. Multiple genes affect sensitivity of
 Caenorhabditis elegans to the bacterial pathogen *Microbacterium nematophilum*. *Genetics* 171, 1033–1045,
 https://doi.org/10.1534/genetics.105.045716 (2005).
- Begun, J. et al. Staphylococcal biofilm exopolysaccharide protects against
 Caenorhabditis elegans immune defenses. *PLoS Pathog* 3, e57,
 https://doi.org/10.1371/journal.ppat.0030057 (2007).
- Wong, D., Bazopoulou, D., Pujol, N., Tavernarakis, N. & Ewbank, J. J.
 Genome-wide investigation reveals pathogen-specific and shared signatures in the response of *Caenorhabditis elegans* to infection.
 Genome Biol 8, R194, https://doi.org/10.1186/gb-2007-8-9-r194 (2007).
- Singh, V. & Aballay, A. Regulation of DAF-16-mediated Innate Immunity in Caenorhabditis elegans. J Biol Chem 284, 35580–35587, https://doi.org/10.1074/jbc.M109.060905 (2009).

- Zugasti, O. & Ewbank, J. J. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. *Nat Immunol* 10, 249–256, https://doi.org/10.1038/ni.1700 (2009).
- Alper, S., McBride, S. J., Lackford, B., Freedman, J. H. & Schwartz, D. A.
 Specificity and complexity of the *Caenorhabditis elegans* innate immune response. *Mol Cell Biol* 27, 5544–5553, https://doi.org/10.1128/MCB.02070-06 (2007).
- Berg, M. et al. TGFβ/BMP immune signaling affects abundance and function of *C. elegans* gut commensals. *Nat Commun* 10, 604, https://doi.org/10.1038/s41467-019-08379-8 (2019).
- Kim, D. H. et al. A conserved p38 MAP kinase pathway in *Caenorhabditis* elegans innate immunity. *Science* 297, 623–626, https://doi.org/10.1126/science.1073759 (2002).
- Mallo, G. V. et al. Inducible antibacterial defense system in *C. elegans*.
 Curr Biol **12**, 1209–1214, https://doi.org/10.1016/s0960-9822(02)00928-4
 (2002).
- Murphy, C. T. et al. Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans*. *Nature* 424, 277–283, https://doi.org/10.1038/nature01789 (2003).

- Garsin, D. A. et al. Long-lived *C. elegans daf-2* mutants are resistant to bacterial pathogens. *Science* **300**, 1921, https://doi.org/10.1126/science.1080147 (2003).
- Padgett, R. W., Das, P. & Krishna, S. TGF-β signaling, Smads, and tumor suppressors. *Bioessays* 20, 382–390, https://doi.org/10.1002/(SICI)1521-1878(199805)20:5<382::AID-BIES5>3.0.CO;2-Q (1998).
- 42 Krishna, S., Maduzia, L. L. & Padgett, R. W. Specificity of TGF-β signaling is conferred by distinct type I receptors and their associated SMAD proteins in *Caenorhabditis elegans*. *Development* **126**, 251–260 (1999).
- Gumienny, T. L. & Savage-Dunn, C. TGF-β signaling in *C. elegans*.
 WormBook, The C. elegans Research Community, 1–34, https://doi.org/10.1895/wormbook.1.22.2 (2013).
- Savage, C. et al. *Caenorhabditis elegans* genes *sma-2*, *sma-3*, and *sma-4* define a conserved family of transforming growth factor beta pathway components. *Proc Natl Acad Sci U S A* **93**, 790–794, https://doi.org/10.1073/pnas.93.2.790 (1996).
- Suzuki, Y. et al. A BMP homolog acts as a dose-dependent regulator of body size and male tail patterning in *Caenorhabditis elegans*.
 Development 126, 241–250 (1999).
- 46 Luo, S., Shaw, W. M., Ashraf, J. & Murphy, C. T. TGF-β Sma/Mab signaling mutations uncouple reproductive aging from somatic aging.

PLoS Genet **5**, e1000789, https://doi.org/10.1371/journal.pgen.1000789 (2009).

- Schultz, R. D., Bennett, E. E., Ellis, E. A. & Gumienny, T. L. Regulation of extracellular matrix organization by BMP signaling in *Caenorhabditis elegans*. *PLoS One* 9, e101929, https://doi.org/10.1371/journal.pone.0101929 (2014).
- Mochii, M., Yoshida, S., Morita, K., Kohara, Y. & Ueno, N. Identification of transforming growth factor-β regulated genes in *Caenorhabditis elegans* by differential hybridization of arrayed cDNAs. *Proc Natl Acad Sci U S A*96, 15020–15025, https://doi.org/10.1073/pnas.96.26.15020 (1999).
- Liang, J., Yu, L., Yin, J. & Savage-Dunn, C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related signaling outputs. *Dev Biol* 305, 714–725, https://doi.org/10.1016/j.ydbio.2007.02.038 (2007).
- Roberts, A. F., Gumienny, T. L., Gleason, R. J., Wang, H. & Padgett, R.
 W. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. *BMC Dev Biol* 10, 61, https://doi.org/10.1186/1471-213X-10-61 (2010).
- Lakdawala, M. F. et al. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body size-associated genes in *Caenorhabditis elegans*.
 Mol Biol Cell 30, 3151–3160, https://doi.org/10.1091/mbc.E19-09-0500 (2019).

Zhang, X. & Zhang, Y. DBL-1, a TGF-β, is essential for *Caenorhabditis elegans* aversive olfactory learning. *Proc Natl Acad Sci U S A* 109, 17081–17086, https://doi.org/10.1073/pnas.1205982109 (2012).

Chapter II

- Medzhitov, R. & Janeway, C. A., Jr. Innate immune recognition and control of adaptive immune responses. *Semin Immunol* **10**, 351–353, https://doi.org/10.1006/smim.1998.0136 (1998).
- 2 MacGillivray, D. M. & Kollmann, T. R. The role of environmental factors in modulating immune responses in early life. *Front Immunol* 5, 434, https://doi.org/10.3389/fimmu.2014.00434 (2014).
- Cheesman, H. K. et al. Aberrant activation of p38 MAP kinase-dependent innate immune responses is toxic to *Caenorhabditis elegans*. *G3* (*Bethesda*) 6, 541–549, https://doi.org/10.1534/g3.115.025650 (2016).
- Pukkila-Worley, R. Surveillance immunity: An emerging paradigm of innate defense activation in *Caenorhabditis elegans*. *PLoS Pathog* 12, e1005795, https://doi.org/10.1371/journal.ppat.1005795 (2016).
- Sellegounder, D., Yuan, C. H., Wibisono, P., Liu, Y. & Sun, J.
 Octopaminergic signaling mediates neural regulation of innate immunity in *Caenorhabditis elegans. mBio* 9, https://doi.org/10.1128/mBio.01645-18 (2018).

- Akira, S., Uematsu, S. & Takeuchi, O. Pathogen recognition and innate immunity. *Cell* 124, 783–801, https://doi.org/10.1016/j.cell.2006.02.015 (2006).
- 7 Engelmann, I. & Pujol, N. Innate immunity in *C. elegans. Adv Exp Med Biol* 708, 105–121, https://doi.org/10.1007/978-1-4419-8059-5_6 (2010).
- Couillault, C. & Ewbank, J. J. Diverse bacteria are pathogens of *Caenorhabditis elegans. Infect Immun* 70, 4705–4707, https://doi.org/10.1128/iai.70.8.4705-4707.2002 (2002).
- Gravato-Nobre, M. J. et al. Multiple genes affect sensitivity of
 Caenorhabditis elegans to the bacterial pathogen *Microbacterium nematophilum*. *Genetics* 171, 1033–1045,
 https://doi.org/10.1534/genetics.105.045716 (2005).
- Begun, J. et al. Staphylococcal biofilm exopolysaccharide protects against
 Caenorhabditis elegans immune defenses. *PLoS Pathog* 3, e57,
 https://doi.org/10.1371/journal.ppat.0030057 (2007).
- 11 Wong, D., Bazopoulou, D., Pujol, N., Tavernarakis, N. & Ewbank, J. J. Genome-wide investigation reveals pathogen-specific and shared signatures in the response of *Caenorhabditis elegans* to infection. *Genome Biol* 8, R194, https://doi.org/10.1186/gb-2007-8-9-r194 (2007).
- Singh, V. & Aballay, A. Regulation of DAF-16-mediated Innate Immunity in Caenorhabditis elegans. J Biol Chem 284, 35580–35587, https://doi.org/10.1074/jbc.M109.060905 (2009).

- 13 Zugasti, O. & Ewbank, J. J. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. *Nat Immunol* **10**, 249–256, https://doi.org/10.1038/ni.1700 (2009).
- 14 Pukkila-Worley, R. et al. Stimulation of host immune defenses by a small molecule protects *C. elegans* from bacterial infection. *PLoS Genet* 8, e1002733, https://doi.org/10.1371/journal.pgen.1002733 (2012).
- Ahamefule, C. S. et al. *Caenorhabditis elegans*-based *Aspergillus fumigatus* infection model for evaluating pathogenicity and drug efficacy.
 Front Cell Infect Microbiol **10**, 320, https://doi.org/10.3389/fcimb.2020.00320 (2020).
- Alper, S., McBride, S. J., Lackford, B., Freedman, J. H. & Schwartz, D. A.
 Specificity and complexity of the *Caenorhabditis elegans* innate immune response. *Mol Cell Biol* 27, 5544–5553, https://doi.org/10.1128/MCB.02070-06 (2007).
- Berg, M. et al. TGFβ/BMP immune signaling affects abundance and function of *C. elegans* gut commensals. *Nat Commun* 10, 604, https://doi.org/10.1038/s41467-019-08379-8 (2019).
- Pujol, N. et al. A reverse genetic analysis of components of the Toll signaling pathway in *Caenorhabditis elegans*. *Curr Biol* 11, 809–821, https://doi.org/10.1016/s0960-9822(01)00241-x (2001).

- Kim, D. H. et al. A conserved p38 MAP kinase pathway in *Caenorhabditis* elegans innate immunity. *Science* 297, 623–626, https://doi.org/10.1126/science.1073759 (2002).
- Mallo, G. V. et al. Inducible antibacterial defense system in *C. elegans*.
 Curr Biol **12**, 1209–1214, https://doi.org/10.1016/s0960-9822(02)00928-4
 (2002).
- Murphy, C. T. et al. Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans*. *Nature* 424, 277–283, https://doi.org/10.1038/nature01789 (2003).
- Garsin, D. A. et al. Long-lived *C. elegans daf-2* mutants are resistant to bacterial pathogens. *Science* 300, 1921, https://doi.org/10.1126/science.1080147 (2003).
- Mochii, M., Yoshida, S., Morita, K., Kohara, Y. & Ueno, N. Identification of transforming growth factor-β regulated genes in *Caenorhabditis elegans* by differential hybridization of arrayed cDNAs. *Proc Natl Acad Sci U S A* 96, 15020–15025, https://doi.org/10.1073/pnas.96.26.15020 (1999).
- Liang, J., Yu, L., Yin, J. & Savage-Dunn, C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related signaling outputs. *Dev Biol* 305, 714–725, https://doi.org/10.1016/j.ydbio.2007.02.038 (2007).
- Roberts, A. F., Gumienny, T. L., Gleason, R. J., Wang, H. & Padgett, R.W. Regulation of genes affecting body size and innate immunity by the

DBL-1/BMP-like pathway in *Caenorhabditis elegans*. *BMC Dev Biol* **10**, 61, https://doi.org/10.1186/1471-213X-10-61 (2010).

- Tan, M. W., Mahajan-Miklos, S. & Ausubel, F. M. Killing of *Caenorhabditis* elegans by *Pseudomonas aeruginosa* used to model mammalian bacterial pathogenesis. *Proc Natl Acad Sci U S A* 96, 715–720, https://doi.org/10.1073/pnas.96.2.715 (1999).
- Zhang, X. & Zhang, Y. DBL-1, a TGF-β, is essential for *Caenorhabditis* elegans aversive olfactory learning. *Proc Natl Acad Sci U S A* 109, 17081–17086, https://doi.org/10.1073/pnas.1205982109 (2012).
- 28 Madhu, B., Salazar, A. E. & Gumienny, T. L. Caenorhabditis elegans egglaying and brood-size changes upon exposure to Serratia marcescens and Staphylococcus epidermidis are independent of DBL-1 signaling. microPublication Biology 2019, 10.17912/2r51-b476 (2019).
- 29 Sifri, C. D., Begun, J., Ausubel, F. M. & Calderwood, S. B. Caenorhabditis elegans as a model host for Staphylococcus aureus pathogenesis. Infect Immun 71, 2208–2217, https://doi.org/10.1128/iai.71.4.2208-2217.2003 (2003).
- 30 Reddy, K. C., Andersen, E. C., Kruglyak, L. & Kim, D. H. A polymorphism in *npr-1* is a behavioral determinant of pathogen susceptibility in *C. elegans. Science* **323**, 382–384, https://doi.org/10.1126/science.1166527 (2009).

- Amrit, F. R., Ratnappan, R., Keith, S. A. & Ghazi, A. The *C. elegans*lifespan assay toolkit. *Methods* 68, 465–475,
 https://doi.org/10.1016/j.ymeth.2014.04.002 (2014).
- Clark, J. F., Meade, M., Ranepura, G., Hall, D. H. & Savage-Dunn, C.
 Caenorhabditis elegans DBL-1/BMP regulates lipid accumulation via interaction with insulin signaling. *G3 (Bethesda)* 8, 343–351, https://doi.org/10.1534/g3.117.300416 (2018).
- Kissoyan, K. A. B. et al. Natural *C. elegans* microbiota protects against infection via production of a cyclic lipopeptide of the viscosin group. *Curr Biol* 29, 1030–1037 e1035, https://doi.org/10.1016/j.cub.2019.01.050 (2019).
- Chang, H. C., Paek, J. & Kim, D. H. Natural polymorphisms in *C. elegans*HECW-1 E3 ligase affect pathogen avoidance behaviour. *Nature* 480,
 525–529, https://doi.org/10.1038/nature10643 (2011).
- 35 Stiernagle, T. Maintenance of *C. elegans. WormBook, The C. elegans Research Community* (2006).
- 36 Portman, D. S. Profiling *C. elegans* gene expression with DNA microarrays. *WormBook, The C. elegans Research Community* (2006).
- 37 Love, M. I., Huber, W. & Anders, S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. *Genome Biol* **15**, 550, https://doi.org/10.1186/s13059-014-0550-8 (2014).

- Madhu, B., Lakdawala, M. F., Issac, N. G. & Gumienny, T. L.
 Caenorhabditis elegans saposin-like *spp-9* is involved in specific innate immune responses. *Genes Immun* 21, 301–310, https://doi.org/10.1038/s41435-020-0108-6 (2020).
- 39 Savage-Dunn, C., Gleason, R. J., Liu, J. & Padgett, R. W. Mutagenesis and imaging studies of BMP signaling mechanisms in *C. elegans*. *Methods Mol Biol* **1891**, 51–73, https://doi.org/10.1007/978-1-4939-8904-1_6 (2019).
- Samuel, B. S., Rowedder, H., Braendle, C., Felix, M. A. & Ruvkun, G.
 Caenorhabditis elegans responses to bacteria from its natural habitats.
 Proc Natl Acad Sci U S A 113, E3941–3949,
 https://doi.org/10.1073/pnas.1607183113 (2016).
- Tenor, J. L. & Aballay, A. A conserved Toll-like receptor is required for *Caenorhabditis elegans* innate immunity. *EMBO Rep* 9, 103–109, https://doi.org/10.1038/sj.embor.7401104 (2008).
- Lakowski, B. & Hekimi, S. The genetics of caloric restriction in
 Caenorhabditis elegans. Proc Natl Acad Sci U S A 95, 13091–13096, https://doi.org/10.1073/pnas.95.22.13091 (1998).
- Greer, E. L. & Brunet, A. Different dietary restriction regimens extend lifespan by both independent and overlapping genetic pathways in *C. elegans. Aging Cell* 8, 113–127, https://doi.org/10.1111/j.1474-9726.2009.00459.x (2009).

- Gelino, S. et al. Intestinal autophagy improves healthspan and longevity in
 C. elegans during dietary restriction. *PLoS Genet* 12, e1006135,
 https://doi.org/10.1371/journal.pgen.1006135 (2016).
- Pradel, E. et al. Detection and avoidance of a natural product from the pathogenic bacterium *Serratia marcescens* by *Caenorhabditis elegans*.
 Proc Natl Acad Sci U S A **104**, 2295–2300, https://doi.org/10.1073/pnas.0610281104 (2007).
- Beale, E., Li, G., Tan, M. W. & Rumbaugh, K. P. *Caenorhabditis elegans* senses bacterial autoinducers. *Appl Environ Microbiol* 72, 5135–5137, https://doi.org/10.1128/AEM.00611-06 (2006).
- 47 Anderson, A. & McMullan, R. Neuronal and non-neuronal signals regulate *Caernorhabditis elegans* avoidance of contaminated food. *Philos Trans R Soc Lond B Biol Sci* **373**, https://doi.org/10.1098/rstb.2017.0255 (2018).
- Olofsson, B. The olfactory neuron AWC promotes avoidance of normally palatable food following chronic dietary restriction. *J Exp Biol* 217, 1790–1798, https://doi.org/10.1242/jeb.099929 (2014).
- 49 Savage, C. et al. *Caenorhabditis elegans* genes *sma-2*, *sma-3*, and *sma-4* define a conserved family of transforming growth factor beta pathway components. *Proc Natl Acad Sci U S A* **93**, 790–794, https://doi.org/10.1073/pnas.93.2.790 (1996).

- Tian, C. et al. The RGM protein DRAG-1 positively regulates a BMP-like signaling pathway in *Caenorhabditis elegans*. *Development* 137, 2375–2384, https://doi.org/10.1242/dev.051615 (2010).
- 51 Lakdawala, M. L. et al. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body size-associated genes in *Caenorhabditis elegans*. *Molecular Biology of the Cell* **30**(26), 3151–3160, https://doi.org/10.1091/mbc.E19-09-0500 (2019).
- Holdorf, A. D. et al. WormCat: An online tool for annotation and
 visualization of *Caenorhabditis elegans* genome-scale data. *Genetics* 214, 279–294, https://doi.org/10.1534/genetics.119.302919 (2020).
- Shapira, M. et al. A conserved role for a GATA transcription factor in regulating epithelial innate immune responses. *Proc Natl Acad Sci U S A*103, 14086–14091, https://doi.org/10.1073/pnas.0603424103 (2006).
- 54 Troemel, E. R. et al. p38 MAPK regulates expression of immune response genes and contributes to longevity in *C. elegans. PLoS Genet* 2, e183, https://doi.org/10.1371/journal.pgen.0020183 (2006).
- 55 Nandakumar, M. & Tan, M. W. Gamma-linolenic and stearidonic acids are required for basal immunity in *Caenorhabditis elegans* through their effects on p38 MAP kinase activity. *PLoS Genet* 4, e1000273, https://doi.org/10.1371/journal.pgen.1000273 (2008).

- 56 Peterson, N. D. et al. The nuclear hormone receptor NHR-86 controls antipathogen responses in *C. elegans*. *PLoS Genet* **15**, e1007935, https://doi.org/10.1371/journal.pgen.1007935 (2019).
- 57 Engelmann, I. et al. A comprehensive analysis of gene expression changes provoked by bacterial and fungal infection in *C. elegans. PLoS One* **6**, e19055, https://doi.org/10.1371/journal.pone.0019055 (2011).
- Zhong, W. & Sternberg, P. W. Genome-wide prediction of *C. elegans* genetic interactions. *Science* 311, 1481–1484, https://doi.org/10.1126/science.1123287 (2006).
- 59 Fletcher, M., Tillman, E. J., Butty, V. L., Levine, S. S. & Kim, D. H. Global transcriptional regulation of innate immunity by ATF-7 in *C. elegans. PLoS Genet* 15, e1007830, https://doi.org/10.1371/journal.pgen.1007830 (2019).
- Chapter III

Barker KR, Hussey RS, Krusberg LR, Bird GW, Dunn RA, Ferris H, et al.,

MacGuidwin AE. 1994. Plant and soil nematodes: Societal impact and focus for

the future. J Nematol. 26(2):127. PubMed

Irazoqui JE, Troemel ER, Feinbaum RL, Luhachack LG, Cezairliyan BO, Ausubel

FM. 2010. Distinct pathogenesis and host responses during infection of C.

elegans by P. aeruginosa and S. aureus. PLoS Pathog. 6(7):e1000982. PubMed

Luo S, Shaw WM, Ashraf J, Murphy CT. 2009. TGF-β Sma/Mab signaling mutations uncouple reproductive aging from somatic aging. PLoS Genet.

5(12):e1000789. PubMed

Mallo GV, Kurz CL, Couillault C, Pujol N, Granjeaud S, Kohara Y, Ewbank JJ. 2002. Inducible antibacterial defense system in *C. elegans*. Curr Biol.

12(14):1209–1214. PubMed

O'Quinn AL, Wiegand EM, Jeddeloh JA. 2001. *Burkholderia pseudomallei* kills the nematode *Caenorhabditis elegans* using an endotoxin-mediated paralysis.

Cellular Microbiol. 3(6):381–393. PubMed

Roberts AF, Gumienny TL, Gleason RJ, Wang H, Padgett RW. 2014. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. BMC Dev Biol. 10(1):61. <u>PubMed</u> Stiernagle T. 2006. Maintenance of *C. elegans*. Pasadena (CA): WormBook; [accessed March 28, 2019]. http://doi.org/10.1895/wormbook.1.101.1.

Chapter IV

- Medzhitov R, Janeway CA, Jr. An ancient system of host defense. Curr Opin Immunol. 1998;10(1):12–5.
- Cui J, Chen Y, Wang HY, Wang RF. Mechanisms and pathways of innate immune activation and regulation in health and cancer. Hum Vaccin Immunother. 2014;10(11):3270–85.
- McSorley R. Adaptations of nematodes to environmental extremes. Florida Entomologist. 2003;86 (2):138–42.

- Murfin KE, Dillman AR, Foster JM, Bulgheresi S, Slatko BE, Sternberg PW, et al. Nematode-bacterium symbioses - cooperation and conflict revealed in the "omics" age. Biol Bull. 2012;223(1):85–102.
- Engelmann I, Pujol N. Innate immunity in *C. elegans*. Adv Exp Med Biol. 2010;708:105–21.
- Cheesman HK, Feinbaum RL, Thekkiniath J, Dowen RH, Conery AL, Pukkila-Worley R. Aberrant activation of p38 MAP kinase-dependent innate immune responses is toxic to *Caenorhabditis elegans*. G3. 2016;6(3):541–9.
- Pukkila-Worley R. Surveillance immunity: an emerging paradigm of innate defense activation in *Caenorhabditis elegans*. PLoS Pathog. 2016;12(9):e1005795.
- Couillault C, Ewbank JJ. Diverse bacteria are pathogens of *Caenorhabditis* elegans. Infect Immun. 2002;70(8):4705–7.
- Gravato-Nobre MJ, Nicholas HR, Nijland R, O'Rourke D, Whittington DE, Yook KJ, et al. Multiple genes affect sensitivity of *Caenorhabditis elegans* to the bacterial pathogen *Microbacterium nematophilum*. Genetics. 2005;171(3):1033–45.
- Begun J, Gaiani JM, Rohde H, Mack D, Calderwood SB, Ausubel FM, et al. Staphylococcal biofilm exopolysaccharide protects against *Caenorhabditis elegans* immune defenses. PLoS Pathog. 2007;3(4):e57.

- 11. Wong D, Bazopoulou D, Pujol N, Tavernarakis N, Ewbank JJ. Genome-wide investigation reveals pathogen-specific and shared signatures in the response of *Caenorhabditis elegans* to infection. Genome Biol. 2007;8(9):R194.
- 12. Singh V, Aballay A. Regulation of DAF-16-mediated innate immunity in *Caenorhabditis elegans*. J Biol Chem. 2009;284(51):35580–7.
- Zugasti O, Ewbank JJ. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. Nat Immunol. 2009;10(3):249–56.
- Pukkila-Worley R, Ausubel FM, Mylonakis E. *Candida albicans* infection of *Caenorhabditis elegans* induces antifungal immune defenses. PLoS Pathog. 2011;7(6):e1002074.
- 15. Dzakah EE, Waqas A, Wei S, Yu B, Wang X, Fu T, et al. Loss of miR-83 extends lifespan and affects target gene expression in an age-dependent manner in *Caenorhabditis elegans*. J Genet Genomics. 2018;45(12):651–62.
- Aballay A, Yorgey P, Ausubel FM. Salmonella typhimurium proliferates and establishes a persistent infection in the intestine of *Caenorhabditis elegans*. Curr Biol. 2000;10(23):1539–42.
- 17. Garsin DA, Sifri CD, Mylonakis E, Qin X, Singh KV, Murray BE, et al. A simple model host for identifying Gram-positive virulence factors. Proc Natl Acad Sci U S A. 2001;98(19):10892–7.

- Alper S, McBride SJ, Lackford B, Freedman JH, Schwartz DA. Specificity and complexity of the *Caenorhabditis elegans* innate immune response. Mol Cell Biol. 2007;27(15):5544–53.
- Kim DH, Ewbank JJ. Signaling in the innate immune response. WormBook, The *C. elegans* Research Community. 2018;2018:1–35.
- Sun L, Zhi L, Shakoor S, Liao K, Wang D. microRNAs involved in the control of innate immunity in *Candida* infected *Caenorhabditis elegans*. Sci Rep. 2016;6:36036.
- 21. Roberts AF, Gumienny TL, Gleason RJ, Wang H, Padgett RW. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. BMC Dev Biol. 2010;10:61.
- Dierking K, Yang W, Schulenburg H. Antimicrobial effectors in the nematode *Caenorhabditis elegans*: an outgroup to the Arthropoda. Philos Trans R Soc Lond B Biol Sci. 2016;371(1695).
- Banyai L, Patthy L. Amoebapore homologs of *Caenorhabditis elegans*.
 Biochim Biophys Acta. 1998;1429(1):259–64.
- Roeder T, Stanisak M, Gelhaus C, Bruchhaus I, Grotzinger J, Leippe M. Caenopores are antimicrobial peptides in the nematode *Caenorhabditis elegans* instrumental in nutrition and immunity. Dev Comp Immunol. 2010;34(2):203–9.

- 25. Hoeckendorf A, Leippe M. SPP-3, a saposin-like protein of *Caenorhabditis elegans*, displays antimicrobial and pore-forming activity and is located in the intestine and in one head neuron. Dev Comp Immunol. 2012;38(1):181–6.
- 26. Hoeckendorf A, Stanisak M, Leippe M. The saposin-like protein SPP-12 is an antimicrobial polypeptide in the pharyngeal neurons of *Caenorhabditis elegans* and participates in defence against a natural bacterial pathogen. Biochem J. 2012;445(2):205–12.
- 27. Evans EA, Kawli T, Tan MW. *Pseudomonas aeruginosa* suppresses host immunity by activating the DAF-2 insulin-like signaling pathway in *Caenorhabditis elegans*. PLoS Pathog. 2008;4(10):e1000175.
- Murphy CT, McCarroll SA, Bargmann CI, Fraser A, Kamath RS, Ahringer J, et al. Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans*. Nature. 2003;424(6946):277–83.
- 29. Ren M, Feng H, Fu Y, Land M, Rubin CS. Protein kinase D is an essential regulator of *C. elegans* innate immunity. Immunity. 2009;30(4):521–32.
- Lakdawala MF, Madhu B, Faure L, Vora M, Padgett RW, Gumienny TL. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body size-associated genes in *Caenorhabditis elegans*. Mol Biol Cell. 2019;30(26):3151–60.
- 31. Madhu BJ, Salazar AE, Gumienny TL. *Caenorhabditis elegans* egg-laying and brood-size changes upon exposure to *Serratia marcescens* and

Staphylococcus epidermidis are independent of DBL-1 signaling. microPublication Biology. 2019, 10.17912/2r51-b476.

- Stiernagle T. Maintenance of *C. elegans*. WormBook, The *C. elegans* Research Community. 2006:1–11.
- Portman DS. Profiling *C. elegans* gene expression with DNA microarrays.
 WormBook. The *C. elegans* Research Community. 2006:1–11.
- 34. Kwon G, Lee J, Lim YH. Dairy *Propionibacterium* extends the mean lifespan of *Caenorhabditis elegans* via activation of the innate immune system. Sci Rep. 2016;6:31713.
- 35. Kim K, Kim R, Sengupta P. The HMX/NKX homeodomain protein MLS-2 specifies the identity of the AWC sensory neuron type via regulation of the *ceh-36* Otx gene in *C. elegans*. Development. 2010;137(6):963–74.
- Low IIC, Williams CR, Chong MK, McLachlan IG, Wierbowski BM, Kolotuev I, et al. Morphogenesis of neurons and glia within an epithelium. Development. 2019;146(4).
- Packer JS, Zhu Q, Huynh C, Sivaramakrishnan P, Preston E, Dueck H, et al. A lineage-resolved molecular atlas of *C. elegans* embryogenesis at single-cell resolution. Science. 2019;365(6459).
- Wheeler JM, Thomas JH. Identification of a novel gene family involved in osmotic stress response in *Caenorhabditis elegans*. Genetics. 2006;174(3):1327–36.

- Lamitina T, Huang CG, Strange K. Genome-wide RNAi screening identifies protein damage as a regulator of osmoprotective gene expression. Proc Natl Acad Sci U S A. 2006;103(32):12173–8.
- 40. Zugasti O, Thakur N, Belougne J, Squiban B, Kurz CL, Soule J, et al. A quantitative genome-wide RNAi screen in *C. elegans* for antifungal innate immunity genes. BMC Biol. 2016;14.
- Dodd W, Tang L, Lone JC, Wimberly K, Wu CW, Consalvo C, et al. A damage sensor associated with the cuticle coordinates three core environmental stress responses in *Caenorhabditis elegans*. Genetics. 2018;208(4):1467–82.
- Larance M, Pourkarimi E, Wang B, Brenes Murillo A, Kent R, Lamond AI, et al. Global proteomics analysis of the response to starvation in *C. elegans*. Mol Cell Proteomics. 2015;14(7):1989–2001.
- Alegado RA, Tan MW. Resistance to antimicrobial peptides contributes to persistence of *Salmonella typhimurium* in the *C. elegans* intestine. Cell Microbiol. 2008;10(6):1259–73.
- 44. Mallo GV, Kurz CL, Couillault C, Pujol N, Granjeaud S, Kohara Y, et al. Inducible antibacterial defense system in *C. elegans*. Curr Biol. 2002;12(14):1209–14.
- 45. Mochii M, Yoshida S, Morita K, Kohara Y, Ueno N. Identification of transforming growth factor-β-regulated genes in *Caenorhabditis elegans* by
differential hybridization of arrayed cDNAs. Proc Natl Acad Sci U S A. 1999;96(26):15020–5.

- Liang J, Yu L, Yin J, Savage-Dunn C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related signaling outputs. Dev Biol. 2007;305(2):714–25.
- 47. Zhang X, Zhang Y. DBL-1, a TGF-β, is essential for *Caenorhabditis elegans* aversive olfactory learning. Proc Natl Acad Sci U S A. 2012;109(42):17081–6.
- 48. Berg M, Monnin D, Cho J, Nelson L, Crits-Christoph A, Shapira M. TGFβ/BMP immune signaling affects abundance and function of *C. elegans* gut commensals. Nat Commun. 2019;10(1):604.
- 49. Garsin DA, Villanueva JM, Begun J, Kim DH, Sifri CD, Calderwood SB, et al.
 Long-lived *C. elegans daf-2* mutants are resistant to bacterial pathogens.
 Science. 2003;300(5627):1921.
- 50. Troemel ER, Chu SW, Reinke V, Lee SS, Ausubel FM, Kim DH. p38 MAPK regulates expression of immune response genes and contributes to longevity in *C. elegans*. PLoS Genet. 2006;2(11):e183.
- 51. Kenyon C, Chang J, Gensch E, Rudner A, Tabtiang R. A *C. elegans* mutant that lives twice as long as wild type. Nature. 1993;366(6454):461–4.
- Kimura KD, Tissenbaum HA, Liu Y, Ruvkun G. *daf-2*, an insulin receptor-like gene that regulates longevity and diapause in *Caenorhabditis elegans*.
 Science. 1997;277(5328):942–6.

- 53. Zhao L, Zhao Y, Liu R, Zheng X, Zhang M, Guo H, et al. The transcription factor DAF-16 is essential for increased longevity in *C. elegans* exposed to *Bifidobacterium longum* BB68. Sci Rep. 2017;7(1):7408.
- 54. Zhou M, Liu X, Yu H, Yin X, Nie SP, Xie MY, et al. Cell signaling of *Caenorhabditis elegans* in response to enterotoxigenic *Escherichia coli* Infection and *Lactobacillus zeae* protection. Front Immunol. 2018;9:1745.
- 55. Kim DH, Feinbaum R, Alloing G, Emerson FE, Garsin DA, Inoue H, et al. A conserved p38 MAP kinase pathway in *Caenorhabditis elegans* innate immunity. Science. 2002;297(5581):623–6.
- 56. Kim DH, Liberati NT, Mizuno T, Inoue H, Hisamoto N, Matsumoto K, et al. Integration of *Caenorhabditis elegans* MAPK pathways mediating immunity and stress resistance by MEK-1 MAPK kinase and VHP-1 MAPK phosphatase. Proc Natl Acad Sci U S A. 2004;101(30):10990–4.
- 57. Liberati NT, Fitzgerald KA, Kim DH, Feinbaum R, Golenbock DT, Ausubel FM. Requirement for a conserved Toll/interleukin-1 resistance domain protein in the *Caenorhabditis elegans* immune response. Proc Natl Acad Sci U S A. 2004;101(17):6593–8.
- Bargmann CI, Hartwieg E, Horvitz HR. Odorant-selective genes and neurons mediate olfaction in *C. elegans*. Cell. 1993;74(3):515–27.
- Troemel ER, Kimmel BE, Bargmann CI. Reprogramming chemotaxis responses: sensory neurons define olfactory preferences in *C. elegans*. Cell. 1997;91(2):161–9.

- Sagasti A, Hobert O, Troemel ER, Ruvkun G, Bargmann CI. Alternative olfactory neuron fates are specified by the LIM homeobox gene *lim-4*. Genes Dev. 1999;13(14):1794–806.
- Olofsson B. The olfactory neuron AWC promotes avoidance of normally palatable food following chronic dietary restriction. J Exp Biol. 2014;217(Pt 10):1790–8.
- Pradel E, Zhang Y, Pujol N, Matsuyama T, Bargmann CI, Ewbank JJ.
 Detection and avoidance of a natural product from the pathogenic bacterium Serratia marcescens by Caenorhabditis elegans. Proc Natl Acad Sci U S A.
 2007;104(7):2295–300.

Chapter V

- Curtis, R. H. Plant parasitic nematode proteins and the host parasite interaction. *Brief Funct Genomic Proteomic* 6, 50–58, https://doi.org/10.1093/bfgp/elm006 (2007).
- Dodd, W. et al. A damage sensor associated with the cuticle coordinates three core environmental stress responses in *Caenorhabditis elegans*.
 Genetics 208, 1467–1482, https://doi.org/10.1534/genetics.118.300827 (2018).
- Page, A. P. & Johnstone, I. L. The cuticle. *WormBook,* The *C. elegans* Research Community, https://doi.org/doi/10.1895/ wormbook.1.138.1 (2007).

- 4 Blaxter, M. L. Cuticle surface proteins of wild type and mutant *Caenorhabditis elegans. J Biol Chem* **268**, 6600–6609 (1993).
- Henry, P. et al. Fatty acids composition of *Caenorhabditis elegans* using accurate mass GCMS-QTOF. *J Environ Sci Health B* 51, 546–552, https://doi.org/10.1080/03601234.2016.1170555 (2016).
- Watts, J. L. & Ristow, M. Lipid and carbohydrate metabolism in *Caenorhabditis elegans. Genetics* 207, 413–446, https://doi.org/10.1534/genetics.117.300106 (2017).
- Schultz, R. D., Bennett, E. E., Ellis, E. A. & Gumienny, T. L. Regulation of extracellular matrix organization by BMP signaling in *Caenorhabditis elegans*. *PLoS One* 9, e101929, https://doi.org/10.1371/journal.pone.0101929 (2014).
- Clark, J. F., Meade, M., Ranepura, G., Hall, D. H. & Savage-Dunn, C.
 Caenorhabditis elegans DBL-1/BMP regulates lipid accumulation via interaction with insulin signaling. *G3 (Bethesda)* 8, 343–351,

https://doi.org/10.1534/g3.117.300416 (2018).

- Roberts, A. F., Gumienny, T. L., Gleason, R. J., Wang, H. & Padgett, R.
 W. Regulation of genes affecting body size and innate immunity by the DBL-1/BMP-like pathway in *Caenorhabditis elegans*. *BMC Dev Biol* 10, 61, https://doi.org/10.1186/1471-213X-10-61 (2010).
- 10 Liang, J., Yu, L., Yin, J. & Savage-Dunn, C. Transcriptional repressor and activator activities of SMA-9 contribute differentially to BMP-related

signaling outputs. *Dev Biol* **305**, 714–725, https://doi.org/10.1016/j.ydbio.2007.02.038 (2007).

- 11 Lakdawala, M. F. et al. Genetic interactions between the DBL-1/BMP-like pathway and *dpy* body size-associated genes in *Caenorhabditis elegans*. *Mol Biol Cell* **30**, 3151–3160, https://doi.org/10.1091/mbc.E19-09-0500 (2019).
- Madaan, U. et al. Feedback regulation of BMP signaling by
 Caenorhabditis elegans cuticle collagens. *Mol Biol Cell* **31**, 825–832,
 https://doi.org/10.1091/mbc.E19-07-0390 (2020).
- Madhu, B., Salazar, A. E. & Gumienny, T. L. *Caenorhabditis elegans* egglaying and brood-size changes upon exposure to *Serratia marcescens* and *Staphylococcus epidermidis* are independent of DBL-1 signaling. *microPublication Biology* 2019, **10**.17912/2r51-b476 (2019).
- Watts, J. L. & Browse, J. Genetic dissection of polyunsaturated fatty acid synthesis in *Caenorhabditis elegans*. *Proc Natl Acad Sci U S A* 99, 5854– 5859, https://doi.org/10.1073/pnas.092064799 (2002).
- 15 Stiernagle, T. Maintenance of *C. elegans. WormBook,* The *C. elegans* Research Community, http://doi.org/10.1895/wormbook.1.101.1 (2006).
- 16 Portman, D. S. Profiling *C. elegans* gene expression with DNA microarrays. *WormBook*, The *C. elegans* Research Community, http://doi.org/10.1895/wormbook.1.104.1 (2006).

- 17 Love, M. I., Huber, W. & Anders, S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. *Genome Biol* **15**, 550, https://doi.org/10.1186/s13059-014-0550-8 (2014).
- 18 Zhang, Y. et al. Comparative genomics and functional study of lipid metabolic genes in *Caenorhabditis elegans*. *BMC Genomics* 14, 164, https://doi.org/10.1186/1471-2164-14-164 (2013).
- Brock, T. J., Browse, J. & Watts, J. L. Genetic regulation of unsaturated fatty acid composition in *C. elegans*. *PLoS Genet* 2, e108, https://doi.org/10.1371/journal.pgen.0020108 (2006).
- Watts, J. L. & Browse, J. A palmitoyl-CoA-specific ∆9 fatty acid desaturase
 from *Caenorhabditis elegans*. *Biochem Biophys Res Commun* 272, 263–
 269, https://doi.org/10.1006/bbrc.2000.2772 (2000).
- Shaye, D. D. & Greenwald, I. OrthoList: a compendium of *C. elegans* genes with human orthologs. *PLoS One* 6, e20085, https://doi.org/10.1371/journal.pone.0020085 (2011).
- Garofalo, A. et al. The FAR protein family of the nematode *Caenorhabditis elegans*. Differential lipid binding properties, structural characteristics, and developmental regulation. *J Biol Chem* **278**, 8065–8074, https://doi.org/10.1074/jbc.M206278200 (2003).

Chapter VI

- Olofsson, B. The olfactory neuron AWC promotes avoidance of normally palatable food following chronic dietary restriction. *J Exp Biol* 217, 1790– 1798, https://doi.org/10.1242/jeb.099929 (2014).
- Zhang, X. & Zhang, Y. DBL-1, a TGF-β, is essential for *Caenorhabditis elegans* aversive olfactory learning. *Proc Natl Acad Sci U S A* **109**, 17081–17086, https://doi.org/10.1073/pnas.1205982109 (2012).
- Zugasti, O. & Ewbank, J. J. Neuroimmune regulation of antimicrobial peptide expression by a noncanonical TGF-β signaling pathway in *Caenorhabditis elegans* epidermis. *Nat Immunol* 10, 249–256, https://doi.org/10.1038/ni.1700 (2009).
- Zhong, W. & Sternberg, P. W. Genome-wide prediction of *C. elegans* genetic interactions. *Science* **311**, 1481–1484, https://doi.org/10.1126/science.1123287 (2006).
- 5 Fletcher, M., Tillman, E. J., Butty, V. L., Levine, S. S. & Kim, D. H. Global transcriptional regulation of innate immunity by ATF-7 in *C. elegans. PLoS Genet* **15**, e1007830, https://doi.org/10.1371/journal.pgen.1007830 (2019).
- 6 Madhu, B., Salazar, A. E. & Gumienny, T. L. *Caenorhabditis elegans* egglaying and brood-size changes upon exposure to *Serratia marcescens* and *Staphylococcus epidermidis* are independent of DBL-1 signaling. *microPublication* Biology, **10**.17912/2r51-b476 (2019).

- Madhu, B. & Gumienny, T. L. The DBL-1/TGF-β signaling pathway regulates pathogen-specific innate immune responses in *C. elegans*.
 bioRxiv, https://doi.org/10.1101/2021.03.30.437693 (2021).
- Alper, S., McBride, S. J., Lackford, B., Freedman, J. H. & Schwartz, D. A.
 Specificity and complexity of the *Caenorhabditis elegans* innate immune response. *Mol Cell Biol* 27, 5544–5553, https://doi.org/10.1128/MCB.02070-06 (2007).
- Madhu, B., Lakdawala, M. F., Issac, N. G. & Gumienny, T. L.
 Caenorhabditis elegans saposin-like *spp-9* is involved in specific innate immune responses. *Genes Immun* 21, 301–310, https://doi.org/10.1038/s41435-020-0108-6 (2020).
- Clark, J. F., Meade, M., Ranepura, G., Hall, D. H. & Savage-Dunn, C.
 Caenorhabditis elegans DBL-1/BMP regulates lipid accumulation via interaction with insulin signaling. *G3 (Bethesda)* 8, 343–351, https://doi.org/10.1534/g3.117.300416 (2018).
- 11 Nandakumar, M. & Tan, M. W. Gamma-linolenic and stearidonic acids are required for basal immunity in *Caenorhabditis elegans* through their effects on p38 MAP kinase activity. *PLoS Genet* **4**, e1000273, https://doi.org/10.1371/journal.pgen.1000273 (2008).
- Ding, W. et al. s-adenosylmethionine levels govern innate immunity through distinct methylation-dependent pathways. *Cell Metab* 22, 633– 645, https://doi.org/10.1016/j.cmet.2015.07.013 (2015).

Anderson, S. M. et al. The fatty acid oleate is required for innate immune activation and pathogen defense in *Caenorhabditis elegans*. *PLoS Pathog* **15**, e1007893, https://doi.org/10.1371/journal.ppat.1007893 (2019).